Abstract

Title of Dissertation:	ASSOCIATION OF LIFE EVENTS WITH DEPRESSIVE SYMPTOMS AMONG PUERTO RICAN YOUTH
	Graciela Jaschek, PhD in Epidemiology, 2014
Dissertation directed by:	Olivia D. Carter-Pokras Department of Epidemiology and Biostatistics

BACKGROUND: Early onset depressive symptoms have amplifying effects on the course of later depression, and serious personal, social and economic consequences throughout the life course. Little has been published regarding socio-demographic determinants of depression and possible mechanisms for the development of depressive symptoms among Puerto Rican youth. This study aims to examine the extent to which life events are associated with the development of depressive symptoms, and how place of residence, parental support, youth self-esteem and youth coping modify that association.

METHODS: Secondary analyses were performed of data from the longitudinal Boricua Youth Study (three annual waves between 2000-2004). The sample consisted of 10-13 year old Puerto Rican youth living in New York and Puerto Rico with no depressive

symptoms at wave 1, and complete information on depressive symptoms (DISC Predictive Scale which includes 9 significant predictors of depression)at waves 2 and/or 3.

RESULTS: Depressive symptoms increase across waves with an increase in total, negative, or positive life events in general; and social adversity, death, and family environment life events specifically. Youth with low coping consistently had a higher number of depressive symptoms than youth with high coping regardless of number of total, negative or positive life events. Youth support from parents was found to be a significant confounder for all types of life events. Parent coping was a significant confounder for social adversity events.

DISCUSSION: This study identifies various kinds of life events as risk factors that contribute to the development of depressive symptoms. Early onset depressive symptoms have amplifying effects on the course of later depression, and serious personal, social and economic consequences throughout the life course. Preventing the development of depressive symptoms at an early age should be a priority if we want to optimize the mental health and well-being of youth so that they can reach their full potential. Our findings could inform the development of a first-stage screening tool for youth at risk of developing depressive symptoms in community settings.

ASSOCIATION OF LIFE EVENTS WITH DEPRESSIVE SYMPTOMS AMONG PUERTO RICAN YOUTH

by

Graciela Jaschek

Dissertation submitted to the Faculty of the Graduate School of the University of Maryland, College Park, in partial fulfillment of the requirements for the degree of Doctor of Philosophy 2014

Advisory Committee: Professor Olivia Carter-Pokras, Chair Professor GlorisaCanino Professor Xin He Professor Sandra L. Hofferth, Dean's Representative Professor Sunmin Lee "©Copyright by

Graciela Jaschek, at gjaschek@umd.edu

Foreword

While working as a community organizer I became fascinated by the energy and optimism that immigrants bring into this country and infuse at every level of a community. I worked helping immigrants connect with their new world and organized educational, art programs and health program for their families. While some children were alienated from their parents, schools and neighborhoods, the majority of children thrived despite their sometimes chaotic lives. Many families endured poverty, discrimination, fear, abuse, and isolation but for the most part they became enmeshed in the fabric of our communities and prospered. Those observations became the inspiration for this dissertation.

Dedication

I dedicate this dissertation to my husband who supported and cheered me on throughout the process to keep me happy, and to my children who encouraged and motivated me to continue.

Acknowledgments

I want to thank my Dissertation Committee members for their support and advice. I want to thank my advisor, Dr. Olivia Carter-Pokras for her patience and help to get me through the process, and Dr. Sandra Hofferth for guiding me through the proposal's revisions during the summer of 2012 and her continued mentorship. I want to thank Dr. He for his patience with my multiple biostatistics questions and I will miss sitting in his office discussing concepts, and I also want to thank Dr. Lee for her interest and thoughtful comments. Lastly, I want to thank Dr. Canino, who joined my committee at a later time but answered all my questions so graciously and kindly. I hope to meet her in person soon.

I also want to thank the leadership at the Maryland Population Research Center for the traineeship they awarded me for the year 2010-2011 which allowed me to perform exploratory work on residential mobility and depression.

Abstract	1
ASSOCIATION OF LIFE EVENTS WITH DEPRESSIVE SYMPTOMS AMONG	
PUERTO RICAN YOUTH	3
Foreword	ii
Dedication	iii
Acknowledgments	iv
Table of Contents	v
List of Tables	. vii
List of Figures	ix
List of Abbreviations	xi
Chapter 1 – Introduction	1
Adolescent Depression	1
Epidemiology of Adolescent Depression	2
The Cost of Adolescent Depression	3
The Course of Adolescent Depression	3
Developmental Perspectives in Depression Error! Bookmark not defin	ied.
Depression among Latino Adolescents	6
Depression among Puerto Rican Adolescents	7
Stress and Stressful Events	9
Stressful Events and Depression	. 10
Effect Modifiers of the Association between Stressful Events and Depression	. 13
Stressful Events, Coping and Depression	. 13
Stressful Events, Social Support and Depression	. 14
Place of Residence and Depression	. 15
Theoretical Framework	. 17
Main Research Question	. 19
Structure of this Document	. 19
Chapter 2 – Methodology	
Overall Study Design and Data Source	. 21
Participants and Criteria for Selection	. 23
Dependent/Outcome Variable and Potential Measurement Issues	. 25
Independent Variable and Potential Measurement Issues	. 29
Potential Confounders and Effect Modifiers	. 37
Power analysis	. 43
Statistical Analyses	. 44
Human Subjects	. 52
Chapter 3 - Association of life events and depressive symptoms among Puerto Rican	
Youth	. 54
Abstract	. 55

Table of Contents

Introduction	56
Methods	59
Data Analysis	65
Results	67
Discussion	
Tables	75
Chapter 4 – Association of Separation, Death, and Social Adversity with Depressive	
Symptoms among Puerto Rican Youth	
Abstract	82
Introduction	83
Methods	85
Data Analysis	90
Results	91
Discussion	93
Tables	96
Chapter 5 – Association of Child Maltreatment with Depressive Symptoms among P	uerto
Rican Youth	101
Abstract	102
Introduction	103
Methods	105
Data Analysis	111
Results	113
Discussion	114
Tables	117
Chapter 6 - Conclusions and public health significance: Policy implications and long	term
relevance	119
References	123
Appendix 1	123
Appendix 2	125
Appendix 3	126
Appendix 4	132
Appendix 5	133

List of Tables

Chapter 2

Table 2.1.Participant's wave completion patterns

Table2.2.Number of participants by wave and missing wave pattern in manuscripts 1 and 2

Chapter 3

Table 3.1. Socio-Demographic Characteristics, and Individual and Family Resources at Wave 1 for 10-13 Year Old Youth without Depressive Symptoms by Geographic Area (weighted)

Table 3.2. Log-Linear Poisson Regressions of Number of Total Life Events, Negative Life Events, or Positive Life Events in the last 12 Months (at waves 1 or 2) on Depressive Symptoms (at waves 2 or 3 respectively)

Chapter 4

Table 4.1. Mean Depressive Symptoms (wave 2) by Socio-Demographic Characteristics (wave 1) and Types of Life Event (wave 1) for 10-13 Year Old Youth without Depressive Symptoms at Wave 1 (weighted)

Table 4.2. List of Life Events used by the Boricua Youth Study Categorized by Type of Life Events

Table 4.3. Log-Linear Regressions of Separation, Death, and Social Adversity Events in the Last 12 Months and Depressive Symptoms

Chapter 5

Table 5.1. Socio-Demographic Characteristics and Mean Depressive Symptoms for
10-13 Year Old Puerto Rican Youth with No Depressive Symptoms at Wave 1
Table 5.2. Log-Linear Poisson Regressions of Lifetime Child Maltreatment (Wave 1)
on Depressive Symptoms (Wave 3)

List of Figures

Chapter 2

Figure 2.1.Distribution of Depressive Symptoms at Wave 2 in the Study Sample
Figure 2.2.Distribution of Depressive Symptoms at Wave 3 in the Study Sample
Figure 2.3.Distribution of Total Life Events in the Study Sample
Figure 2.4. Distribution of Negative Life Events in the Study Sample
Figure 2.5. Distribution of Positive Life Events in the Study Sample
Figure 2.6. Distribution of Social Adversity Life Events in the Study Sample
Figure 2.7. Distribution of Separation Life Events in the Study Sample
Figure 2.8. Distribution of Death Life Events in the Study Sample
Figure 2.9. Distribution of Family Life Events in the Study Sample
Figure 2.10. Model of the Hypothetical Relations between Confounders, Effect
Modifiers, Life Events, and Depressive Symptoms

Chapter 3

Figure 3.1a. Mean Depressive Symptoms (wave 2) and Number of Total, Negative, and Positive Life Events by Level of Youth Coping at (wave 1) for Youth 10-13 Years with No Depressive Symptoms at Wave1 Figure 3.1b. Mean Depressive Symptoms (wave 3) and Number of Total, Negative,

and Positive Life Events by Level of Youth Coping at (wave 2) for Youth 10-13

Years with No Depressive Symptoms at Wave1

Figure 4.1. Mean Depressive Symptoms (wave 2) by Types of Life Event (wave 1)

for 10-13 Year Old Youth with No Depressive Symptoms at Wave 1

List of Abbreviations

- ADHD: Attention Deficit Hyperactivity Disorder
- ANCOVA: Analysis of covariance
- BYS: Boricua Youth Study
- **DPS: DISC Predictive Scale**
- **DS:** Depressive Symptoms
- DISC-IV: Diagnostic Interview Schedule for Children, Version IV
- DISC-P: Diagnostic Interview Schedule for Children, Parent Version
- DISC-Y: Diagnostic Interview Schedule for Children, Youth Version
- DSM-IV: Diagnostic and Statistical Manual of Disorders, Version IV
- **GED:** General Education Development
- GLM: Generalized Linear Model
- LEC: Life Event Checklist
- MECA: National Institute of Mental Health Methods for the Epidemiology of Child and
- Adolescent Mental Disorders
- NCANDS: National Child Abuse and Neglect Data System
- **PSU:** Primary Sampling Unit
- SRE: Schedule of Recent Experiences
- SRSS: Social Readjustment Rating Scale
- SAS: Statistical Analysis System
- STATA: Statistical software package developed by Stata Corp
- SUDAAN: Statistical software package developed by RTI International
- TEQ: Traumatic Experiences Questionnaire

Chapter 1 – Introduction

Little has been published regarding the development of depressive symptoms between childhood and adolescence among Puerto Rican children (1). This study aims to further understand the complex relationship between life events and depressive symptoms among 10-13 years old Puerto Rican youth, and the potential modifying role of youth coping, youth support from parents, youth self-esteem, and place of residence (Bronx, New York, and San Juan, Puerto Rico). Greater knowledge of the risk factors for depression would improve the specificity of evaluation, diagnosis and treatment of depression in mental health services for adolescents. Comparing Puerto Rican children living in South Bronx, New York to children living in San Juan, Puerto Rico represents an opportunity to examine the association between life events and depressive symptoms in two culturally similar populations who live in different environments (2).

Adolescent Depression

Depression is the leading cause of disability among adults in the U.S., and is projected to become the second cause of disability in the world by 2020 (3). Although adult depression has been recognized as a disease since ancient times, the field of child and adolescent psychiatry is relatively new and only became established over the past thirty years (4). The 1999 landmark U.S. Surgeon General's report on Mental Health encouraged continued research in the area of child and adolescent mental health, and motivated stakeholders to establish several initiatives to address the lack of national

statistics(5).As a result, two national surveys were created to monitor: 1) children's symptoms, impairment, and service use; and 2) services used by children, the costs of those services, and how benefit coverage affects service use and costs (6). In addition, the National Comorbidity Survey – Adolescent Supplement was developed to assess the mental health of youth ages 13-18 years by incorporating a broad range of DSM-IV disorders (7).

Epidemiology of Adolescent Depression

Depression is rare in childhood but one of the most common mood disorders (major depressive, dysthymic, and bipolar disorders) among adolescents with a high recurrence rate in adulthood (8). About one in five adolescents in the U.S. has a psychiatric disorder (most common among adolescents are anxiety, behavior, mood, and substance abuse) by the end of adolescence, and lifetime prevalence of depression ranges from 4% to 25% depending on the study(9-13). Results from the National Comorbidity Survey Replication – Adolescent Supplement (the first nationally representative survey of adolescents 13 to 18 years old) show that the lifetime prevalence of all psychiatric disorders among adolescents ages 13-18 is 22.2%, while the lifetime prevalence of mood disorders is 14.3%(11). The results also show that the prevalence of either major depression or dysthymia increases between early and late adolescence, almost doubling between ages 13-14 (8.3%) and 17-18 (15.4%). Despite recent research that shows that half of all lifetime psychiatric disorders have their onset during adolescence, epidemiological data

on child and adolescent mental health remain scarce in comparison to adult mental health data (10).

The Cost of Adolescent Depression

Depressive disorders in adolescence are strongly associated with psychiatric comorbidity, school failure, social difficulties, the adoption of risky behaviors, obesity, physical impairment and suicide(14-17). In addition, early onset depressive symptoms (18, 19), depression and comorbidity lead to serious depressive disorders and major functional impairment in adulthood (8, 9, 20-23)that result in serious personal, social and economic costs (\$57.5 billion total expenditure for mental disorders in 2006)(3).

The Course of Adolescent Depression

The development of psychopathology and specifically of child and adolescent depressive disorders (depression and dysthymia) are influenced by many biological, individual, family, social and contextual factors(24, 25). These factors lead to different age of onset, duration, severity, and remission and relapse rates over the life-course (8). The mean length of an episode of major depressive disorder is approximately 6-9 months (22, 26), remission of approximately 90% of major depressive episodes takes place 1.5 to 2 years after the first onset (27-29), with a 40% cumulative probability of recurrence after 2 years, and 70% after 5 years (30). Fergusson et. al. show that a small number of young people experience a high frequency of depressive episodes with close to 4% of the group reporting 10 or more episodes of depression between ages 16 and 21(31). Among all

characteristics of depression, the early age of onset has the greatest impact on individuals by affecting them for a longer proportion of their lifetime and influencing their social development and life course transitions (9).Harrington et. al. found that 60% of individuals with child or adolescent onset depression experienced one or more depressive episodes in adulthood, compared to 27% individuals with no depression (13). In addition, approximately 5% of adolescents who had a depressive episode experienced another episode within 6 months, 12% had a second episode within 12 months and 33% within 4 years (22). Most importantly, depressive symptoms in adolescence are important predictors of depressive disorders in early adulthood (32).

In a 2002 meta-analysis of 310 studies using the Children's Depression Inventory among children and adolescents 8 to 16 year old, Twenge (33) found that depression increased with age during adolescence. Several prospective studies agree that the average age of onset for major depression and dysthymia takes place in early adolescence, between the ages of 11 and 14 (11, 34-36). A review by Kessler based on the National Comorbidity Survey - Replication (the first nationally representative community survey of mental health in the United States) shows that significant risk for major and minor depression starts in the early teens (13-14 years old) and continues to increase almost linearly through young adulthood (9).

There are marked gender differences in the age of onset and the rates of depression. Research on adolescent depression agrees that girls start exhibiting more depressive symptoms than boys in adolescence (33, 37). For example, the odds of current and

lifetime depression are 2-3 times higher for girls than for boys and these differences in odds persist into adulthood (30, 38-40). Twenge found in his meta-analysis that girls' scores remained stable between ages 8 to 11 and then increased between ages 12 and 16 whereas boys' scores remained constant between ages 8 to 16 (33).

The etiology of depressive disorders and in particular depression depends on a complex interaction between genes and the environment (41). This interaction creates a situation in which children of depressed parents not only inherit a genetic vulnerability for the disorder but also have to interact with the environment of parental dysfunction (42, 43). Several studies, including twin studies, suggest that children of depressed parents are more sensitive to environmental stressors and are at greater risk to develop a diagnosable mental disorder (44). For example, Birmaher found that children of depressed parents are three times more likely to have a lifetime episode of major depression (30), while Zubenko and colleagues found that the prevalence of depression among study subjects (adults ages 18 and over) was 7.7 times greater in first-degree relatives and 3.8 greater in more distant relatives (45). More importantly, familial transmission of early onset recurrent depression is stronger than familial transmission of depression in general due to the shared environment (46).

Developmental Perspectives in Depressive Symptoms

Research indicates that the experience of a high number of depressive symptoms in adolescence is a risk factor for developing depressive disorders in adulthood (32). The

prevalence of depressive symptoms increases with age for some adolescents while it remains the same or decreases for other individuals (47). Stoolmiller et al. found that youth with persistent depressive symptoms not only had a higher mean level of symptoms but that the symptoms were stable over time (48). Garber et al. found that about a third of children with a high number of depressive symptoms reported the same level of depressive symptoms 6 to 24 months later (49).

Although studies generally show that depressive symptoms increase with age, there is no consensus about the developmental trajectories depressive symptoms follow. Stoolmiller et al. (2005) found that pre-adolescent boys experience more depressive symptoms compared to girls but that girls experience a greater increase in depressive symptoms during adolescence while depressive symptoms among boys remain stable (48).Saluja et al. (2004) found similar results with girls having a higher number of depressive symptoms than boys and a greater increase during adolescence. Dekker etal. (2007) also found that depressive symptoms among boys decreased during adolescence .

Mental Health among Latino Adolescents

The Latino population is projected to almost triple by 2050 to 132.8 million, doubling its share of the population to 30%. Estimations project that by 2020, 40% of the U.S. children will be of Latino origin (50). Few studies have been published on the mental health of Latino youth, most studies group together Latino subpopulations, or have too small sample sizes for different Latino subgroups, or fail to study important demographic

variables such as generation status or degree of acculturation (51). In 2001, a Surgeon General's Report regarding health disparities in mental health concluded that Latino children and adolescents are at significantly higher risk for negative mental health outcomes (52). Then 2001-2004 National Comorbidity Survey - Adolescent Supplement found higher rates of mood disorders among 13 to 18 year old Latino children compared to other populations (11). A 2009 Youth Risk Behavior Survey found that 32.1% of Latino youth were more likely to have felt sad or hopeless in the previous 12 months than 28.5% of youth nationwide, and 14.8% of Latino youths had contemplated suicide (one of the strongest indicators of depression) compared to 8.4% nationwide (53).

Although Latino adolescents have been found to represent a growing high risk group for poor mental health outcomes including depression, Latino adolescents remain relatively under-represented in mental health research. This lack of knowledge has direct repercussions on the development and implementation of prevention and treatment programs that are culturally informed and effective (54, 55).

Depression among Puerto Rican Adolescents

Presently, Puerto Ricans constitute the second largest group of Latinos living in the United States (9.2% of the Latino population) with persons of Mexican origin comprising the largest group (64% of the Latino population)(50). According to the U.S. Census Bureau, there are approximately 3.7 million Puerto Ricans living in the Commonwealth of Puerto Rico and approximately 4.6 million living in the 50 states and the District of Columbia(50). Puerto Ricans have a predominantly low socioeconomic status: about 44% Puerto Ricans in the Commonwealth of Puerto Rico (referred from now on as the island) and 24% in the 50 states and the District of Columbia live in poverty compared to 15.3% in the general U.S. population (56, 57). The median household income for Puerto Ricans is \$19,800on the island, and \$38,010 in the 50 states and the District of Columbia compared to \$50,046 in the general U.S population. The unemployment rate of Puerto Ricans is 18.9% on the island, and 15.0% in the 50 states and the District of Columbia compared to 9.6% in the general U.S. population. Of all the Latino subgroups, Puerto Ricans have some of the worst health outcomes reporting higher fair or poor self-assessed health status, more disability and more hospitalizations compared to other Latino groups(1).In addition to higher poverty and unemployment rates, adult Puerto Ricans have higher rates of female householders with no spouse present compared to the general U.S. population (56-58), all characteristics that threaten the stability of the family and are directly associated with increased risk for mental health disorders among children and adolescents (59).

In 2010, a report indicated that Puerto Ricans were the most disadvantaged group in New York with lower rates of school enrollment, educational attainment, and employment compared to all other racial/ethnic groups (60). Results from the 2011 Youth Risk Behavior Surveillance System from Puerto Rico show that 30.6% Puerto Rican youth grades 9-12 had *felt sad or hopeless for 2 or more weeks in a row so they stopped doing some usual activities*, compared to 24.7% of black youth and 27.2% of white youth in the U.S. (61). Puerto Rican children and adolescents have some of the worst health outcomes

for Type 1 diabetes (62), asthma (63), obesity (64), and teenage childbearing (65) compared to other Latino groups and the general U.S. population.

Relatively little is known about the mental health of Puerto Rican youth living in the U.S. compared to several studies of youth living in Puerto Rico. The National Institute of Mental Health Methods for the Epidemiology of Child and Adolescent Mental Disorders (MECA) study of 1996 found that children living in Puerto Rico exhibit higher levels of psychiatric disorders than other U.S populations (66). Canino et al. found that among 4 to 17 year old children living in Puerto Rico, the prevalence of major depression was 4.1% (67). In a study that compares Puerto Rican youth living in New York and Puerto Rico, Ramos et al. found that youth in New York had higher levels of internalizing symptoms, including depressive symptoms than youth living in Puerto Rico (68). These differences were accounted for by the experience of discrimination and exposure to violence within the two different contexts.

Stress and Stressful Events

Selye's definition of stress as "a non-specific response of the body to any demand" stimulated research on the body's response to stress in terms of a general adaptation syndrome in which the organism may experience a range of physiological changes in preparation to dealing with a perceived threat (69). Later, Lazarus defined stress as a transaction between an individual and the environment (70). Stress is implicitly understood as the result of experiencing events or situations that are seen by the individual as threatening or 'stressful'. It follows that the stressful nature of an event

(stressor) within the context of the environment is dependent on the person's perception of the event as threatening or nonthreatening. A stressful event is "any response to a situation that taxes the individual's adaptive resources." A stressful life event is then conceived as an event that is cognitively appraised by the individual as a negative or unpleasant experience.

Stressful Events and Depression

The stressful life events literature has focused on separate characteristics of events and how they affect individuals. Two competing concepts regarding the association between life events and depression have been proposed. Holmes and Masuda considered life events as having a cumulative effect with strong etiologic implications for mental health that are independent of predisposing characteristics (65). As events accumulate, they reach a crisis level that elicits adaptive responses that then lead to disease. Other researchers assign weights to life events based on the significance that each life event had for the individual (71). Certain life events are defined as 'undesirable' and consistently judged to be more stressful across age groups, socio-economic status, race, religion, and marital status. Stressful events are more likely to result in major depression (72). Research shows a definite relationship between negative events and an increase in mental disorders among children (73) and adolescents. Moreover, the presence of negative life events results in a slower recovery from mental disorders (74). Other categories of events that have been associated with psychopathology include uncontrollable events (75), and life threatening events (71). Several studies on the effects of natural and man-made

disasters (earthquakes, hurricanes, tsunamis) provide strong evidence that stressful events can lead to an increase in depressive, anxiety and post-traumatic stress symptoms (76, 77). However, disasters constitute extraordinary circumstances that most people will not experience because they live in relatively disaster-free environments and are more likely to encounter ordinary and frequent events that are less extreme (78). Research on the effect of 'daily hassles' shows that daily hassles constitute a better predictor of later psychological symptoms than life events in general (78, 79).

Child maltreatment (i.e. physical, sexual, and emotional abuse, and neglect) (80) is a traumatic event with significant health, social, and economic costs (81). Exposure to maltreatment has been associated with becoming depressed (82), suicidal thoughts (83), poor school performance (84, 85), criminal behavior, drug and alcohol use (86), and risky sexual behavior (87). The average lifetime cost of a non-fatal child maltreatment case in the U.S. has been estimated to be approximately \$210,000 in 2010 dollars (81). The cost includes productivity losses, and medical, child welfare, criminal justice, and special education costs.

Although not all occurrences of child maltreatment are reported, the number of officially reported cases highlights a widespread problem (88). In 2007, child protective agencies in the U.S. reported approximately 5.8 million cases of child abuse or neglect in the general population of which 3.5 million cases were screened for further evaluation (89). Estimates from the Developmental Victimization Survey, a nationally representative sample of children 2-17 years of age, indicate that 1 in 7 children in the U.S. will

experience some form of child maltreatment in their lifetime (90). Furthermore, a report from Child Protective agencies estimate that 2.2 per 100,000 children in the U.S. will die from abuse or neglect (89).

In 2011, child maltreatment rates among Latinos (8.6 per thousand) were comparatively higher than the rates for non-Hispanic Whites (7.9 per thousand) however, Latino cases have been found to be underreported (91). Most studies use broad categories for Latinos to describe child maltreatment (92). As a result, little is known about child maltreatment in specific Latino populations with the exception of Puerto Rico which is a territory of the U.S. Puerto Rico is included in the National Child Abuse and Neglect Data System (NCANDS), a federally sponsored system established in response to the Child Abuse Prevention and Treatment Act of 1988 to collect and analyze annual data on child abuse and neglect (89). Estimates from the 2012 NCANDS indicate that there were 10 per 1,000 child victims in Puerto Rico compared to 9.2 per 1,000 in the U.S. (89)

Early research on child maltreatment focused on the effects of single types of maltreatment but more recent studies have shown that children who are exposed to one type of maltreatment are often exposed to other types (93). As a result, researchers started examining the combined effects of different types of child maltreatment and found that there is strong evidence that the effects of exposure to multiple types of maltreatment represent a substantial risk for mental health disorders (94), major depression (5 or more depressive symptoms) (86), and minor depression (2 to 4 depressive symptoms) in particular (95).

Effect Modifiers of the Association between Stressful Events and Depression

Most people will experience stressful events in the course of their lives and while some people will suffer acute distress, others will recover from the stressful events without psychological repercussions. A large body of literature has established the importance of protective factors to prevent illness following a stressful event (96).

Stressful Events, Coping and Depression

Adolescence is considered a period of profound cognitive, social and emotional changes (97) with rapid increases in both the experience of stressful life events and the associated increase in stress (98-100). Several studies among adolescents and adults show an association between stressful life events and depression or depressive symptoms in clinical and general population studies (101, 102). However, many depressed youth do not experience a stressful life event prior to their depression and not all youth who experience stressful life events become depressed. Successful adaptation to stress depends on the environment and the resources available to the youth to cope with the stressful life events (103). Coping has been described as a response to external demands that serve to prevent, avoid or control emotional stress (104), and as an effort to mobilize personal resources and competences to deal with internal or external demands (105, 106). Coping is not a specific behavior that can be observed but rather is a complex set of situation-specific behaviors, cognitions and perceptions that individuals use to deal with a stressful situation. Studies show that individuals use multiple strategies to cope with life

situations that depend on the life events they confront (107, 108). Coping influences the choice and/or efficiency of coping strategies used in response to stressors (109) and the way an individual copes with a situation will determine the way the individual is affected by stress (25, 110, 111). Coping skills are essential for the healthy emotional and social development of children (112).

There is little agreement on how to conceptualize and measure coping and as a result there is a multitude of domains and categories used in coping scales to describe different coping strategies (103). In 2003, Skinner (111) reviewed and critiqued over 100 coping assessments compiling a list of 400 ways of coping and 13 coping dimensions or "families of coping." His review led him to conceptualize the structure of coping as a set of higher order categories -or coping domains- that reflect lower order categories -or ways of coping. He concluded that the confusion and lack of consensus about coping domains and ways of coping makes research results difficult to compare or aggregate. He also highlighted the need to establish precise coping domains that are functionally homogeneous and distinctive, and ways of coping categories that are conceptually clear, mutually exclusive and exhaustive.

Stressful Events, Social Support and Depression

Social support is usually studied as a psychosocial resource and refers to the functions provided to an individual whenthey are offered instrumental, informational, emotional, and appraisal assistance (113). The different cognitive and behavioral aspects of support

are highly correlated and can usually be summarized as perceived or received social support. Research indicates that perceived emotional social support is more strongly associated with mental health than received support. Henderson reviewed numerous studies that suggest that social support moderates the influence of stressful events on depression (114). Studies indicate that the ongoing support from family members, peers and other adults help the individual to cope with a stressful event by buffering the individual from the psychological consequences of stressors (115, 116). Studies show that the presence of social support alleviates the psychological stress created by a life event and has been implicated as a protective factor for depression (117, 118). The most influential theoretical perspective on social support states that the perception of available support and the supportive actions of others reduce the effects of stressful life events on health (119). Social support research indicates that although peer group relationships are especially important in adolescence, the most psychologically beneficial relationships that show significant stress-buffer effects are supportive family relationships, particularly supportive relationships with parents (120, 121). However, the lack of consistency in social support findings reflect methodological differences, the lack of a common definition for social support (too broad, too vague, no consensus), and the diversity of measures of social support (58, 122).

Place of Residence and Depression

Many studies show that island Puerto Ricans exhibit significantly poorer physical health but better mental health than mainland Puerto Ricans and other Latino subgroups (38). This differential in physical health may be explained by poverty rates being higher in the island compared to the mainland. For example, children living in Puerto Rico have higher rates of poverty and report worse physical and mental health compared to children living in mainland which may be due to worse access to services (38, 123). Research on internalizinf disorders by Ramos-Olazagasti et al. indicate that differences in internalizing disorders among Puerto Rican youth living in the island where they are a majority, and New York where they are a minority were accounted for by experiences of discrimination and exposure to violence(124).

Place of residence can indirectly influence the development of psychological distress through processes such as personal control, social support, and stress (125). Latino adult immigrants have been found to have lower risk of lifetime depression than U.S. born Latinos, and those proficient in English (126). Immigrants also have lower rates of depression when they arrive in the U.S. but within a generation, the risk increases (127). This finding suggests that for Latinos the process of adaptation to the new culture may be responsible for the increase (128). The place of residence may become a segregated environment in which individuals are concentrated in urban, poor neighborhoods that offer minimal resources and opportunities and exposes individuals to more stressful life events (124). Among Puerto Rican adolescents, cultural stress (the distress experienced by being exposed to a different culture) has been shown to be associated with psychiatric symptoms (antisocial behavior and internalizing disorders) among youth but the strength of the association between cultural stress and psychiatric symptoms decreased over time (129).

Theoretical Framework

The life-course perspective states that distinct life events at critical periods have an impact on an individual's future health. Health outcomes later in life are the result of both differential early exposures and differential developmental trajectories across the life-span (130, 131). McLeod states that the long-term developmental impact of life transitions depends on the forces of continuity and change: internal dispositions, the stability of environments, the timing and sequencing of life events, and the influence of life experiences, social relationships, the intersection of historical, family and individual time, and human agency on these events (132). It is assumed that numerous biological, environmental and social factors act independently, cumulatively or interactively across the life span to impact later health (133).

Adolescence is a critical period in an individual's development characterized by rapid pubertal and psychosocial maturation (97). Many studies show that there is an increase in depressive disorders between childhood and adolescence associated with pubertal changes (11, 134). The early onset of mood disorders disrupts the maturational process and interferes with critical life choices and long term adjustment (101).

Hill's theory of family stress states that the accumulation of acute stressors may lead families to physical, emotional or relational crises but that the impact of stressors can be buffered by protective factors. In Hill's model, there is a precipitating event (stressor) that interacts with the family resources and the family's perception of the event and results in the outcome. Lazarus (70) describes a stressor as an event that results in stress which is defined as "disturbances of biological and psychological functioning ... brought about by unusual threatening, damaging or demanding life conditions." Those disturbances require an adaptation to the disruption/threat of an individual's well-established personal and social values. McCubbin(135) categorized the four factors that define family resources (personal, family systems, social support and coping) and notes that perception of an event needs to be viewed in the context of coping and is a critical factor in determining whether a stressor is experienced by a family and its individual family members as a crisis or not. Conger (136) further interpreted the family stress model to explain how family stressors impair the parent's mental health and the quality of their marital relationship which in turn result in less effective parenting. Problems in parenting lead to problematic child adjustment.

We based our work in chapter 5 on the ecological-transactional model developed by Cicchetti and Lynch (137). In this model, the balance between risk and protective factors present at different ecologic levels (i.e. community, family) interact to influence the course of the child's psychological development and functioning across the life-span. For example, child maltreatment is believed to disrupt the conditions that lead to normal development, and will result in maladaptive cognitive and behavioral strategies, and psychopathology (88). The model seeks to explore the ecological complexity of the child's contexts which are conceptualized as consisting of a number of levels nested within each other. There are three nested levels: the *macrosystem* of cultural beliefs and

values at the societal level that influences family functioning, the *exosystem* includes the community environment in which families and children live, play, and work, and the *microsystem* which pertains to the family environment (138).

Main Research Question

The research questions for this dissertation focus on the association of life events and depressive symptoms among 10-13 year old Puerto Rican youth. First, we examine 'total', 'negative', and 'positive' life events to examine the effects of negative and positive appraisal of events. Second, we examine the differential effects of the number of social adversity, separation, death, and/or family environment events and depressive symptoms. Lastly, we examine the effects of single versus multiple child maltreatment on depressive symptoms. Each time we assess significant predictors and the effect modification of place of residence and youth resources (support from parents, coping, and self-esteem).

Structure of this Document

Chapter 2 presents detailed information on the methodology used in the three studies. The chapter includes a discussion of each variable used in the final model, how variables were defined and measured, and the strengths and limitations of each variable.

Chapter 3 presents information for Manuscript 1. The broad objective of Manuscript 1 is to compare the longitudinal association between cumulative life events, cumulative

perception of life events and depressive symptoms among 10-13 year old Puerto Rican adolescents and possible modifying roles of youth coping, youth self-esteem, and youth support from parents.

Chapter 4 presents information for Manuscript 2. The broad objective of Manuscript 2 is to examine the longitudinal association between individual life events, types of life events and depressive symptoms among 10-13 year old Puerto Rican adolescents and the possible modifying role of the youth ways of coping, youth self-esteem, the youth support from parents, and place of residence (living on the island or on mainland).

Chapter 5 presents information for Manuscript 3. The broad objective of Manuscript 3 is to examine the longitudinal association between experiencing child maltreatment and depressive symptoms among 10-13 year old Puerto Rican adolescents and the possible modifying role of the youth coping, youth self-esteem, youth support from parents, and place of residence (living on the island or on mainland).

Chapter 6 includes information on the public health significance of the association between life events and depressive symptoms among 10-13 year old Puerto Rican adolescents as well as the policy implications and long term relevance.

Chapter 2 – Methodology

Overall Study Design and Data Source

The Boricua Youth Study is a longitudinal study of psychiatric disorders among Puerto Rican children and adolescents ages 5 to 13 living in the South Bronx in New York City and the San Juan and Caguas Standard Metropolitan Areas in Puerto Rico. Data from multistage probability samples were weighted to represent the target areas according to the 1990 U.S. Census and were collected at both sites over three waves between 2000 and 2004: wave 1 (2000-2001), wave 2 (2001-2002) and wave 3 (2002-2004).

The Sampling Design

A more detailed description of the sampling design and procedures is presented by Bird et al. (139, 140). The sampling design and sampling procedures were similar in both sites (the Bronx, New York and San Juan, Puerto Rico). On the mainland, the sampling frame included all the census blocks in the geographical area designated as the South Bronx, New York City. On the island, the sampling frame included all of the census blocks in the seven municipalities that constitute the San Juan Metropolitan Area (SJMA). The primary sampling units (PSU) consisted of randomly selected household clusters (150 in the Bronx and 163 in Puerto Rico) within block groups as defined by the US Bureau of the Population and Housing 1990 Census. A secondary sampling unit was created by a random selection of households (clusters) at both sites from each primary sampling unit using sample probabilities that are proportional to the estimated number of households with children in the target ages in each primary area (block). A household was eligible if

it met the criteria of: 1) at least one child between the ages of 5 to 13 who were permanent residents at the time of enumeration; and 2) at least one parent living in the household identified himself/herself as of Puerto Rican background. Up to three eligible children per household could participate in the study. In households with more than three eligible children, children were selected at random using Kish tables. Any child with mental retardation or developmental disabilities was excluded. The original sample included 1,138 children from the Bronx and 1,353 children from Puerto Rico. Completion rates for waves 2 and 3 were 92% and 88% of the baseline sample respectively. The Wave 2 and Wave 3 follow-up interviews were conducted at 1 and 2 years after baseline respectively. Bird et al. (139) state that those who dropped out of the study after the baseline interview were found to live in more highly educated families and were more likely to be from the South Bronx sample. Ramos Olazagasti(141) conducted missing data analyses at the two follow-ups to test whether 15 background characteristics measured at baseline predicted missing status at follow-up and found that children living in single-parent households were more likely to be missing at wave 2 [b(SE) =0.43(0.21), p<0.05].

Procedures

Parents provided signed informed consent for their children to participate in the study while children 7 and older provided signed assent. Children and parents were interviewed at the same time by different interviewers at home. The structured in-person interviews were conducted separately with parents and children by trained lay interviewers at their
home, either in English (75.5% of parents and 95.6% of children in the Bronx were interviewed in English) or Spanish (99.5% of parents and 99.9% of children in Puerto Rico were interviewed in Spanish) depending on the respondent's preference. Audiotapes of interviews were systematically used to evaluate the quality of the interviewers. In addition, a computerized interview protocol was used to enter data automatically and avoid missing data, inappropriate skips, and out-of-range scoring. Children 10 or older at baseline had information on family socio-demographic factors, a wide range of risk factors, and child psychiatric disorders.

Participants and Criteria for Selection

My initial sample included 10-13 year old adolescents at wave 1 from South Bronx, New York City (n = 573) and San Juan, Puerto Rico (n = 655) for a total of 1,228 children. The sub-sample was chosen because measures of depressive symptoms were not administered to the younger participants (5-9 years old) due to issues of poor reliability of the Diagnostic Interview Schedule for Children, Version IV (DISC-IV) (139, 142, 143)in children younger than 10 years old (144-146).

The study examines life events at wave 1 and depressive symptoms at wave 2, and life events at wave 2 and depressive symptoms at wave 3. Table 1.1 describes all the possible wave completion patterns:

Manuscript	Wave Completion Pattern	Wave 1	Wave 2	Wave 3
1, 2, and 3	Waves 1, 2, 3	yes	yes	yes
1 and 2	Waves 1, 2	yes	yes	no

Table 1.1 Participant's wave completion patterns

1 and 3	Waves 1, 3	yes	no	yes
Excluded	Wave 1	yes	no	no

Manuscripts 1 and 2:

Inclusion Criteria: included a total of 977 Puerto Rican children 10-13 years old who met the following conditions: 1) no depressive symptoms at wave 1 (2000-2001); 2) participated at least in wave 1 (2000-2001) and either wave 2 (2001-2002) and/or wave 3 (2002-2004); 3) had complete life events data for both waves 1 and 2; and 4) had complete depressive symptoms data at waves 2 and 3.

Exclusion Criteria: excluded from the analytic sample youth with depressive symptoms at wave 1 (n=155) or youth missing life events information at wave 1 or wave 2 (n=5). We excluded children with depressive symptoms at wave 1 in order to establish a temporal sequence between life events at wave 1 and depressive symptoms at wave 2, and life events at wave 2 and depressive symptoms at wave 3. Children 5-9 years of age were excluded from the analytic sample because the instrument measuring psychiatric disorders in this age group has poor reliability (139).

Completion Pattern	Wave 1		Wave 2		Wave 3	
	No DS	DS*	No DS	DS*	No DS	DS*
Waves 1, 2, 3	881	0	835	63	839	59
Waves 1, 2	79	0	74	5	-	-
Missing DS (wave 1)	17		0		0	
Totals	977		977		898	

Table 1.2. Number of participants by completion patterns in manuscripts 1 and 2

* At wave 1 the sample only included children with no depressive symptoms (DS)

Manuscript 3:

<u>Inclusion criteria</u>: included 855 Puerto Rican children 10-13 years old who met the following conditions: 1) had wave 1 data on maltreatment; and 2) had complete data on depressive symptoms at wave 3.

Exclusion Criteria: excluded from the analytic sample were youth with depressive symptoms at wave 1 (n=155), youth not having completed wave 3 (n=79), and youth missing information on child maltreatment (n=41). Children with depressive symptoms at wave 1 were excluded in order to establish a temporal sequence between lifetime maltreatment at wave 1 and the development of depressive symptoms at waves 3. Children 5-9 years of age were excluded from the analytic sample because the instrument measuring psychiatric disorders in this age group has poor reliability (139).

Dependent/Outcome Variable and Potential Measurement Issues

In the three manuscripts we used a subset of questions from the major Depression and Dysthymia schedule of the National Institute of Mental Health Diagnostic Interview Schedule for Children, Version IV (DISC-IV) (142). The DISC-IV was administered to both children and their primary parent/caregiver to obtain a combined measure of depressive symptoms. The assessment of depression in the National Institute of Mental Health's Diagnostic Interview Schedule for Children, Version IV (DISC-IV) (147, 148)is based on the Diagnostic and Statistical Manual of Disorders (DSM-IV). Both the English and Spanish versions of the DISC-IV were used in this study. We chose to use the 9 items form the Diagnostic Interview Schedule for Children, Version IV that have been found to be significant predictors of depression (149). Lucas et al. (2001) report that the scale has good reliability (Cronbach's $\alpha = 0.82$). In all manuscripts we use the count of 0 to 9 depressive symptoms in the past 12 months reported at waves 2 (2001-2002) and 3 (20002-2004) by both youth and parents/caregivers. See Appendix 1 for a list of depressive symptoms. Depressive symptoms were the combined measure of responses to symptoms of major depression in the last 12 months reported by either the child or the primary parent/caregiver (142) to capture as many counts of symptoms as possible.

The distribution of depressive symptoms as a count variable in the study sample was skewed to the right, ranging from 0 to 9with a mean of 0.203 (Figures 1.1 and 1.2). In manuscripts 1 and 2 the prevalence of depression (at least 5 or more depressive symptoms) at wave 1 was 4.1% which is within the range provided in community studies. Community studies using DISC-IV report prevalence rates of 0.4% to 8.3% among adolescents (30).



Figure 1.1. Distribution of Depressive Symptoms at Wave 2 in the Study Sample



Figure 1.2. Distribution of Depressive Symptoms at Wave 3 in the Study Sample

Validity and Reliability of DISC-IV

The DISC-IV is a reliable tool widely used both in epidemiological surveys and clinical settings that provides data on psychiatric disorders and symptoms. Appendix 1 includes a description of mood disorders, shows an example DISC-IV question (major depression) and gives the list of the 9 DISC-IV questions for major depression that are used in this study. Despite the greater length and complexity of DISC-IV which classifies mental disorders according to DSM-IV diagnostic criteria and includes three time frames (last month, last 12 months, and lifetime), instead of one (last 6 months) as in DISC-2.3.(150), the DISC-IV compares favorably with DISC-2.3. Shaffer reported results on the test-retest reliability of the English version for the "last 12 month" reports. Test-retest reliability for the English version in clinical samples is moderate for parent reports of major depression (κ =.66) and very good for youth reports (κ =.92). Bravo reports that children 11-17 years old were less reliable informants for anxiety and depressive disorders than parents when reporting about the same disorders (143). In this study we chose to use the DISC Predictive Scale (DPS) which focuses on the 9 items that are

significant predictors of depression (149). The DPS has been used as a routine tool for the evaluation of mental disorders in children and adolescents. Lucas et al. (2001) report that the scale has good reliability (Cronbach's $\alpha = 0.82$).

DISC-IV is only one of four instruments that have been translated into Spanish and only one of two instruments with psychometric data available for U.S. Latinos (151). The DISC-IV was translated into Spanish by an international team of bilingual investigators (152). The Spanish version of the major depression schedule was assessed and found to have moderate test-retest reliability for most disorders. The moderate test-retest reliability of the Spanish version was found to be similar to the English version (κ =0.48) (143).

Validity has not been tested for the DISC-IV however the validity of DISC-2.3 in a community sample was moderate to very good for the English version (κ =.79 for major depression). The DISC-IV has been used and tested for its criterion-related validity in several psychiatric studies (153-155).

Strengths and Limitations of DISC-IV

Some of the strengths of the DISC-IV are that to insure the accuracy of the responses, the language and sentence structure was simplified, complex questions were sub-divided into shorter, simpler questions, and children were asked several questions to clarify their responses(142). Respondent-based interviews are useful in large studies because they are usually administered by lay interviewers who use computer-assisted software, lowering the total cost of the interview. Structured interviews such as the DISC-IV are instruments

that have improved the reliability of diagnoses but a great limitation is that the validity of a diagnostic interview is no greater than the validity of the diagnosis itself(156). Also, reliability of the responses is low for children 6-9 years old (that is why we excluded children 6-9 years old from this study), and it is not possible to record atypical symptoms not described in the DSM and ICD classification. Other limitations of structured diagnostic interviews are the invalid responses (usually over-reporting) that may be given by a child who misunderstands a question or is not aware of the concept being described(142).

Independent Variable and Potential Measurement Issues

Life Events in the first manuscript: Life events in the preceding 12 months were based on child reports at waves 1 and 2. Three measures were developed for analysis: (1) number of 'total' life events whether identified as positive ("mostly good or no effect") or negative ("mostly bad") by the respondent (count of 20 out of 21 positive and negative life events, excluding the item 'other' life events), (2) number of 'negative' life events (positive count of life events identified as "mostly bad" by the respondent), and (3) number of 'positive or no effect' life events (positive count of life events identified as "mostly bad" by the respondent). The list of life events in this study is derived from the original 46 items included in the Life Events Checklist and asks respondents to indicate if each event was positive or no effect' (κ =0.69) and 'negative' events (κ =0.72) (158). The advantages of using checklists in community surveys are that they are brief and easy to administer. However, checklists

have several limitations related to their retrospective nature (e.g. recall bias, recall dropping over time), and how participants interpret events (e.g. questions about events are imprecise, personal disposition influences the recall of events) (159, 160). A complete list of life events can be found in Appendix 1.

Total Life Events: The distribution of total life events in the study sample was slightly skewed to the right, ranging from 0 to 12 with a mean of 2.002 (SD=1.93). (Figure 1.3).

Distribution of cle

Figure 1.3. Distribution of Total Life Events in the Study Sample

Negative Life Events: The distribution of negative life events in the study sample was slightly skewed to the right, ranging from 0 to 7 with a mean of 0.79 (SD=1.22) (Figure 1.4).

Figure 1.4. Distribution of Negative Life Events in the Study Sample



Positive or No Effect Life Events: The distribution of 'positive or no effect' life events in the study sample was slightly skewed to the right, ranging from 0 to 9 with a mean of 1.21 (SD=1.31) (Figure 1.5).



Figure 1.5. Distribution of Positive Life Events in the Study Sample

<u>Life Events in the second manuscript</u>: The four types of life events used in this study (e.g. separation, death, family environment and social adversity) were adapted from the research on types of life events conducted by Grover et al. (2005). A complete list of types of life events can be found in Appendix 2. Each type of life events was an

aggregate of positive responses to the life events included in that category. The life events used in this study were derived from the 46 items that were originally included in the Life Events Checklist (157).

Social Adversity Life Events: The distribution of social adversity life events in the study sample was slightly skewed to the right, ranging from 0 to 4 with a mean of 0.24 (SD=0.52) (SD=1.31) (Figure 1.6).



Figure 1.6. Distribution of Social Adversity Life Events in the Study Sample

Separation Life Events: The distribution of separation life events in the study sample was slightly skewed to the right, ranging from 0 to 5 with a mean of 0.92 (SD=1.04) (SD=1.31) (Figure 1.7).

Figure 1.7. Distribution of Separation Life Events in the Study Sample



Death Life Events: The distribution of death life events in the study sample was slightly skewed to the right, ranging from 0 to 4 with a mean of 0.54 (SD=0.70) (SD=1.31) (Figure 1.8).



Figure 1.8. Distribution of Death Life Events in the Study Sample

Family Environment Life Events: The distribution of family environment life events in the study sample was slightly skewed to the right, ranging from 0 to 4 with a mean of 0.31 (SD=0.61) (SD=1.31) (Figure 1.9).

Figure 1.9. Distribution of Family Environment Life Events in the Study Sample



Validity and Reliability of the Life Events Checklist

Positive and negative life events in the LEC have acceptable validity and test-retest reliability. In this study, when we calculated the internal consistency of the list of 20 life events we found that it was poor (Cronbach's alpha=0.61).

Strengths and Limitations Life Events Checklists

There has been an ongoing debate over the advantages of checklists versus investigatorbased interviews which are considered by many as the gold-standard to measure life events (161). Checklists are brief, easy to administer and provide the potential for anonymity (162). In addition, studies based on adolescent reports show good reliability estimates for negative and positive events (163). Lastly, when interview and checklist methods were compared, fair to substantial agreement between the two methods was found for the number and perceived impact of life events (162). The potential measurement problems of life events checklists are numerous. The lack of consensus on the contents of life events checklists make comparison between studies difficult. The definition and selection of life events varies by study so checklist contents may vary in the number of items it includes but also in the items they contain. Checklists only include a limited number of events that are treated as equivalent (164) and the events may reflect a subjective appraisal that does not address the objective circumstance of the event (165). Checklists constrain the respondent to recall and report on specific events retrospectively which may introduce recall bias. In addition, the recall of the events decreases over time and the amount of events reported by respondents prior to viewing the event checklist may differ after they view the checklist (166). Determining if an event is the cause or consequence of psychopathology is particularly difficult when the onset of the psychopathology is unclear or if the psychopathology is of long duration. Using life events that are themselves signs or symptoms of depression or other mental disorders may confound the association (167). Assigning a causal effect between life events and depression incorrectly assumes that exposure to life events occurs randomly with respect to other causes of depression (164). Retrospective report of events may reflect the emotional state of the person, and the differential desire to report more stressful experiences by depressed compared to non-depressed individuals as shown by Cohen (164, 165, 168). Patients with mental disorders consider typical stressors as having greater weight than do non-disordered people (75).

<u>Child Maltreatment</u>: In the third manuscript a child maltreatment variable was created using the answers to 11 items from the Traumatic Experiences Questionnaire (TEQ) (see

Appendix 1). The TEQ is a combination of self-reported items from different scales that assess physical and sexual abuse, and neglect by family and/or non-family members. The child maltreatment variable was a combination of lifetime experiences of maltreatment assessed at wave1. The variable 'maltreatment' was constructed to have five levels: 'no maltreatment', 'neglect only', 'physical abuse only', 'sexual abuse only', and 'multiple maltreatment'. This five level maltreatment variable is similar to the variable used by Gonzalez-Tejera et al. (2005), and Sledjeski et al. (2009) in their studies. The reliability of sub-scales included in the TEQ was previously calculated (95) and found to be poor to good: physical abuse (Cronbach's alpha = .70), neglect (Cronbach's alpha = .47), sexual abuse (Cronbach's alpha = .55), and multiple abuse (Cronbach's alpha = .72). Physical abuse was derived from four questions from the child version of the Parental Discipline Scale (66). Physical abuse included having ever been hit by a caregiver with an object sometimes or many times, or ever been hit by a caregiver with a fist, kicked hard, beaten very hard, or purposely injured at least once. Sexual victimization included two items derived from a scale developed by Finkelhor and Dziuba-Leatherman (169), and was defined as being forced to look at or touch an adult's or older child's private parts, or having an adult or older child ever try to touch, grab, or kiss the child in a sexual way one or more times. Neglect was derived from four items in the Parental Discipline Scale (66): having ever been left alone at least two times, or left without food, necessary medical care, or having a caretaker who could not take care of the child due to substance abuse at least one time. Based on the co-occurrence of child abuse and neglect, child maltreatment was categorized into five levels depending on the type of maltreatment: no maltreatment, neglect only, physical abuse only, sexual abuse only, and multi-type maltreatment.

Potential Confounders and Effect Modifiers

Potential Confounders

In manuscripts 1 and 2, potential confounders were identified based on the theoretical framework (Hill's Theory of Family Stress) (170) and risk factors associated with life events and mood and/or other mental disorders in the Boricua Youth Study and the National Comorbidity Survey Replication - Adolescent Supplement literature. In this study, risk factors include youth's age and gender, and indicators of family socio-economic status such as family composition (single or two-parent household), maternal education and per capita household income (11, 56, 141, 171-173). Family resources were outlined by McCubbin (135) based on Hill's Theory of Family Stress and include: personal resources (financial, educational, health resources), the family system's internal resources (how family manages stress), social support, and coping.

In manuscript 3, potential confounders were identified based on the ecologicaltransactional model developed by Cicchetti and Lynch (137). In this model, the balance between risk and protective factors present at different ecologic levels (i.e. community, family) interact to influence the course of the child's psychological development and functioning across the life-span. For example, child maltreatment is believed to disrupt the conditions that lead to normal development, and will result in maladaptive cognitive and behavioral strategies, and psychopathology (88).

Age – Integer calculated from the date of birth to the interview date for children ages 10 to 13: *What is your date of birth?* - given as month/day/year

Gender - Obtained at the baseline interview from observation by the interviewer: *Boy/Girl*

Per Capita Household Income - Obtained from the parent report and calculated by dividing the household income by the number of residents living in the household unit at the time of the interview. The per capita household income was obtained from the household listing (0-12 household members) and parental report of income (23 income categories ranging from no income to \geq \$100,000): *Income: Total income to* ***'s *household before taxes for the PAST YEAR. Please include salaries, wages, social security, welfare, and any other income for everyone living in this household. (Include child support or alimony)*

Household listing: How many other persons, children and adults, live in the household, besides the child, biological mother, biological father, other mother figure, and other father figure?

Mother's Education - Obtained from the parental demographic questionnaire (highest level completed by each parent). Mother's education was categorized as less than high school, high school or GED, and some college:(172)*How many years of schooling did* [you/***'s biological mother] complete? (In general: School=00-12; College=13-16; Graduate school=17+) Did [you/she] receive a high school diploma or GED?

IF YES, ASK: A. Did [you/she] receive a college degree (including an Associate degree) beyond high school?

Family Composition - Obtained at the time of the interview from the parent report. The variable was developed for the study and included 2 categories: single or two parent household. Parent was defined as biological mother (89%), biological father (1.8%); other mother and other father figure. The 'other' parent figure included grandmother/grandfather (4.7%), adoptive (3.0%) or step parent (0.4%), biological/adoptive sister (0.1%), aunt, cousin, other female figure (1.0%).

Parent/Caregiver Received Social Support - Derived from the sum to answers to three items of the Parent Social Support scale in the adult questionnaire. Three items with the best internal consistency scores in the acceptable range (Cronbach's $\alpha = .70$, at wave 1) were selected by performing multiple correlation analyses (174). The three items are consistent with received social support scales discussed in the literature (128, 175, 176) and relate to how often a parent gets together with family members, how often parents attend family gatherings and how often family members take care of the respondent's children. Response categories include 4 levels ranging from 0 = never/once a year or*less*, to 3 = at *least once a week* for all items.

Parent/Caregiver Perceived Social Support - Measured by one item of the Parent Social Support scale of the adult questionnaire: "In general how satisfied are you with the amount of support that you receive in your life?" Response categories ranged from 0 = *very satisfied*, to 3 = very *unsatisfied* for all items.

Parent/Caregiver Coping - Count of answers to the scale of Parental Coping from the adult questionnaire. The scale consists of 8 items such as "Have you felt that you were unable to control the important things in your life?" (reverse coded to unify the scale in the same direction) and "Have you successfully dealt with irritating life hassles?". The internal consistency of the scale was calculated (Cronbach's $\alpha = .70$, at wave 1) to establish its reliability and was found to be acceptable (174). Response categories included 3 items ranging from 0 = almost never to 2 = often for all items.

Maternal Depression - Assessed through the parent's report. Information on maternal depression was obtained from an adaptation of the depression schedule of prime maternal depression from Spitzer (177). The variable is dichotomized as indicating the presence or absence of maternal depression in the last 12 months.

Potential Effect Modifiers

Potential effect modifiers tested in the three manuscripts included place of residence, youth support from parents, youth coping, and youth self-esteem.

Place of Residence - Based on the interviewer's report in the Profile section of the child questionnaire and were either South Bronx, New York or San Juan, Puerto Rico.Place of residence has been found to influence patterns of internalizing disorders (178).

Youth Support from Parents - Derived from positive count of answers to the Parent/Child Relationship scale of the child questionnaire and measured at waves 1 and 2. The Parent/Child Relationship scale consisted of 12 items (e.g. "How often do your parents/caretakers help you make important decisions?") and had good reliability when calculated for this study (Cronbach's $\alpha = .74$, at wave 1). Responses consisted of 2 categories, where 0 = rarely or never and 1 = sometimes or often for all items. Based on the sample distribution, Youth support from parents was categorized into three levels (low= <9 items, moderate= 9-11 items, high= >11 items).

One limitation of social support measurement is the lack of a gold standard so it is difficult to draw conclusions across studies about the effects of social support on mental health. Another limitation is the variability of support over time and the difficulty of assessing those variations and the impact they have on depression (120).

Youth Coping - Assessed through the Ways of Coping scale of the child questionnaire (179) which included nine items at baseline. For this study we selected five items to represent five coping styles. These five coping styles appeared in 25-50% of the approximately 100 coping systems reviewed by Skinner (111) and were used in scales with children, adolescents, and adults. Coping was only measured at wave 1 (trait coping) and was treated both as a count variable and was also dichotomized (low= <3 items, and high= \geq 3 items) based on the weighted sample mean (3.28 items). It included items such as "When you have a problem... you do what you have to do to solve it?" Responses in

the child questionnaire consisted of two categories, where 0 = rarely or never and 2 = sometimes or often (the latter category was recoded as 1 for the purpose of the analyses).Cronbach's alphas were calculated for several combinations of the Ways of Coping items but dropping items did not improve the coefficients and a decision was made to use the full scale. The full scale had low reliability (Cronbach's $\alpha = .45$, at wave 1) possibly because the scale was not developed for Latino populations and may not be as reliable for that population (28). However, we retained this scale because it has a theoretical base, and results can be compared with other Boricua Youth Study research (180).

One of the main limitations to measuring 'ways of coping' is the large number of ways of coping that have been identified and categorized into dimensions but there is a lack of consensus on the dimensions or categories of coping (111). Also, there has been a lack of consensus in determining if coping is a process that depends on specific situations or a style that individuals use in all situations (111, 112, 181).

Youth Self-Esteem – Count of positive responses to eight questions from the youth Self Esteem questionnaire. The questionnaire included items related to how youth felt about themselves, how they liked the way they behaved, and if they felt as smart as other children, if they thought they were good in sports, popular among children their own age, felt that grown-ups liked them, wished they were somewhere else, and/or felt pleased with themselves. Responses to questions consisted of two categories, where 0 = no and 2

= yes (the latter category was recoded as 1 for the purpose of the analyses). We calculated the reliability of the scale and found it to be good (Cronbach's α = .90, at wave 1).

Power analysis

Most of the studies of the association between total life events, negative life events, or positive life events and depressive symptoms are cross-sectional. This study is longitudinal and examined total life events, negative life events, or positive life events and the association with depressive symptoms between three waves. The power calculations are based on correlations between wave 1 and wave 2 data, and between wave 2 and wave 3 data.

Ramos Olazagasti conducted a study on the contextual risks and promotive processes in internalizing trajectories among 10-13 year old Puerto Rican adolescents.(68) The study which is based on the Boricua Youth Study shows that the correlation of internalizing symptoms at three points in time varies between 0.34 and 0.48. Mean internalizing symptoms decrease from 14.62 at wave 1, to 10.23 for wave 2, and 8.22 for wave 3.

The STATA/IC 10.0 test of means for repeated measures function was used for the power calculations. The ANCOVA method was used in the power calculations because it was found to have the highest statistical power and be the method of choice when analyzing the results of baseline and follow-up results (182). The following are the results of the power calculations:

Assumptions	Wave 1 – Wave 2	Wave 2 – Wave 3	Wave 1 – Wave 3
Alpha	0.05	0.05	0.05
Mean 1	14.62	10.23	14.62
(Time 1)			
Mean 2	10.23	8.22	8.62
(Time 2)			
Sd1	1	1	1
Sd2	2	2	2
Sample size*	977	977	855
Correlation between	0.40	0.34	-
F/U measurements			
Correlation between	0.34	0.48	0.48
baseline and F/U			
Estimated power	1.00**	1.00**	1.00**
(ANCOVA method)			

*Analytic sample of 10-13 year old youth (598=South Bronx, New York City; 673=San Juan, Puerto Rico)

** Actual estimated power in STATA output

Statistical Analyses

Quality assurance procedures were strictly followed throughout the dissertation project in order to maintain the highest quality of data. To minimize errors, the accuracy of the data was checked periodically before each analysis. The data were checked for variable ranges, against reference data, for typographical errors, and for logical consistency. The data were also checked for inconsistencies and missing values. Data were analyzed and the integrity of the system was maintained by protecting the data by storing it in a password protected computer with limited access.

Descriptive analysis and diagnostics

Descriptive statistics were calculated to check for missing values, outliers and errors and to provide a quantitative description of the study population in terms of family and individual socio-demographic variables, and resources variables. Chi-square and *t*-tests were performed to examine categorical and continuous variables. In manuscript 3, GLM procedures to examine mean depressive symptoms for each categorical and continuous variables and youth support from parents. Although in manuscript 3 we treated youth support from parents as a continuous variable, we created a binary measure of youth support from parents based on the mean social support: low (<10 items) and high (\geq 10 items) social support or the purpose of presenting our results. Multicollinearity between variables was assessed with tolerance scores using the standard cut off of 0.1. All tolerance values were found to be in the acceptable range between 0.97 and 0.99.

Log-linear Poisson regression models were used in manuscripts 1 and 2 to examine the association between the life events in the previous 12 months at wave 1 and reports of depressive symptoms at wave 2, and the life events in the previous 12 months at wave 2 and reports of depressive symptoms at wave 3 among youth with no depressive symptoms at wave 1. The covariates used in manuscripts 1 and 2 were from wave 1 when analyzing life events at wave 1, and from wave 2 when analyzing life events at wave 2.

In manuscript 3, we used a log-linear Poisson regression model to examine the association between lifetime maltreatment reported in wave 1 and depressive symptoms in wave 3 among youth with no depressive symptoms at wave 1. The covariates used in the third model were from wave 1.

SAS® software Version 9.3 (SAS Institute Inc., Cary, NC, USA) was used to perform variable selection. SUDAANTM software (release 11, 2011) (183) was used to adjust standard errors for correlations resulting from the complex multistage sampling (observations nested within individuals, siblings nested within households, and households nested within census blocks). Weights were used in all analyses to adjust for the difference in probability of selection and to represent the age and gender distribution of the 2000 census. All p-values were considered significant at the < 0.05 level.

Analysis of potential confounders and effect modifiers

Potential Confounders

Potential confounders were identified by conducting a bivariate analysis to assess their association with total, negative, or positive life events and depressive symptoms (conditional on exposure) at the alpha=0.10 level (184, 185). Once identified as potential confounders, the variables were further tested by adding them one at a time to the unadjusted model between depressive symptoms, and total life events or negative life events. A 5% change in the main effect estimates between the unadjusted and adjusted models further confirmed the variable as a confounder and was added to the final model.

Potential Effect Modifiers

Baron and Kenny identify a moderator (effect modifier) as a variable that affects the direction and/or strength of the relationship between the exposure and the outcome (186). Log-linear Poisson distribution regression analyses were performed to examine each potential effect modifier. To test for multiplicative effect modification, we added an interaction term between the exposure and the potential moderator to the model.

In this study, only multiplicative interaction was studied. One interaction term was added at a time as we studied three variables as effect modifiers. If the interaction term was statistically significant -which meant that the variable was modifying the effect of events on depressive symptoms- it was added to the final model along with the main effect (187).

Missing Data

Responses recorded as 'don't know' or 'refused' (to answer) were recorded as a missing value. Only participants with waves 1, 2 and 3 information (82.06%) and participants with waves 1 and 2 information (7.12%) were included in the study for a total of 977 individuals. Participants with only wave 1 information (7.86%) or waves 1 and 3 information (2.97%) were excluded. The group of participants who were excluded from the study had more mean depressive symptoms (0.17 compared to 0.10), experienced more mean life events (2.7 compared to 2.1), included more females (53% compared to 47.4%), were slightly younger (11.7 versus 12.2), were more likely to live in a single parent homes (54.8% compared to 44%), were more likely to live in the Bronx (93.2% compared to 86.5%), had higher per capita household income (\$4,458 compared to

\$3,898), and included more children with mothers with more than a high school education (20% compared to 10%).

Our sample only included children with no depressive symptoms at baseline (wave 1), excluding children with depressive symptoms (n=155). The excluded children who had depressive symptoms at wave 1 were compared to the children with no depressive symptoms at wave 1. It was found that excluded children who had depressive symptoms at wave 1 were: more likely to be female (58% compared to 48%), more likely to have mothers with more than high school education (37% compared to 20%), more likely to live in Puerto Rico (20% compared to 15%), and more likely to live in a single parent home (52% compared to 46%).

Additionally, children who started at wave 1 but dropped out by wave 3 were compared to children who completed the three waves of information using the Wilcoxon Mann Whitney test(a non-parametric analog to the independent samples t-test when the dependent variable is not assumed to be normally distributed). It was found that they were similar in the number of life events (p=0.975), number of negative life events (p=0.810), mean age (p=0.733), per capita household income (p=0.145), youth social support (p=0.245), youth ways of coping (0.922), and parent coping (p=0.7602). However, they were more likely to live in a two parent/other guardian household (p<0.0001), or with mothers who had attained a higher educational level (p<0.0001), and/or mothers who had been diagnosed with depression (p<0.0001) compared to children who were included in the study.

Sample Weights

The two samples in the Boricua Youth Study were representative samples of 5-13 year old children with at least one parent identifying themselves as Puerto Rican residing in 2,000 households in the San Juan Metropolitan Area, and in 2,000 households in the South Bronx. The samples were weighted to correct for differences in the probability of selection and the 2000 Census in the age/gender distribution. All statistical analyses were weighted to represent the age and gender distribution for the South Bronx and the standard Metropolitan Areas of Puerto Rico according to the 2000 US Census data (139).

Final Model Testing

The model used needed to be able to handle longitudinal data, weights, complex sampling design and a Poisson distribution. Three statistical software packages were assessed for their strengths and limitations regarding these capabilities:

Software	Longitudinal	Poisson	Weights	Complex
				sampling
SAS	Yes	Yes	Yes	No
STATA	No	Yes	Yes	Yes
SUDAAN	Yes	No	Yes	Yes

SUDAAN[™] was selected for the analyses because it is a powerful tool for analyzing repeated measures and cluster-correlated data like in the Boricua Youth Study.

Further analyses were conducted to check for the usefulness of using sample clusters and sample weights in the study. First, the calculation of the number of PSUs minus the number of strata yielded a number that was greater than 20 which according to Korn et al. (188) roughly corresponds to an inefficiency of 13% so clustering was used in the analyses. Second, we calculated the degree of inefficiency (weighted compared to unweighted analyses) using the following formula: $1 - (SE_{unwt}/SE_{wt})^2$ (weighted and unweighted SEs of the mean difference). The inefficiency was less than 10% so sample weights were used in the analysis(188).

The model in manuscripts 1 and 2 was formulated with the following equations:

 $log[E(Y_{ij}|b_i)] = DS_{ij} = \beta_0 + \beta_1 x_{ij-1} + b_i$

 $(x_{ij-1} = life events, types of life events)$

 $(I_{ij} = indicator to identify individuals with 2 or 3 waves of information)$

The model in manuscript 3 was formulated with the following equation:

 $log[E(Y_{ij}|b_i)] = DS_{ij} = \beta_0 + \beta_1 x_{1ij-1} + \beta_2 x_{2ij-1} + \beta_3 x_{3ij-1} + \beta_4 x_{4ij-1} + \beta_5 x_{5ij-1} + b_i$ (x_{1-5ij-1} = 5 levels of maltreatment)

 $(I_{ij} = indicator to identify individuals with 2 or 3 waves of information)$

Effect modification was tested by adding interaction term and main effects. Only statistically significant interaction terms were added in the final model:

 $\log[\mathbf{E}(\mathbf{Y}_{ij}|\mathbf{b}_i)] = \mathbf{D}\mathbf{S}_{ij} = \beta_0 + \beta_1 \mathbf{x}_{ij-1} + \beta_2 \mathbf{x}_{ij-1} + \mathbf{b}_i$

 $(x_{ij-1} = life events)$

 $(t_{ij}=$ indicator to identify individuals with 2 or 3 waves of information) $(z_{ij}=$ effect modifier)

In manuscripts 1 and 2, the analytic sample was divided into two groups. Each group included: 1) Group 1: life events, negative life events, and positive life events at wave 1 and depressive symptoms at wave 2, and 2) Group 2: life events, negative life events, and positive life events at wave 2 and depressive symptoms at wave 3. Unadjusted regressions were conducted with the two groups to confirm that the direction and magnitude of the effects were similar and did not violate assumptions for a linear model(Model 1: b=0.21, p<0.0001; Model 2: b=0.22, p<0.0001) and then the two groups were combined to be examined as one model. Each respondent had therefore two counts of measures for the two time points (wave 1 covariates and wave 2 depressive symptoms; and wave 2 covariates and wave 3 depressive symptoms). A time variable distinguished the counts from the same individual.

In manuscript 3, the analytic sample used information from wave 1 for the main exposure and covariates and wave 3 for depressive symptoms. The covariates used in the final model were reported at wave 1. We analyzed three models to examine the association between child maltreatment and depressive symptoms. Model 1 was unadjusted, Model 2 was an adjusted model including significant covariates, and Model 3 was an adjusted model including significant covariates and the interaction term between maltreatment and youth support from parents. Model

Figure 2.10. Model of the Hypothetical Relations between Confounders, Effect Modifiers, Life Events, and Depressive Symptoms



Human Subjects

The Boricua Youth Study was originally approved by the Institutional Review Board of the University of Puerto Rico (UPR) and UPR was the IRB on record. Approval from the Institutional Review Boards of UPR and University of Maryland College Park were obtained (Project Number 383803-1) for this dissertation. The current study used deidentified data and no additional data were collected. All applicable institutional regulations concerning the ethical use of human subjects were followed during the research project. Information for the Boricua Youth Study was collected and processed under strict confidentiality to protect participant privacy and has been only used for research. Chapter 3 - Association of life events and depressive symptoms among Puerto Rican

Youth

This manuscript follows the authors' instructions for the Journal of Abnormal Psychology at http://www.apa.org/pubs/journals/abn/index.aspx

Abstract

Little has been published regarding socio-demographic determinants of depression and possible mechanisms for the development of depressive symptoms from childhood to adolescence in Latino populations. This study aims to examine the extent to which life events, place of residence, parental support, youth self-esteem and youth coping are related to the development of depressive symptoms among Puerto Rican youth. Secondary analyses were performed of data from the longitudinal Boricua Youth Study (three annual waves between 2000-2004). The sample consists of 10 to 13 year old Puerto Rican youth residing in New York and Puerto Rico with no depressive symptoms at baseline (n=977). Our results indicate that: 1) depressive symptoms increase across waves with an increase in the numbers of 'total', 'negative', and 'positive or no effect' life events; 2) no multiplicative interaction was found for place of residence, youth support from parents, youth self-esteem and youth coping; 3) youth with low coping consistently had a higher number of depressive symptoms than youth with high coping regardless of number of 'total', 'negative' or 'positive or no effect' life events. This study identifies 'total', 'negative', and 'positive or no effect' life events as risk factors that may be considered in early interventions before depression develops.

Introduction

Although Latino adolescents have been found to represent a growing high risk group for poor mental health outcomes (52), little has been published regarding social determinants of depression among Latinos (189). Specifically, little is known of the possible mechanisms for the development of depressive symptoms from childhood to adolescence among Latino subgroups(1).

Results from the 2011 Youth Risk Behavior Surveillance System from Puerto Rico show that 30.6% Puerto Rican youth grades 9-12 had "felt sad or hopeless for 2 or more weeks in a row so they stopped doing some usual activities", compared to 24.7% of black youth and 27.2% of white youth in the U.S. (61). Puerto Rican youth also suffer from chronic conditions and negative outcomes that place them at higher risk for depression (190). For example, Puerto Rican youth have high rates of Type 1 diabetes (62), asthma (63), obesity (64), and teenage childbearing (65) compared to other Latino groups and the general U.S. population.

Puerto Ricans living in New York are one of the most disadvantaged groups with lower rates of school enrollment, educational attainment, and employment compared to all other groups (60). Puerto Ricans living in the island are not better off. A 2000 Population Reference Bureau report revealed that although the conditions for children improved between 1990 and 2000 in Puerto Rico, 58% of children still lived in families with incomes below the poverty line compared to 16% of children in the U.S., and one out of

seven adolescents ages 16 to 19 did not receive a high school diploma compared to one in ten in the general U.S. population (191). In addition, Puerto Ricans in the U.S. live under conditions that threaten the stability of the family and are directly associated with increased risk for mental health disorders among children and adolescents (59). For example, Puerto Ricans in the U.S. have higher rates of unemployment, divorce, and female householders with no spouse present compared to the general U.S. population (56-58).

Major depression is recognized as a serious psychiatric illness among adolescents, disrupting their maturation process and interfering with critical life choices and long term adaptation (101). Nationally representative surveys indicate that the prevalence of major depression and dysthymia increases between early adolescence and adolescence (11, 134), almost doubling between ages 13-14 (8.3%) and 17-18 (15.4%). However, approximately 50-67% of depressed adolescents will not be identified as being depressed by their health care provider and will therefore not receive any type of care for their depression (9, 29, 42, 192). The failure to identify depressed adolescents is amplified by the lack of understanding by the parent/caregiver of their child's psychiatric symptoms and by the parent/caregiver's uncertainty of when to seek help (146). Adolescents may present with comorbid mental health disorders (e.g. 30-80% anxiety, 10-80% disruptive, and 20-30% substance abuse) that make the diagnosis of depression uncertain (193). The major issue is that adolescents may present with depressive symptoms that do not meet the full criteria for depression as specified in the *Diagnostic and Statistical Manual of Mental Disorders*(194). However, youth with sub-clinical depression have been shown to

be significantly impaired, use more mental health services than youth who had clinical depression (95), and be more likely to develop adult depression (32, 136).

Adolescence is a critical developmental period (97) characterized by rapid increases in the experience of life events (98-100). Children's awareness of life events starts to emerge late in childhood as children experience more autonomy (195), and face big physical, social and cognitive changes (196). Research shows a robust and causal association between life events, stress and depression both in community and clinical populations (72, 98, 101, 164, 165, 197-200). The experiences of positive or negative life events lead individuals to make psychological adjustments (201). The cumulative effect of psychological adjustments can trigger adaptive responses that cause stress and eventually lead to disease (201). The presence of depressive symptoms can serve as an indicator of maladaptive responses to internal and external events (99). Other research has focused on undesirable (202), uncontrollable (75), or life-threatening (71) life events to show that the accumulation of negatively appraised events is more stressful than the accumulation of both positive and negative events. Individuals who experience stressful life events compared to individuals who experience non-stressful life events are more likely to develop depression and other mental disorders (99, 203).

This study focuses on life events that take place at a critical developmental period. According to the life course approach (130, 131), events that occur during childhood and adolescence could have an impact on future health through both differential early exposures and developmental trajectories across the life-span. In addition we seek to test
Hill's theory of family stress (135, 156) which states that there is a set of major stressors that lead families to physical, emotional or relational crises, and a set of key protective factors such as family environment, and family and individual resources that lead to a successful adaptation to stress (103). Protective factors that modify the psychological consequences of stressors and stressful events on depression may include individual coping and social support from family members and other adults (e.g. advice, access to resources) (25, 110, 111, 114-116).

This longitudinal study aims to examine a sample of 10-13 year old Puerto Rican youth living in New York and Puerto Rico from the Boricua Youth Study to determine the extent to which life events - whether negative, positive, or in combination - are associated with the development of depressive symptoms. We hypothesize that an increase in the experience of negative life events is associated with a greater number of depressive symptoms compared to an increase of either total or positive life events. We also want to test whether ways of coping, self-esteem, youth support from parents, and place of residence have a potential moderating role in the association between life events and depressive symptoms.

Methods

The Boricua Youth Study is a longitudinal study of psychiatric disorders among 5-13 year old Puerto Rican children and adolescents living in the South Bronx, New York and the San Juan and Caguas Standard Metropolitan Areas in Puerto Rico. Data from multistage probability samples that represent the target areas according to the 1990 U.S.

Census were collected at both sites over three annual waves (2000-2004). Up to three children per household were eligible if the household met the following criteria - the presence of a 5-13 year old child, and both the child and at least one of the primary caregivers had to be of Puerto Rican origin. Structured in-person interviews were conducted in English or Spanish by trained lay interviewers with parents and children at their homes. Further details on design and procedures for the Boricua Youth Study are described by Bird et al. (2006).

This study included children 10-13 years of age at baseline who met the following criteria: 1) no depressive symptoms at wave 1 (2000-2001); 2) participated at least in wave 1 (2000-2001) and either wave 2 (2001-2002) and/or wave 3 (2002-2004); 3) had complete life events data for both waves 1 and 2; and 4) had complete data on depressive symptoms at waves 2 and 3. Youth were excluded if they had depressive symptoms at wave 1 (n=155) or were missing life events information at wave 1 or wave 2 (n=5) resulting in an analytic sample of 977 youth. We limited the analytic sample to youth 10 years old and older who answered the youth version of the National Institute of Mental Health Diagnostic Interview Schedule for Children, Version IV (DISC-IV) (142) because reliability for this instrument in younger populations is poor (139). We excluded children with depressive symptoms at wave 1 to establish a temporal relationship between life events and the onset of symptoms after the event.

Outcome Variable – In this study we use the count of 0 to 9 depressive symptoms in the past 12 months reported at waves 2 (2001-2002) and 3 (20002-2004) using the

Diagnostic Interview Schedule for Children, version IV (142). The DISC-IV is an acceptable and widely used instrument to ascertain a wide range of child and adolescent psychiatric diagnoses. Despite the large number of diagnostic instruments, the DISC-IV is one of only two instruments that have psychometric data available for U.S. Latinos (151). The instrument was translated and adapted into Spanish, and two studies were conducted in Puerto Rico to assess test-retest reliability (143). Overall parent reliability was found to be moderate (κ =0.57) for major depression (5 or more depressive symptoms) compared to the reliability of children 11-17 years of age (κ =0.15) (143). Similar results were reported for the English DISC IV. In this study we chose to focus on the nine items that are significant predictors of depression (149). We chose to focus on a count of depressive symptoms since depression is known to have its onset in adolescence, and our population was too young to include a significant number of youth with diagnosable depression. Lucas et al. (2001) report that the scale has good reliability (Cronbach's $\alpha = 0.82$). For the purpose of this study, depressive symptoms were the combined measure of positive counts of symptoms for major depression in the last 12 months reported by either the child or the primary parent/caregiver (142) to capture the most complete information on depressive symptoms.

Exposure Variable – Life events in the preceding 12 months were based on child reports at waves 1 and 2. Three measures were developed for analysis: count of 'total' life events whether identified as 'positive or no effect' ("mostly good" or "affected them not at all or only a little") or 'negative' ("mostly bad") by the respondent (count of 20 out of 21 positive and negative life events, excluding the item 'other' life events), number of

'negative' life events (count of life events identified as "mostly bad" by the respondent), and 'positive or no effect' life events (count of life events identified as "mostly good" or "affected them not at all or only a little" by the respondent). The list of life events in this study is derived from the original 46 items included in the Life Events Checklist and asks respondents to indicate if each event was positive or negative (157). Brand and Johnson (1982) found that the test-retest reliability for the Life Events Checklist was found to be substantial for both 'positive or no effect' (κ =0.69) and 'negative' events (κ =0.72). The advantages of using checklists in community surveys are that they are brief and easy to administer. However, checklists have several limitations related to their retrospective nature (e.g. recall bias, recall dropping over time), and how participants interpret events (e.g. questions about events are imprecise, personal disposition influences the recall of events), (159, 160).

Potential Confounders and Effect Modifiers - Potential confounders were identified through the literature review, and their selection was based on Hill's Family Stress Theory (170). Socio-demographic and family resource variables included: age (in years), gender (reported by the interviewer), place of residence (San Juan, Puerto Rico; and South Bronx, New York), family composition (single or two biological/other parent household), per capita household income (calculated by dividing household income by the number of household dwellers not adjusted for the U.S. and Puerto Rico), and parental education (less than high school, high school, and more than high school). Parent/caregiver **received** social support was a sum derived from answers to three items of the Parent Social Support scale in the adult questionnaire. Three items with the best internal consistency scores in the acceptable range (Cronbach's $\alpha = .70$, at wave 1) were selected by performing multiple correlation analyses (174). The three items are similar to those found in other received social support scales in the literature (128, 175, 176). Items relate to how often a parent gets together with family members who do not live at home, how often parents attend family gatherings and how often family members take care of the respondent's children. Response categories include 4 levels ranging from 0 = *never/once a year or less*, to 3 = *at least once a week* for all items.

Parent/caregiver **perceived** social support was measured by one item of the Parent Social Support scale of the adult questionnaire: "In general how satisfied are you with the amount of support that you receive in your life?" Response categories ranged from 0 = *very satisfied*, to 3 = very *unsatisfied* for all items.

Potential effect modifiers included place of residence (either living in San Juan, Puerto Rico, or the Bronx, New York), youth support from parents, youth self-esteem, and youth coping. Place of residence was based on the interviewer's report in the Profile section of the child questionnaire. Place of residence was selected because it has been found to influence patterns of internalizing disorders (178). Ramos et al. found that youth in New York had higher levels of internalizing symptoms, including depressive symptoms compared to youth living in Puerto Rico. In the study, the differences were accounted for by differences in the experience of discrimination and exposure to violence between the two contexts. Youth Support from Parents was a count derived from the Parent/Child Relationship scale of the child questionnaire and was measured at waves 1 and 2. The Parent/Child Relationship scale consisted of 12 items (e.g. "How often do your parents/caretakers help you make important decisions?") and had good reliability when calculated for this study (Cronbach's $\alpha = .74$, at wave 1). Responses consisted of 2 categories, where 0 = rarely or never and 1 = sometimes or often for all items. Based on the sample distribution, Youth support from parents was categorized into three levels (low= <9 items, moderate= 9-11 items, high= >11 items). Youth ways of coping was only measured at wave 1 (trait coping) and dichotomized (low = <3 items, and high = ≥ 3 items) based on the weighted sample mean (3.28 items) and was assessed through the Ways of Coping scale of the child questionnaire (179). The variable consisted of the count of 5 items such as "When you have a problem... you do what you have to do to solve it?" Responses in the child questionnaire consisted of two categories, where 0 =rarely or never and 2 = sometimes or often (the latter category was recoded as 1 for the purpose of the analyses). We calculated Cronbach's alphas for several combinations of the Ways of Coping items but dropping items did not improve the coefficients and a decision was made to use the full scale. The full scale had low reliability (Cronbach's α = .45, at wave 1) possibly because the scale was not developed for Latino populations and may not be as reliable for that population (28). However, we retained this scale because it has a theoretical base, and results can be compared with other Boricua Youth Study research (180). Youth self-esteem was assessed through the Self Esteem scale of the child questionnaire (204). Youth self-esteem was based on a count of eight items provided by the self-perception profile which included five domains: school and athletic competence,

social acceptance, behavior, and physical appearance. The scores for the Self Esteem scale ranged from 0 to 8, with 8 representing the highest score of self-esteem. Responses to the items in the scale consisted of two categories where 0 = no and 2 = yes (the latter category was recoded as 1 for the purpose of the analyses). The reliability of the Self Esteem scale was calculated by Bird et al. (139) and found to be 0.46. Despite the low reliability, the scale was kept because it has a theoretical base and results can be compared with other Boricua Youth Study research.

Data Analysis

Descriptive statistics were calculated to check for missing values, outliers and errors and to provide a quantitative description of the study population in terms of family and individual socio-demographic variables, and resources variables. Chi-square tests and *t*-tests were performed to examine categorical and continuous variables by place of residence. Multiplicative effect modification was tested through the addition of an interaction term between the main effect and the potential effect modifier to the model. Although place of residence was not a significant effect modifier, it was used because in previous studies has been found to have a mental health impact (124). Potential confounders were identified by conducting a bivariate analysis to assess their association with 'total', 'negative', or 'positive or no effect' life events and depressive symptoms (conditional on exposure) at the alpha=0.10 level (184, 185). Once identified as potential confounders, the variables were further tested by adding them one at a time to the unadjusted model between depressive symptoms, and total life events or negative life events. A 5% change in the main effect estimates between the unadjusted and adjusted

models further confirmed the variable as a confounder and the variable was added to the final model. Multicollinearity between variables was assessed with tolerance scores using the standard cutoff of 0.1. All tolerance values were found to be in the acceptable range between 0.97 and 0.99. Missing data were not imputed because the missing observations constituted less than 10% of the analytic sample and were dropped from the analytic sample.

Two separate models were assessed between 'total', 'negative', and 'positive or no effect' life events at wave 1 and depressive symptoms at wave 2; and between 'total', 'negative', and 'positive or no effect' life events at wave 2 and depressive symptoms at wave 3. The two models were tested through regressions and when found to be similar (Model 1: b=0.21, p<0.0001; Model 2: b= 0.22, p<0.0001) they were combined into one model. Log-linear Poisson regression models were used to examine the association between the number of 'total', 'negative' or 'positive or no effect' life events in the previous 12 months at waves 1 or 2, and reports of depressive symptoms at subsequent waves (waves 2 or 3 respectively) among those with no depressive symptoms in the previous 12 months at wave 1. If life events were measured at wave 1, then the socio-demographic and individual resource variables used in the analyses were also measured at wave 2 with the exception of coping which was only measured at wave 1.

SAS® software Version 9.3 (SAS Institute Inc., Cary, NC, USA) was used to perform variable selection. SUDAANTM software (release 11, 2011) (183) was used to adjust standard errors for correlations resulting from the complex multistage sampling (observations nested within individuals, siblings nested within households, and households nested within census blocks). Weights were used in all analyses to adjust for the difference in probability of selection and to represent the age and gender distribution of the 2000 census. All p-values were considered significant at the < 0.05 level.

Results

Socio-demographic characteristics of children without depressive symptoms at wave 1 by place of residence indicate that youth support from parents, youth self-esteem, and youth coping were significantly higher in the Bronx than in Puerto Rico. Puerto Rico youth had a greater number of youth living in two parent households, a higher per capita household income, and had mothers with a higher level of education and depression then youth living in the Bronx (Table 3.1). Results for mean depressive symptoms are not shown in Table 3.1 as the analytic sample in wave 1 only includes youth with no depressive symptoms. However, mean depressive symptoms for the sample which includes mostly youth with no depressive symptoms was 0.20 at wave 2 and 0.22 at wave 3. The weighted sample mean of 'positive or no effect' life events (1.33) was higher than mean 'negative' life events (0.93).

Although not shown here, 69% of youth reported at least one 'positive or no effect' event, 41.4% of youth reported at least one 'negative' event, and 80.4% of youth reported

any event. Thirty-eight percent of youth reported at least one 'positive or no effect' event and no 'negative' events, while only 9% reported at least one 'negative' event and no 'positive or no effect' events. Close to 25% of youth reported at least one 'positive or no effect' and at least one 'negative' event. The same life events could be interpreted as negative or positive. The top five 'negative' events were: lost a close friend/a close friend moved away, someone in the family got arrested, someone in the family went to jail, someone in the family you loved died, and your pet died. The top five 'positive or no effect' events (children answered that the event was "mostly good" or had "affected them not at all or only a little") were: you started a new school, lost a close friend/a close friend moved away, you broke up with a boy/girlfriend, someone in the family you loved died, and your pet died. The top 5 combinations of 'negative' and 'positive or no effect' events were: 1) someone in the family you loved died (negative) and lost a close friend/a close friend moved away (positive or no effect); 2) someone in the family you loved died (negative) and your pet died (positive or no effect); 3) your pet died (negative) and lost a close friend/a close friend moved away (positive or no effect); 4) your pet died (negative) and you started a new school (positive or no effect); and 5) lost a close friend/a close friend moved away (negative) and your pet died (positive or no effect).

Socio-demographic and family characteristics for children included in this study were compared to those children who were excluded because they had depressive symptoms at wave 1 (n=155). Excluded children were compared to included children using chi-square and *t*-tests. Excluded children were more likely to live in a two parent/other guardian household (p<0.0001), or with mothers who had attained a higher educational level

(p<0.0001), and/or mothers who had been diagnosed with depression (p<0.0001) compared to children who were included in the study. Merikangas et al. found some similar socio-demographic characteristics: rates of mood disorders were higher for children living with currently married or cohabitating parents (11). In addition, studies show that the risk for youth depression is the result of a combination of multiple family factors, including depression (205).

The percent of youth with depressive symptoms increased from 0% at wave 1 (no depressive symptoms at wave 1) to 7.0% in wave 2 and 6.6% in wave 3 (not shown). Of the 107 youth who reported depressive symptoms in waves 2 and/or 3, 19 (17.7%) reported depressive symptoms at both waves 2 and 3 (persistent symptoms), 46 (43.0%) reported depressive symptoms only at wave 2, and 42 (39.3%) reported depressive symptoms only at wave 3 (intermittent symptoms). The mean number of total life events or negative life events show a significant decrease between both waves 1 and 2 (p<0.0001), and waves 2 and 3 (p<0.0001).

Although youth coping, parent coping, and youth support from parents were initially identified as confounders, they lost significance when tested for their effect on the association between life events and depressive symptoms, and were therefore removed from the final models. Place of residence, youth support from parents, youth self-esteem and youth coping were tested as effect modifiers but had no multiplicative interaction effects on the association between life events and depressive symptoms. Table 3.2 shows that a unit increase in 'total' life events [b(SE)=0.20(0.05), p<0.0001] resulted in a 22.1%

increase in depressive symptoms, a unit increase in 'negative' life events [b(SE)=0.22(0.07), p<0.0001] resulted in a 24.6% increase in depressive symptoms, and a unit increase in 'positive or no effect' life events [b(SE)=0.26(0.09), p<0.0001] resulted in a 29.7% increase in depressive symptoms. Percentage increases were calculated by exponentiating the beta coefficient for total, 'negative' or 'positive or no effect' life events.

Although youth coping was not a significant effect modifier in the multiplicative scale, we included figures to show that additive interaction between life events and youth coping may be present, although youth coping does not show a multiplicative interaction effect. Figures 3.1a and 3.1b show a comparison of mean depressive symptoms and 'total', 'negative', and 'positive or no effect' life events by level of youth coping for youth 10-13 years old with no depressive symptoms at wave 1. In Figure 3.1a (events at wave 1, depressive symptoms at wave 2), youth with low coping had consistently higher numbers of depressive symptoms than youth with high coping regardless of the number of 'total' life events, 'negative' life events, or 'positive or no effect' life events. In Figure 3.1b (events at wave 2, depressive symptoms at wave 3), there is no such pattern between life events and youth coping.

Discussion

Our results show that depressive symptoms increase with the number of total and negative life events which support other findings in the literature (98, 199, 206-209). The increase in depressive symptoms between childhood and adolescence (37) is concurrent

to an increase in exposure to new experiences (196). As youth grow older they will undergo significant physical, social, and cognitive changes, gain autonomy, and shift their orientation from their parents to outside the family children (195, 210). The effect of negative life events on depressive symptoms was higher than the effect of number of total life events. This result is similar to previous research that found negative life events are more likely to result in depressive symptoms than any other life event (72, 188). However, we did not observe the strong association between negative life events and depressive symptoms described in previous studies in both adult and adolescent populations (211, 212).

Our data suggest that 'positive or no effect' life events (as appraised by children when asked if the event was 'mostly good' or had 'affected them not at all or only a little') result in an increase in depressive symptoms. Most studies report that positive events mitigate the effects of life stresses (164, 213). Our conflicting results may be explained by the percentage of youth in our sample who had reported at least one 'positive or no effect' life event including the deaths of a family member and/or a pet. In the literature, loss events have been consistently reported as risk factors for depression (98, 214). We also found that the youth who reported only 'positive or no effect' events had a lower self-esteem than youth who reported negative and total life events. Fresco et al. (2006) show that an external, unstable and specific attributional style for positive events was related to a high level of depression (215).

Although the increases in depressive symptoms are modest, over time they get compounded and may contribute to later depression. Our findings confirm research that conceptualizes depressive disorders as a continuum that does not depend on meeting a specific diagnostic threshold (48). Although the standardization of diagnostic criteria allows for the classification of distinct subtypes of depression dividing youth into cases and non-cases of depression misses the number of individuals who do not meet the criteria for depression but have relevant depressive symptoms (216). Adolescents showing sub-clinical depressive symptoms (e.g. youth with depressive symptoms who do not meet the criteria for a depressive disorder) show higher rates of early adult depression (32,136) as well as a similar risk profile to those who meet the criteria for a depressive disorder (95). Depressive symptoms have been shown to be risk factors for various depressive disorders other than major depression. An example is the study conducted by Gonzalez-Tejera et al. (2005) which shows that a large proportion of youth with symptoms of depression may not meet the criteria for major depression but instead meet the criteria for minor depression (at least two symptoms but less than five).

Contrary to previous findings, youth support from parents, youth coping, youth selfesteem, and place of residence did not show multiplicative interaction in the association between life events and depressive symptoms (109, 217, 218). However, we observed that mean depressive symptoms for youth with low coping and at least three total or negative life events was at least 10 times higher than the mean depressive symptoms for youth with high coping and no life events. Studies have shown an association between stressful life events and depression. Coping influences the response to stressors (109),

and the way an individual copes with a situation will determine the way the individual is affected by stress (25, 110, 111).

The strengths of this study include three waves of depressive symptom assessments using a widely used psychometrically reliable and valid measure (DISC-IV). To our knowledge, this is the only study on the longitudinal association between life events and depressive symptoms among Puerto Rican youth. Lastly, the statistical software SUDAAN is ideal for the analysis of correlated data encountered in complex survey design.

Several limitations are identified for consideration: First, 'number of depressive symptoms is not a diagnosis of depression. Second, causality between life events and depressive symptoms cannot be established with certainty because each can modify the reporting of the other (219). Third, adolescents with depression may be biased toward recalling or reporting more life events and as a result may artificially inflate the association (166). Fourth, children excluded from the study were found to have higher mean depressive symptoms and higher mean life events. Finally, SUDAAN is not recommended for Poisson distributions with an overdispersion of negative answers as it results in larger standard errors and lower statistical power. Our results may therefore be overly conservative.

Future studies with this data set should assess the longitudinal association of life events and depressive symptoms among youth who experience intermittent symptoms and youth

who experience persistent symptoms. Depressive symptoms should also be examined empirically as a categorical variable based on the number of depressive symptoms a standard deviation above or below the median and mean number of symptoms. Another study could follow-up youth to examine the risk factors of those who develop depressive disorders in young adulthood. Further research into 'no effect' and 'positive only' life events should be conducted to better understand their effects on depressive symptoms. Lastly, future studies should consider specific types of negative life events such as exposure to violence and victimization which have been found to be associated with internalizing disorders (124), and child maltreatment.

Given that the number of depressive symptoms in adolescence has been found to predict depression later in life, it is important to identify early predictors in order to develop interventions that may mitigate the impact of depression before it becomes a chronic disorder. This study identifies 'total', 'negative', and 'positive or no effect' life events as risk factors that may be considered in early interventions before depression develops. Particular attention should be given to youth who report 'positive or no effect' events. In addition, youth with low coping appear to be at particularly high risk of developing depressive symptoms following life events.

Tables

Table 3.1. Socio-Demographic Characteristics, and Individu	al and Family Resour	ces at Wave 1 for 10	0-13 Year Old Youth withou
Depressive Symptoms by Geographic Area (weighted)			

	Total	Bronx	Puerto Rico	
	(n=1,068)	(n=509)	(n=559)	p-values
Number of Total Life Events (Mean, SE)	2.26 (0.06)	2.30 (0.09)	2.10 (0.08)	0.0430^{a}
Number of Negative Live Events (Mean, SE)	0.93 (0.04)	0.95 (0.06)	0.86 (0.05)	0.2693 ^a
Number of Positive or No Effects Life Events (Mean, SE)	1.33 (0.04)	1.36 (0.06)	1.20 (0.06)	0.0491 ^a
Male (%)	51.53	51.6	51.52	0.5873 ^b
Age in Years (Mean)	11.44 (0.03)	11.44 (0.05)	11.46 (0.05)	0.1037 ^a
Age Distribution (%)	× ,			<0.0001 ^b
10	27.82	28.11	26.16	
11	24.74	24.61	25.49	
12	22.67	22.39	24.22	
13	24.77	25.88	24.14	
Household Day Conita Income (Mean SE)	3906	3810	4432	0 0250 ^a
Household Fer Capita Income (Mean, SE)	(129.2)	(181.5))	(209.6)	0.0230
Mother's Education (%)				<0.0001 ^b
Less than High School	42.16	45.63	22.86	
High School Diploma/GED	43.33	43.47	42.53	
More than High School	14.51	10.9	34.61	
Family Composition (Number of Parents Living in Household) (%)				< 0.0001 ^b

Two Biological/Other Parent	54.86	51.99	70.98	
One Biological/Other Parent	45.14	48.01	29.02	
Youth Support from Parents (Mean, SE)	3.28 (0.07)	3.32 (0.09)	3.09 (0.10)	0.0001 ^a
Low (≤ 9)	15.2	14.1	21.1	< 0.0001 ^b
Medium (9-11)	42.6	42.1	45.3	
High (> 11)	42.6	43.8	33.6	
Youth Coping	3.28 (0.04)	3.32 (0.06)	3.09 (00.04)	< 0.0001 ^b
Low (< 3)	22.5	22.7	21.3	
High (≥ 3)	77.5	77.3	78.7	
Youth Self Esteem (Mean, SE)	6.2 (0.04)	6.3 (0.05)	5.9 (0.06)	<0.0001 ^a
Parental Received Social Support (Mean, SE)	5.15 (0.08)	4.28 (0.12)	5.86 (0.10)	<0.0001 ^a
Parental Perceived Social Support (Mean, SE)	2.53 (0.02)	2.39 (0.03)	2.61 (0.03)	<0.0001 ^a
Parental Coping (Mean, SE)	11.83 (0.09)	11.84 (0.13)	11.77 (0.12)	0.6599 ^a
Diagnosed Maternal Depression (%)	3.32	2.98	5.2	< 0.0001 ^b

^a*t*-test ^b Chi-square test

Table 3.2. Log-Linear Poisson Regressions of Number of Total Life Events, Negative Life Events, or Positive or No Effects Life Events in the last 12 Months (at waves 1 or 2) on Depressive Symptoms (at waves 2 or 3 respectively)

	Unadjusted Models			Final Models ^a		
	b	SE	p- value	b	SE	p-value
Number of Total Life Events				0.20	0.05	0.0001
Family Composition (ref=1	0.22	0.24	0.2126			
parents household)	0.52	0.24	0.2120			
Parents Social Support	0.10	0.05	0.0733			
Youth Support from Parents						
<9	ref.	ref.	ref.			
9-10	-0.42	0.33	0.1895			
≥11	-0.54	0.34	0.1153			
Youth Coping						
<3	ref.	ref.	ref.			
≥ 3	-0.38	0.13	0.1600			
Number of Negative Life				0.22	0.07	0 0037
Events				0.22	0.07	0.0037
Parents Received Social	0.11	0.05	0.0596			
Support	0.11	0.05	0.0570			
Youth Support from Parents						
<9	ref.	ref.	ref.			
9-10	-0.45	0.33	0.1818			
≥11	-0.58	0.34	0.0951			
Youth Coping						
<3	ref.	ref.	ref.			
≥ 3	-0.38	0.27	0.1626			
Number of Positive or No				0.26	0.09	0 0005
Effects Life Events				0.20	0.07	0.0002
Parents Received Social	0.08	0.05	0.0949			
Support						
Youth Support from Parents	C	C	c			
<9	ref.	ref.	ref.			
9-10	-0.45	0.35	0.1824			
<u>≥</u>]]	-0.38	0.35	0.1073			
Youth Coping	2	0	2			
<3	ref.	ref.	ref.			
≥ 3	-0.38	0.26	0.1440			

^a Neither confounders nor interaction terms were significant and were therefore not added to the final model





Figure 3.1b. Mean Depressive Symptoms (wave 3) and Number of Total, Negative, and Positive or No Effects Life Events by Level of Youth Coping at (wave 2) for Youth 10-13 Years with No Depressive Symptoms at Wave1



Chapter 4 – Association of Separation, Death, and Social Adversity with Depressive Symptoms among Puerto Rican Youth

This manuscript follows the authors' instructions for the Journal of Clinical Child and Adolescent Psychology at http://www.tandfonline.com/toc/hcap20/current#.Uw-rgONdWVN

Abstract

Objective: This study aimed to examine the association between four types of life events (family environment, separation, social adversity, and death) and the development of depressive symptoms among Puerto Rican youth, and the potential modifying role of place of residence, youth support from parents, and youth coping.

Method: A secondary analysis was performed using three waves (2000-2004) of the Boricua Youth Study interview data for 10-13 year old Puerto Rican youth residing in New York and Puerto Rico with no depressive symptoms at baseline (n=977).

Results: Key findings were: 1) depressive symptoms increase across waves with an increase in social adversity, separation, and death events; 2) this pattern is seen regardless of place of residence, youth support from parents, youth self-esteem, and youth coping for social adversity and separation events; 3) youth support from parents confounded the association between death events and depressive symptoms; 4) youth experiencing any type of event had higher mean depressive symptoms than youth not experiencing any type of event.

Conclusions: Death, family environment, and social adversity life events are potentially relevant to use in conjunction with other screening tools to identify youth at risk of developing depressive symptoms.

Introduction

Depression is a rare psychiatric disorder in childhood but becomes one of the most common disorders in adolescence (8). There are few studies on adolescent depression, and the mechanisms by which only some adolescents develop depression while others do not are not well understood (134). Despite the growth of the Latino population by 43% between 2000 and 2010 (50), Latino youth remain relatively under-represented in mental health research (55), and little is known about the determinants of Latino youth depression by Latino subgroup (220). Puerto Ricans living in the U.S. are the second largest group of Latinos (4.6 million), representing 9% of the Latino population (50). Of all the Latino subgroups, adult Puerto Ricans are more likely to face social disruption (e.g. high alcohol consumption, divorce rates, criminality, and school dropout rates) and a lower socioeconomic status (38). These social characteristics have been shown to threaten family stability and place youth at an increased risk for developing depression (59).

Research shows that early onset depression in adolescence has severe social and health consequences later in life (30). Early onset depression may result in severe chronic depression and poorer psychosocial outcomes in young adulthood (221), and in high recurrence rates in adulthood (40-60% recurrence), (222). In addition, as adolescents grow and gain more autonomy, they are increasingly exposed to potentially disruptive events which may also lead to increases in rates of depression (97-100, 210).

Several checklists were developed to assess life events (e.g. Life Events Checklist, Life Experience Survey, Stressful Life Events Schedule) (157, 223, 224) however, the number and nature of events included in the checklists varied depending on the focus of the research that was being conducted. Critics argued that not all life events had the same weight and that some events were more stressful or had more impact in some populations (225). Continuing interest in the etiology of mental illness led researchers to focus on the impact of certain types of life events (209). Aggregating life events allowed researchers to further study the association between life events and specific mental disorders. (102, 194, 212, 226, 227).

Approaches to grouping life events have included identifying events that are stressful (228); severe or threatening (229); desirable or undesirable (230); acute or chronic (231); or used multiple categories (e.g. loss, death, autonomy, deviance, accident/illness, health, relocation events) (212, 232). Despite the lack of agreement about the nature and definition of categories of life events (233), researchers found that some types of life events were associated with the development of specific mental disorders (197, 227, 234).

Stressful, severe or threatening life events have been significantly associated with depression in several studies (229). However, although most individuals will be exposed to traumatic events and losses in their lifetime, only some will develop depression while the majority will recover and move ahead (235). The mechanisms by which stressful events affect individuals are not fully understood (236). An explanation for differences in

response to stressful events is that resilient individuals have been found to overcome challenges by mobilizing personal and social resources to cope with stress and adversity (237).

The main objective of this study is to identify and compare types of life events that are significantly associated with depressive symptoms among 10-13 year old Puerto Rican youth. This study seeks to examine different types of life events that lead families to physical, psychological or interpersonal crises, as postulated in Hill's Theory of Family Stress (170). Our research questions aim to find if certain types of events result in greater number of depressive symptoms, and whether the resolution of crises were modified by place of residence, youth coping or social support (25, 110, 111, 116).

Methods

This study uses data collected in three annual waves between summer 2000 and fall 2004 of the Boricua Youth Study (BYS). The BYS is a longitudinal study of child and parent/caregiver data on psychiatric disorders that uses multistage probability samples to represent 5-13 year old Puerto Rican children living in the South Bronx, New York and San Juan, Puerto Rico according to the 2000 U.S. census. Eligible households included at least one5 to 13 year old child residing in the household at the time of the census, with at least a parent living in the household who was of Puerto Rican background. Up to three children were selected among households with multiple children. Trained lay interviewers conducted home interviews at the same time but separately with children

and parents/caregivers in Spanish or English based on language preference. The BYS includes data on family socio-demographic characteristics, child psychopathology and other risk factors for mental disorders. Parents provided signed informed consent, while children 7 or older provided signed informed assent. The attrition rate was 7.95% at wave 2, and 11.88% at wave 3. For a more detailed description of the study's background, design and methods refer to work conducted by Bird et al. (2006). Institutional Review Board approval for this project was obtained both from the University of Puerto Rico and the University of Maryland College Park.

The analytic sample included 977 Puerto Rican children 10-13 years old who met the following conditions: 1) no depressive symptoms at wave 1; 2) participated in both waves 1 and 2; 3) had complete life events data for both waves 1 and 2; and 4) had complete depressive symptoms data at waves 2 and 3. Excluded from the analytic sample were youth with depressive symptoms at wave 1 (n=155) or youth missing life events information at wave 1 or wave 2 (n=5). Children 5-9 years of age were excluded from the analysis because the reliability of the instrument measuring psychiatric disorders in this population is poor (139). We excluded children with depressive symptoms at wave 1 in order to establish a temporal sequence between life events at waves 1 and 2 and the development of depressive symptoms at waves 2 and 3.

Outcome variable – Depressive symptoms (reported at waves 2 and 3) was the count of symptoms experienced in the past 12 months. We chose to use the 9 questions that have been shown to accurately predict depression (149). Depressive symptoms are derived

from the National Institute of Mental Health Diagnostic Interview Schedule for Children, Version IV (DISC-IV) Major Depression Schedule (142) and has been recommended to be used as a standardized tool for quick evaluation of children and adolescent diagnoses. Lucas et al. (2001) calculated the reliability to be good (0.82). Although depression has its onset in adolescence, we chose to focus on the count of depressive symptoms because our population was too young to rely on diagnosable depression. Depressive symptoms were a combined measure of responses provided by children and their primary parents/caregivers using the DISC-IV(142). The English version of the DISC-IV is a commonly used instrument to ascertain a wide range of child and adolescent psychiatric diagnoses. DISC-IV is only one of four instruments that have been translated into Spanish and only one of two instruments with psychometric data available for U.S. Latinos (151). The moderate test-retest reliability of the Spanish version was found to be similar to the English version (κ =0.48) (143). We used a combination of children and parent reports of depressive symptoms in order to obtain the most reliable information (143).

Exposure Variable – The four types of life events used in this study (separation, death, family environment, social adversity) were adapted from the research on types of life events conducted by Grover et al. (2005). The types of events included: 1) separation events (move to a new home, start a new school, parents separate, parental divorce, you break up with a girl/boyfriend, a close friend moved far away, a family member went to jail), 2) death events (your pet died, a close friend died, a close family member died), 3) social adversity events (you were victim of a crime, you saw a crime, a family member

was arrested, a family member has an alcohol/drug problem), and 4) family environment events (parents argue more, have a new brother/sister, have a new stepfather/stepmother, you got sick or injured, a family member has a mental or emotional problem, family member was sick or injured). See Table 4.2 for the complete list of types of life events. Each type of life event was an aggregate of positive responses to the life events included in that category. The life events used in this study were derived from a list of 46 items from the Life Events Checklist (157).

Potential Confounders and Effect Modifiers – Potential confounding covariates were identified through a literature review on types of life events and depression, and were based on risk and protective factors described in Hill's Family Stress Theory (170). Socio-demographic variables included: age (integer in years), gender (reported by the interviewer), place of residence (San Juan, Puerto Rico; and South Bronx, New York), family composition (single or two biological/other parent family), per capita household income (calculated by dividing household income by the number of household residents), and parental education (less than high school, high school, more than high school).

Parent/caregiver **received** social support was based on the count of positive responses to three out of 15 items selected from the Parent Social Support scale. We calculated the internal consistency for different combinations of items and selected the items that yielded the highest reliability coefficient (Cronbach's $\alpha = .70$, at wave 1). The items focused on how often a parent gets together with family members, how often parents attend family gatherings, and how often family members take care of the respondent's

children. Similar concepts on received social support can be found in the literature (128, 175, 176). Responses ranged from 0 = never/once a year or less, to 3 = at least once a week for the three items.

One item was selected from the Parent Social Support scale to measure parent/caregiver **perceived** social support. Responses to the item ranged from 0 = never/once a year or *less*, to 3 = at *least once a week*: "In general how satisfied are you with the amount of support that you receive in your life."

Parent/caregiver coping was a count variable of positive responses to all questions in the Parental Coping scale. The scale consisted of eight items ranging from 0 = almost never to 2 = often for all items such as "Have you felt that you were unable to control the important things in your life?" (reverse coded) and "Have you successfully dealt with irritating life hassles?" The internal reliability of this scale was calculated for this study (Cronbach's $\alpha = .70$, at wave 1) and was found to be in the acceptable range (238).

Place of residence, youth support from parents, and youth coping were analyzed as potential effect modifiers. Place of residence (San Juan, Puerto Rico and the Bronx, New York) was recorded by the interviewer in the Profile section. In a previous study, internalizing disorders were found to vary by place of residence but the differences were accounted for by youth's exposure to violence and experiences of discrimination (178). Aggregated positive responses to the Parent/Child Relationship scale measured youth support from parents. The scale has 12 items and when calculated for this study, it was found to have good reliability (Cronbach's $\alpha = .74$, at wave 1). Responses range from 0 = rarely or never and 1 = sometimes or often for all items. Youth coping was derived from the Ways of Coping scale which consisted of 5 items ranging from 0 = rarely or never to 2 = sometimes or often. The items on the scale were only measured at wave 1 and categorized into two levels (low=<3 items, high=≥3 items) based on the mean (3.28 items). The reliability for several combinations of Ways of Coping items was calculated. The best combination included the full five items and was retained despite its low reliability (Cronbach's $\alpha = .45$, at wave 1) because it is based on theory and will allow the comparison of our results to other Boricua Youth Study studies (180).

Data Analysis

Univariate analyses were conducted to generate descriptive statistics and to check for missing values, errors, and outliers. Bivariate analyses were conducted to identify potential confounders and multiplicative effect modification was tested by adding an interaction term to the model. We assessed the internal consistency of all our study scales by calculating correlations between scale items in the full analytic sample using Cronbach's alpha coefficient. Multicollinearity was tested using tolerance tests with the standard cutoff of 0.1 and tolerance values were found to be in the acceptable range between 0.97 and 0.99. A log-linear Poisson model was used to test he association between the types of life events (social adversity, separation, death, family environment) in the previous 12 months at wave 1 and depressive symptoms at wave 2, and types of life

events at wave 2 and depressive symptoms at wave 3 among youth with no depressive symptoms at wave 1.

Selection of covariates and descriptive statistics were conducted using SASTM Software Version 9.3. Final models were evaluated with SUDAAN® software (release 11, 2012) (183) which is ideal for repeated measures and to adjust for cluster correlations resulting from multistage design. Sampling weights were used in all analyses to adjust for different probabilities of selection and to be representative of the age and gender distribution of the 2000 census. All p-values were considered significant at the < 0.05 level.

Results

Table 4.1 shows mean depressive symptoms at wave 2 for 10-13 year old youth without depressive symptoms at wave 1 by socio-demographic characteristics and types of life events at wave 1. Overall, mean depressive symptoms were higher for youth who experienced any of the four types of life events compared to youth who did not experience any type of life events. Overall, mean youth support from parents, parent **received** social support, parent **perceived** social support, and parent coping was lower for youth who experienced any of the four types of life event compared to children who did not experience any type of life event.

Table 4.2 is a list of life events used in the Boricua Youth Study by type of event. Although not shown here, 63.2% of youth experienced separation events (e.g. move to a

new home, start a new school, parents separate) compared to 41.8% who experienced death events (e.g. your pet died, a close friend died), and 22.6% who experienced social adversity events (e.g. you were victim of a crime, you saw a crime).

Table 4.3 shows that a unit increase in social adversity events [b(SE)=0.72(0.12), p<0.0001] results in a 105% increase in depressive symptoms; a unit increase in family events [b(SE)=0.41(0.03), p0.0037] results in a 50.7% increase in depressive symptoms; a unit increase in separation death events [b(SE)=0.43(0.14), p=0.0033] results in a 53.7% increase in depressive symptoms; and, a unit increase in separation events [b(SE)=0.28(0.08), p0.0013] results in a 32.3% increase in depressive symptoms. Youth support from parents is a significant confounder in the association between all types of events and depressive symptoms. Parent coping was a significant confounder in the association between social adversity events and depressive symptoms. Youth coping, youth support from parents, youth self-esteem, and place of residence were tested as effect modifiers by including an interaction term in the final model but they did not modify the association between types of life events and depressive symptoms.

Figure 1 shows mean depressive symptoms at wave 2 with confidence intervals by types of life events at wave 1 for 10-13 year old youth with no depressive symptoms at wave 1. Mean depressive symptoms were consistently higher for youth who experienced types of events compared to youth who experienced no events. Mean depressive symptoms were highest for social adversity events, followed by separation events, and then death events.

Discussion

Research shows that the experience of life events is significantly associated with depression (37, 206). More specifically, certain types of life events (e.g. severe loss, interpersonal trauma) increase the likelihood of developing depression (164, 239). Overall, our findings lend support to these studies and as expected, the association between types of events (e.g. social adversity, death, and separation events) and depressive symptoms varied in strength. We found that the percentage of separation events being reported was higher than the percentage of death and social adversity events however the results suggest that although not as frequent, social adversity events had a greater negative impact on psychological well-being.

Our study found that youth support from parents did not modify any of the associations between type of life events and depressive symptoms, but was a significant risk factor. These results support the literature that shows that family and personal resources, and in particular parent's social support have been found to promote youth mental health (240). Youth coping, youth self-esteem, or place of residence were tested as moderators of the association but none of the interaction terms were significant indicating that they were not significant effect modifiers.

There are several strengths to the present study. Depressive symptom are assessed with a measure that has been found to be reliable and valid (DISC-IV) (142). To our knowledge there are no other longitudinal studies of the association between types of life events and

depressive symptoms among Puerto Rican youth. Looking at different types of life events provides an opportunity to understand how each type of life event impacts depressive symptoms (102, 241).

Several limitations are identified for this study. First, although the reliability of depressive symptoms was found to be substantial in this study (κ =0.64) (142), depressive symptoms do not constitute a diagnosis of depression. Second, it is difficult to establish causality between types of life events and depressive symptoms because each can modify the reporting of the other (219). Third, the recall of certain types of events may decrease at different rates over time, and recall is dependent on the event (159, 160). Fourth, we do not know which combinations of types of life events are experienced by each individual so it is possible that more than one type of life event may be experienced by the same individual (231). In addition, each type of life event may have included both positive and negative life events and therefore their effect on depressive symptoms may have been weakened. Fifth, children excluded from the study had more depressive symptoms and experienced more life events. Last, in SUDAAN the overdispersion of negative answers results in larger standard errors and results are more conservative

Future studies may examine the extent to which single life events within types of life events (e.g. death of a family member, death of a close friend, death of a pet) contribute to the association of type of life event with depressive symptoms. Likewise, studies may examine different ways of categorizing life events and how these categories affect depressive symptoms. A 'type of life event' variable that includes four levels (i.e. 'social
adversity events only', 'death events only', 'separation events only', and 'family environment events only') should be created to allow comparison of the magnitude of effect between types of life events and depressive symptoms. Last, future studies should further explore possible mechanisms for developing depressive symptoms by type of life events.

The identification of risk and protective factors that contribute to the early development of depressive symptoms is important since early onset depression has lifelong economic, social and occupational outcomes (242). Relying on standard diagnostic tools are ideal to identify youth meeting the criteria for a diagnosis of depression but knowledge of a diagnosis is not enough to detect youth who present with sub-clinical levels of depression (do not meet the criteria for a diagnosis) (95). Youth with sub-clinical levels of depression will not get treated and are at increased risk of developing depression later in life (194, 243). Using symptoms from the DISC Predictive Scale for depression coupled with additional measures such as reports of life events may prove useful in predicting the course of depression. This study shows that separation, death, and social adversity types of life events are useful predictors of depressive symptoms among 10-13 year old Puerto Rican youth and that youth support from parents is a significant protective factor in the association between types of life events and depressive symptoms. These findings may be useful to consider when trying to identify youth at risk of developing depression.

Tables

Table 4.1. Mean Depressive Symptoms (wave 2) by Socio-Demographic Characteristics (wave 1) and Type of Life Event (wave 1) for 10-13 Year Old Youth without Depressive Symptoms at Wave 1 (weighted)

	Total	No Social Adversity Events	Social Adversity Events (1-4)	No Separation Events	Separation Events (1-7)	No Death Events	Death Events (1-3)	No Family Environment Events	Family Environment Events (1-6)
	(n=1,044)	(n=784)	(n=260)	(n=420)	(n=624)	(n=553)	(n=491)	(n=757)	(n=287)
Bronx		0.19	0.22	0.17	0.33	0.18	0.43	0.21	0.45
Puerto Rico		0.58	0.76	0.17	0.49	0.32	0.36	0.27	0.58
Female		0.20	0.69	0.27	0.37	0.28	0.40	0.28	0.44
Male		0.17	0.52	0.08	0.34	0.13	0.44	0.17	0.51
Age Distribution (%)									
10		0.00	0.66	0.00	0.10	0.06	0.07	0.07	0.01
11		0.27	0.30	0.12	0.36	0.13	0.41	0.19	0.46
12		0.19	0.43	0.11	0.27	0.25	0.21	0.18	0.39
13		0.10	0.73	0.26	0.43	0.23	0.33	0.25	0.32
Mother's Education (%)									
School		0.25	0.38	0.08	0.37	0.15	0.44	0.18	0.66
High School Diploma/GED		0.15	0.62	0.17	0.33	0.21	0.37	0.21	0.44
More than High School		0.15	0.88	0.40	0.32	0.28	0.44	0.32	0.42
Family									
Composition (Number of									

Parents Living in								
Household) (%)								
Two								
Biological/Other	0.17	0.68	0.20	0.34	0.18	0.44	0.21	0.47
Parent								
One								
Biological/Other	0.22	0.54	0.14	0.37	0.23	0.40	0.24	0.48
Parent								
Diagnosed								
Maternal								
Depression (%)								
Yes	0.44	1.07	0.19	0.75	0.69	0.58	0.23	0.39
No	0.18	0.58	0.17	0.33	0.18	0.41	0.02	1.80

Table 4.2. List of Life Events used by the Boricua Youth Study Categorized by Type of Life Events

Social Adversity

During the last 12 months were you the **victim** of a crime, a violent act, or assault? During the last 12 months, did you **see a crime or accident** where someone was mugged, hurt or killed?

During the last 12 months, was **someone in your family arrested?** During the last 12 months, did a **family member have a drug or alcohol problem** (not including you)?

Separation

During the last 12 months, did you **move to a new home** (permanently, not a temporary residence that is not your home)?

During the last 12 months, did you start going to a new school?

During the last 12 months, did your parents separate?

In the last 12 months, did your **parents get divorced**?

During the last 12 months, did you **break up** with a girlfriend or boyfriend?

During the last 12 months, did you **lose a close friend** not by death, or did a **close friend move far away?**

During the last 12 months, did someone in your family go to jail?

Death

During the last 12 months, did a close friend die?

During the last 12 months, did someone in the family that you loved a lot die?

During the last 12 months, did a pet of yours die?

Family Environment

During the last 12 months, did your **parents argue more than previously?** In the last 12 months, did **a new brother or sister arrive in your home** (because one was born, or one moved into the household)?

During the last 12 months, did a new **stepmother or stepfather move into your house?** During the last 12 months, did you get **seriously sick or injured**?

During the last 12 months, did a **family member** (not including you) **have a mental or emotional problem?**

During the past 12 months, was someone in your family seriously sick or injured?

Table 4.3. Log-Linear Regressions of Separation, Death, and Social Adversity Events in the Last 12 Months and Depressive Symptoms

	Unadjusted Model			Adjusted Model		
	b	SE	p-value	b	SE	p-value
Separation Events	0.28	0.08	0.0013	0.27	0.08	0.0022
Youth Support from Parents				-0.11	0.04	0.0020
Death Events	0.43	0.14	0.0033	0.43	0.14	0.003
Youth Support from Parents				-0.12	0.04	0.0017
Social Adversity Events	0.72	0.12	<0.0001	0.68	0.16	<0.0001
Youth Support from Parents				-0.1	0.04	0.0087
Parent Coping				-0.09	0.04	0.0132
Family Events	0.41	0.14	0.0037	0.39	0.14	0.0068
Youth Support from Parents				-0.11	0.04	0.0038



Figure 1. Mean Depressive Symptoms (wave 2) by Type of Life Event (wave 1) for 10-13 Year Old Youth with No Depressive Symptoms at Wave 1

Chapter 5 – Association of Child Maltreatment with Depressive Symptoms among Puerto Rican Youth

This manuscript follows the authors' instructions for the journal Child Abuse and Neglect at http://www.journals.elsevier.com/child-abuse-and-neglect/

Abstract

The experience of single and multiple types of child maltreatment has been strongly associated with depression. However, little is known about maltreatment and depressive symptoms among Puerto Rican youth as the result of using broadcategories to describe Latinos. This study examines the predictive strength of different types of lifetime child maltreatment (i.e. physical, sexual, and emotional abuse, and neglect), and the effect of youth support from parents, youth coping, youth self-esteem, and place of residence on depressive symptoms among Puerto Rican youth. Secondary data analyses were performed using three annual waves (2000-2004) of data from the Boricua Youth Study. The sample consists of 855 10-13 year old Puerto Rican youth living in New York and Puerto Rico with no depressive symptoms at wave 1. Our results indicate that close to one in four children experienced some type of lifetime maltreatment. Physical abuse only was the most common type of maltreatment (12.0%), followed by neglect only (4.6%), sexual abuse only (3.8%), and multiple maltreatment (3.4%). The mean depressive symptoms for youth who experienced multiple maltreatment was 1.85 higher than the mean depressive symptoms for children who experienced no maltreatment. We found no association between different types of lifetime child maltreatment at wave 1 and depressive symptoms at wave 3.

Introduction

Depression is one of the most common mental disorders in adolescence with serious consequences which extend into adulthood. Adolescent depression is strongly associated with psychiatric comorbidity, school failure, social difficulties, the adoption of risky behaviors, obesity, physical impairment and suicide (5, 14, 16). Early onset depression often leads to the recurrence of depression, severe depressive disorders and major functional impairment in adulthood (9, 20). In turn, adult depression has been linked to high personal, social and economic costs causing a substantial loss in human capital (\$57.5 billion total expenditure for mental disorders in 2006) (3, 21, 23, 244).

Depression has been strongly associated with the experience of lifetime child maltreatment (82). Maltreatment (i.e. physical, sexual, and emotional abuse, and neglect) (80) is also a serious public health problem with significant health, social, and economic costs (81). It is estimated that the average lifetime cost of a non-fatal child maltreatment case in the U.S. is approximately \$210,000 in 2010 dollars (81), including productivity losses, and medical, child welfare, criminal justice, and special education costs. Although not all occurrences of child maltreatment are reported, the number of officially reported cases highlights a widespread problem (88). Findings from the Developmental Victimization Survey, a nationally representative sample of children 2-17 years of age, indicate that 1 in 7 children in the U.S. will experience some form of child maltreatment in their lifetime (90). In 2011, child maltreatment rates among Latinos (8.6 per thousand) were comparatively higher than the rates for non-Hispanic Whites (7.9 per thousand), however, Latino cases have been found to be underreported (91).

Most studies of child maltreatment among Latinos do not distinguish specific Latino subgroups (92) with the exception of Puerto Ricans who live in a U.S. territory. As such, Puerto Rico is included in the National Child Abuse and Neglect Data System (NCANDS), a federally sponsored system established in response to the Child Abuse Prevention and Treatment Act of 1988 to collect and analyze annual data on child abuse and neglect (89). Estimates from the National Child Abuse and Neglect Data System indicate that there were 10 per 1,000 child maltreatment victims in Puerto Rico compared to 9.2 per 1,000 in the U.S.

Early research on child maltreatment focused on the effects of single types of maltreatment however more recent studies have shown that children who are exposed to one type of maltreatment are often exposed to other types (93). As a result, researchers started examining the combined effects of different types of child maltreatment and found that exposure to multiple types of maltreatment represent a substantial risk for mental health disorders (94), and major depression in particular (86).

The present study is based on the ecological-transactional model developed by Cicchetti and Lynch (137). In this model, the balance between risk and protective factors present at different ecologic levels (i.e. community, family) interact to influence the course of the child's psychological development and functioning across the life-span. For example,

child maltreatment is believed to disrupt the conditions that lead to normal development, and will result in maladaptive cognitive and behavioral strategies, and psychopathology (88).

This longitudinal study aims to examine a sample of 10-13 year old Puerto Rican youth living in New York and Puerto Rico from the Boricua Youth Study to determine the predictive strength of individual and multiple types of child maltreatment on the development of depressive symptoms. We hypothesize that an increase in individual or multiple types of child maltreatment results in an increase of depressive symptoms, and that the strength of the association varies by type of maltreatment. In addition, we want to test whether ways of coping, self-esteem, youth support from parents, and place of residence have a potential moderating role in the association between child maltreatment and depressive symptoms.

Methods

The study background, design, and methodology are described in detail by Bird et al. (2006). In brief, Boricua Youth Study (BYS) data were collected in three annual waves between summer 2000 and fall 2004. The BYS is a longitudinal study that uses representative community samples of 5-13 year old children at wave 1 at two sites: the Bronx, New York (n= 1,138) and San Juan Puerto Rico (n=1,353). Children and their parents/caretakers were interviewed at wave 1 and re-interviewed at two one-year intervals after that. The three waves of interviews were conducted by trained lay

interviewers in Spanish or English based on language preference. The samples were weighted to adjust for differences in their probability of selection due to the multistage sample design. The BYS provides a wealth of information on family socio-demographic characteristics, child psychopathology and other risk factors for mental disorders.

The final analytic sample included 855 Puerto Rican children 10-13 years old who met the following conditions: 1) had wave 1 data on lifetime child maltreatment; and 2) had wave 3 data on depressive symptoms. Excluded from the analytic sample were youth with depressive symptoms at wave 1 (n=186) or youth missing depressive symptoms at wave 3 (n=122). We excluded children with depressive symptoms at wave 1 in order to establish a temporal sequence between child maltreatment at wave 1 and the development of depressive symptoms at waves 3. Children 5-9 years of age were excluded from the analytic sample because the instrument measuring psychiatric disorders in this age group has poor reliability (139). Institutional Review Board approval for this project was obtained from both the University of Puerto Rico and the University of Maryland College Park.

Outcome variable – Depressive symptoms (reported at wave 3) was the count of symptoms experienced in the past 12 months. The count was the combined measure of responses to 9 questions that have been found to accurately predict depression (with higher scores indicating more depressive symptoms). Responses were provided by both children and their primary parents/caregivers. We concentrated on the count of depressive symptoms among our young population because we had an insufficient number of youth

with diagnosed depression and symptom scales have better test-retest reliability than categorical diagnoses that use a threshold number of criteria for depression (142). Depressive symptoms were derived from the questions found in the major depression and dysthymia schedule of the National Institute of Mental Health Diagnostic Interview Schedule for Children, Version IV (DISC-IV) (142). Both the English and Spanish versions of the DISC-IV are commonly used instruments to ascertain a wide range of child and adolescent psychiatric diagnoses. The Spanish version of DISC-IV is the only one of two instruments with psychometric data available for U.S. Latinos (151). The test-retest reliability for both the Spanish and English versions was found to be moderate (κ =0.48) (143). (149). A combination of children and parent reports was used to obtain the most complete information on depressive symptoms since reliability for last year reports from parents only (κ =0.27) and children only (κ =0.12) was found to be low to fair (143).

Exposure Variable – Lifetime child maltreatment was constructed using the answers to 11 items from the Traumatic Experiences Questionnaire (TEQ) (see Appendix 9). The questionnaire includes self-report items to assess physical and sexual abuse, and neglect by family and/or non-family members. The maltreatment measure was a combination of lifetime experiences of maltreatment assessed at wave 1. Physical abuse was derived from four questions from the child version of the Parental Discipline Scale (66). Physical abuse included having ever been hit by a caregiver with an object sometimes or many times, or ever been hit by a caregiver with a fist, kicked hard, beaten very hard, or purposely injured at least once. Sexual abuse included two items derived from a scale

developed by Finkelhor and Dziuba-Leatherman (169), and was defined as being forced to look at or touch an adult's or older child's private parts, or having an adult or older child ever try to touch, grab, or kiss the child in a sexual way one or more times. Neglect was derived from four items in the Parental Discipline Scale (66): having ever been left alone at least two times, or left without food, necessary medical care, or having a caretaker who could not take care of the child due to substance abuse at least one time. Based on the co-occurrence of different types of maltreatment, the variable was categorized into five levels depending on the type of maltreatment: no maltreatment, neglect only, physical abuse only, sexual abuse only, and multiple maltreatment (2 or more types of maltreatment). This five level maltreatment variable is similar to the variable used by Gonzalez-Tejera et al. (2005), and Sledjeski et al. (2009) in their studies. The reliability of sub-scales included in the TEQ was previously calculated and found to be poor to good: physical abuse (Cronbach's alpha = .70), neglect (Cronbach's alpha = .47), sexual abuse (Cronbach's alpha = .55), and multiple maltreatment (Cronbach's alpha = .72) (95).

Potential Confounders and Effect Modifiers – Potential confounders were identified through a literature review on types of child maltreatment and depression, and through the ecological-transactional model (137). In this study we used various sociodemographic variables assessed at wave 1: age (integer in years), gender (reported by the interviewer), family composition (single or two biological/other parent family), per capita household income (calculated by dividing household income by the number of household

residents), and parental education (less than high school, high school, more than high school).

Parent/caregiver coping was the count of positive responses to questions in the Parental Coping scale. The scale consisted of eight items ranging from 0 = almost never to 2 = often for all items such as "Have you felt that you were unable to control the important things in your life?" (reverse coded) and "Have you successfully dealt with irritating life hassles?" The internal reliability of this scale was calculated for this study (Cronbach's $\alpha = .70$, at wave 1) and was found to be in the acceptable range (238).

Exposure to community violence was based on 11 items from the Exposure to Violence questionnaire (Raia, 1995, Richters and Martinez, 1993). Each item included questions about the types of exposure to community violence: it happened to self, saw it happen to someone, and heard of it happening to someone. The measure was weighted for different levels of exposure, so the response was coded 3 if the youth experienced direct exposure, the response was coded 2 if it happened to someone else, and the response was coded 1 if the youth heard about the violence happening to someone else,. The higher scores indicated higher exposure to violence.

Co-morbidity was a dichotomous variable based on parent reports of mental health disorders and impairment other than depression as determined by the questions from the National Institute of Mental Health Diagnostic Interview Schedule for Children, Version IV (DISC-IV) (142). Co-morbidity included conduct disorder, oppositional defiant disorder, social phobia, separation anxiety, panic disorder, generalized anxiety, posttraumatic stress disorder, alcohol, nicotine and other substances use, and ADHD.

Potential effect modifiers were place of residence, youth support from parents, youth selfesteem, and youth coping. Place of residence had two levels: San Juan, Puerto Rico and the Bronx, New York and was recorded by the interviewer in the Profile section. Sledjeski et al. found that site differences in maltreatment prevalence between Puerto Rico and New York emerged over time specifically for physical abuse while risk factors were the same at both sites (173). In addition, Ramos et al. found higher levels of internalizing symptoms, including depressive symptoms among youth living in New York compared to youth living in Puerto Rico. However, the differences in the level of discrimination and exposure to violence between the two contexts accounted for the results (95). Youth support from parents was derived from a count of positive responses to 12 questions from the Parent/Child Relationship youth questionnaire. The questions included how often the parent/caretakers do things with the child, help make important decisions, take time to talk about things, talk about problems the child had at school or with a friend, play some sport/game, help with school work, go to the movies together, come see the child for a special activity, pay attention to what the child says, the child does what the parent asks them to do, how often the child talks back, how often the child helps parents with a task. We calculated the reliability of the scale and it was acceptable (Cronbach's $\alpha = .74$, at wave 1). Responses ranged from 0 = rarely or never and 1 = 1sometimes or often for all items. Youth coping consisted of positive answers to 5 questions derived from the Ways of Coping scale. Possible responses ranged from 0 =

rarely or never to 2 = sometimes or often. The reliability of this measure was low (Cronbach's $\alpha = .36$, at wave 1) but the measure was kept because it is based on theory and results can be compared to other Boricua Youth Study studies (180). Questions were about school and athletic performance, competence, social acceptance, behavioral conduct and physical appearance. Youth self-esteem was a count to 8 questions from the youth Self Esteem questionnaire. We calculated the reliability of the scale and found it to be good (Cronbach's $\alpha = .90$, at wave 1).

Data Analysis

Descriptive analyses were conducted to assess the distributions of the variables and to check for errors and missing values. GLM procedures were used to examine mean depressive symptoms for socio-demographic variables by youth support from parents. Youth support from parents was analyzed separately as a continuous variable, andas a binary measure based on mean social support: low (<10 items) and high (\geq 10 items) social support for the purpose of presenting results. A power analysis was conducted using STATA/IC 10.0 for the study sample and the actual estimated power output was 1.0 ANCOVA method).

Covariates were identified as potential confounders when they were significantly associated (alpha<0.10) (184, 185) with child maltreatment or depressive symptoms (conditional on exposure). Identified confounders were added one at a time to the unadjusted model for further testing, and if a 5% change was observed in the main effect

they were further confirmed as confounders. Interaction terms were added to the final model to assess multiplicative interaction. Reliability was assessed calculating Cronbach's alpha coefficients for all our study scales in the full analytic sample. Tolerance tests with the standard cutoff of 0.1 were used to assess multicollinearity. Tolerance values were found to be in the acceptable range between 0.97 and 0.99.

We used a log-linear Poisson model to examine the association between lifetime child maltreatment reported in wave 1 and depressive symptoms in wave 3 among youth with no depressive symptoms at wave 1. The covariates used for analyses were reported at wave 1. We first present an unadjusted analysis of the association between child maltreatment and depressive symptoms, an adjusted analysis that includes significant covariates, and an adjusted model that includes significant covariates and the interaction term between lifetime child maltreatment and youth support from parents. All beta coefficients were exponentiated to calculate the effect of each covariate on depressive symptoms.

Descriptive statistics were conducted using SASTM Software Version 9.3. Covariates and final models were evaluated with SUDAAN® software (release 11, 2012) (183) which is ideal to handle repeated measures and cluster correlations resulting from the multistage design. Sampling weights were used in all analyses to adjust for unequal probabilities of selection due to the complex design, and to be representative of the age and gender distribution of the 2000 census. All p-values were considered significant at the < 0.05 level.

Results

Table 5.1 shows weighted results for socio-demographic characteristics and mean depressive symptoms at wave 1 for 10-13 year old Puerto Rican youth who had no depressive symptoms at wave 1. Approximately 24% of children experienced some type of maltreatment during their lifetime. 'Physical abuse only' was the most commonly reported type of abuse (12.0%), followed by 'neglect only' (4.6%), 'sexual abuse only' (3.8%) and 'multiple maltreatment' (3.4%). The proportion of children reporting any child maltreatment was higher for youth who did not have support from parents (28.4%)compared to those who had support from parents (20.6%). Mean depressive symptoms were highest for 'multiple maltreatment'. The mean depressive symptoms for youth who experienced 'multiple maltreatment' was1.85 times the mean depressive symptoms for youth who experienced no maltreatment. Mean depressive symptoms were higher for youth living in a two parent/caregiver household compared to those who lived in single parent/caregiver household, for youth who had depressed mothers compared to youth living with mothers with no depression, and for youth who had a co-morbid mental disorder compared to those who had no co-morbid disorders.

Poisson log-linear regressions were conducted to examine whether different types of child maltreatment predicted follow-up depressive symptoms but no significant association was found for 'neglect only' [b(SE)=-0.57(0.99); p=0.5630], 'physical abuse only' [b(SE)=0.49(0.45); p=0.2778], 'sexual abuse only' [b(SE)=-0.41(0.89); p=0.6448] and' multiple maltreatment' [b(SE)=43(0.95); p=0.6546] (table 5.2).

Discussion

We found that a history of abuse was common in our Puerto Rican youth sample (24%) supporting previous research. However, the prevalence of each type of abuse was lower than in other studies examining an older adolescent population in Puerto Rico (95, 173). As expected, the mean depressive symptoms for youth who experienced 'multiple maltreatment' was the highest.

Our regression results do not show an association between single (neglect only, physical abuse only, and sexual abuse only), multiple maltreatment and depressive symptoms. The lack of association cannot be explained by the lack of power. These findings are surprising since the significant association between physical abuse (245), sexual abuse (246), neglect (82), and depression has been substantially documented. In addition, more recent studies on the co-occurrence of different types of maltreatment have found that multiple maltreatment is strongly linked to depression (247).

We wanted to test the ecological/transactional model to examine if family and personal resources promote youth mental health (240). However, we found no main effects for such resources. Results in the literature talk about the buffering role that social support (218, 248, 249), coping styles (250), and youth self-esteem (251) play on the association between stressful events and depressive symptoms. However, we were unable to find any significant effect moderation.

This study to our knowledge is the first study that attempted to examine the longitudinal association between single and multiple types of child maltreatment and depressive symptoms among Puerto Rican Youth. Study strengths include the use of depressive symptoms which are assessed with the DISC-IV which is a widely used reliable and valid instrument (142). The BYS is a powerful dataset that includes a wide variety of risk factors with adequate psychometric properties. Lastly, SUDAAN is a powerful tool for analyzing repeated measures and cluster-correlated data like in the Boricua Youth Study.

There are several limitations in the study. First, in our model we excluded youth who had depressive symptoms at wave 1 but in doing so we may have excluded those who were most affected by lifetime child maltreatment and may experience more depressive symptoms at wave 3. Second, a count of depressive symptoms does not constitute a diagnosis of depression despite the fact that the reliability of depressive symptoms scales is substantial (κ =0.64) (142). Third, child maltreatment is usually underreported and may have biased the results towards the null (93). Lastly, the children who were excluded from the study had more depressive symptoms and experienced more life events.

Future studies may look at the association between lifetime child maltreatment and change in depressive symptoms between waves 1 and 3. Another study could examine time as an effect modifier in the association between child maltreatment and mean number of depressive symptoms among those who had no depressive symptoms, those who met the criteria for minor depression and those who met the criteria for major

depression at wave 1. In addition, future research may examine the interaction between types of maltreatment and youth support from parents by category of abuser and severity of maltreatment. Specifically, what is the effect of support from a caregiver who is the abuser compared to a caregiver who is not the abuser on the association between child maltreatment and depressive symptoms. Also, studies should examine the effects between lifetime abuse, abuse in the last 12 months, a combination of lifetime abuse and abuse in the last 12 months, and depressive symptoms. Lastly, studies should include the association between psychological abuse and depressive symptoms in the family context. The 2012 NCANDS report shows that psychological abuse is the second most common type of abuse (44.4% of reports) in Puerto Rico after neglect (63.5% of cases) and is over 5 times higher than for the U.S. (89).

Given the serious consequences of child maltreatment and depression, it is important to further refine our understanding of the association between the two. We need to identify the risk factors that contribute to the evolution of sub-clinical depression into major depression. Many youth who do not meet the criteria for depression have been shown to be at greater risk of developing depression later in life (194, 243) and have been found to be significantly impaired, and to use more mental health services than children with diagnosed depression (95). We need to keep highlighting the importance of developing interventions to effectively screen for youth at risk of progressing to clinical depression.

Tables

Table 5.1. Socio-Demographic Characteristics and Mean Depressive Symptoms for 10-13 Year Old Puerto Rican Youth with No Depressive Symptoms at Wave 1

	Total	Total	
	(%)	Mean (SE)	p-value
Child Maltreatment			0.6533
No Maltreatment	76.3	0.20 (0.05)	
Neglect Only	4.6	0.27 (0.11)	
Physical Abuse Only	12.0	0.33 (0.21)	
Sexual Abuse Only	3.8	0.35 (0.26)	
Multiple Maltreatment	3.4	0.37 (0.26)	
Age in Years			0.0993
10	28.1	0.24 (0.07)	
11	24.7	0.33 (0.04)	
12	23.9	0.10 (0.04)	
13	23.3	0.24 (0.07)	
Mother's Level of			
Education			0.4998
< High School	41.4	0.18 (0.07)	
High School or GED	43.8	0.24 (0.07)	
> High School	14.9	0.31 (0.04)	
Mother's Depression			0.5050
No	95.6	0.21 (0.03)	
Yes	4.4	0.46 (0.19)	
Place of Residence			0.7771
Bronx	84.1	0.22 (0.05)	
New York	15.9	0.22 (0.05)	
Gender			0.6429
Male	47.6	0.22 (0.05)	
Female	52.4	0.22 (0.05)	
Family Composition			0.3758
Single	54.0	0.10(0.04)	
Parent/Caregiver	54.9	0.19 (0.04)	
Two Parent/Caregiver	45.1	0.28 (0.06)	
Exposure to Violence			
No	47.7	0.14 (0.03)	
Yes	52.3	0.30 (0.06)	
Co-Morbidity			0.0009
No	87.8	0.15 (0.03)	
Yes	12.2	0.73 (0.19)	

Table 5.2. Log-Linear Poisson Regressions of Lifetime Child Maltreatment (Wave 1) on Depressive Symptoms (Wave 3)

		Model 1	L
	b	SE	p-value
Child Maltreatment			
No Maltreatment	ref.	ref.	ref.
Neglect Only	0.58	0.99	0.5581
Physical Abuse Only	0.45	0.43	0.3055
Sexual Abuse Only	0.41	0.88	0.6408
Multiple Maltreatment	0.27	0.96	0.7793

Chapter 6 - Conclusions and public health significance: Policy implications and long term relevance

Depression is a serious public health problem with social, economic and educational costs throughout the life course that are too great to ignore (30). Depression is one of the most common mental disorders in adolescence (8) almost doubling between ages 13 and 18, with one in five adolescents reporting lifetime depression by age 18 (9, 10). Recent research started focusing on depressive symptoms as a continuum in which a high number and particular combinations of symptoms result in the diagnosis of a depressive disorder (19). In the early literature depressive symptoms among adolescents were viewed as a stage in normal development but studies now show that a high number of depressive symptoms are associated with psychosocial dysfunction and depression (32, 47). In addition, the early onset of depressive symptoms places youth at risk for higher rates of recurrence in adulthood (221). However, the course of depressive symptoms is still not well understood and while some youth with depressive symptoms progress to depression, others will not (48). Although some youth with depressive symptoms do not meet the criteria for a diagnosis of major depression they remain at a heightened risk for developing depression (19). Identifying relevant depressive symptoms at the earliest possible age and determining who will continue to have depressive symptoms is an alternative step in the prevention of depression.

In this study we identified life events as a significant risk factor for developing depressive symptoms, and our findings further support similar studies. For example, Sanchez et al. conducted a study in a similar age African-American population and found that the experience of life events across ecological levels (peer, family, and individual levels) resulted in an increase in depressive symptoms over time (252). In another study, Ge et al. examined the trajectory of depressive symptoms of 8th and 9th grade males and females from rural Iowa over an 11 year period and found a curvilinear trajectory with increases in depressive symptoms throughout early adolescence (98). However, more research is needed to identify other risk factors that determine the course of depressive symptoms, and to identify youth at risk for developing depression.

Internal and external factors such as youth coping, self-esteem, social support or life events have been researched extensively and are significantly associated with depression (252).Few studies have been conducted to explore the association between internal and external factors with depressive symptoms. However, measuring affect, behavior, and cognition is challenging because there is no gold standard (161). Often there is no consensus on what type of instrument to administer, while interviews are considered best by some, they are often lengthy and expensive to be used in a community setting (161). Checklists are short and easy to administer but self-report responses depend on the quality of the instrument, the person who is reporting and their reliability (162). Variations, conceptualizations, and measurements of key variables such as life events, coping, social support and self-esteem may explain the different findings on the effects of life events on depressive symptoms (134). In addition, the information provided through

a self-report is subjective, so retrospective self-reports may reflect the person's current emotional state and recall may be biased (165).Certain instruments were developed for a specific population and may not be reliable for other populations (28). Lastly, studying risk factors in Latino populations pose further challenges. Most studies group together Latino subpopulations, or have too small sample sizes for different Latino subgroups, or fail to study important demographic variables such as generation status or degree of acculturation (51).

Preventing the development of depressive symptoms at an early age is important to optimize the mental health and well-being of youth so they can reach their full potential. We need to further develop our understanding of how depressive symptoms may progress into depression among some youth and what factors play a role in this evolution. Our results indicate that depressive symptoms increase over time when youth experience life events whether 'total', 'negative', or 'positive or no effect'. We also found that depressive symptoms increase in the presence of certain types of life events and that some events may have a greater impact in the development of depressive symptoms. These findings could inform the development of a first-stage screening tool for youth at risk for developing depressive symptoms in community settings. We did not find any multiplicative effect modification however we found possible additive effect modification for youth coping. Youth coping was the sum of positive responses to the different coping styles the youth used. A possible explanation of the additive effect modification may be that as the number of life events increases and youth are faced with more challenges, youth who use a variety of coping styles are more effective. Our results

show that youth support from parents is a protective factor that needs to be cultivated as a resource when involving parents in interventions to improve the mental health of their children. More health care providers, caregivers, school personnel and other professionals in community settings need to get involved in identifying children at risk for developing depression. In particular, health care providers and school personnel can guide parents to focus on improving the mental health of their children.

Optimizing the mental health of youth is supported by federal policy. Healthy People 2020 seeks to "reduce the proportion of adolescents aged 12 to 17 years who experience major depressive episodes" (253). In addition, the U.S. Preventive Services Task Force recommends screening 12-18 year old adolescents for major depressive disorder to ensure their timely diagnosis, psychotherapy and follow-up(254).

Appendix 1

Mood disorders are a subset of mental disorders described in the Diagnostic and Statistical Manual of Disorders (DSM-IV). The main underlying feature of mood disorders is a disturbance in mood (inappropriate, exaggerated or a limited range of feelings). Mood disorders are divided into four categories: major depressive, dysthymic, cyclothymic and bipolar disorders. In this study, I focused on depressive symptoms as described in DSM-IV. Depressive symptoms include:

- Depressed or *irritable* mood most of the day (irritability is the only difference with an adult's diagnosis)
- Diminished pleasure or interest in all or most activities
- Significant weight loss
- Insomnia or hypersomnia
- Psychomotor agitation or retardation
- Fatigue or loss of energy
- Feelings of worthlessness or excessive guilt
- Diminished ability to think or concentrate
- Recurrent thoughts of death or suicidal ideation

The symptoms should not be due to the direct psychological effects of substance abuse, medication, a medical condition or bereavement. The approach to children and adolescent's mental health is markedly different than that of adults and the field of developmental psychopathology has provided a framework for conceptualizing childhood psychopathology. In general, anhedonia (lack of pleasure), hopelessness, hypersomnia, weight change and drug use are more common among adolescents than in adults.(8)

The DISC-IV is a respondent-based interview that is a highly structured diagnostic instrument that requires answers without interpretation by the interviewer. It was originally developed for large epidemiological studies but since then has been used in clinical settings and for prevention and screening. It is usually administered by a nonclinician (a trained lay interviewer). The instrument was originally developed in 1979 and has undergone several changes to match different disease classification systems. The DISC-IV is based on DSM-IV and ICD-10 and was first introduced in the field in 1997. It is administered and scored with computer assisted software and is available both in English and Spanish. The DISC-IV assesses more than 30 child and adolescent psychiatric diagnoses including major depressive episode/dysthymia. There are two versions of the instrument: the DISC-P for parents or knowledgeable caretakers for 6-17 years old, and the DISC-Y for children and youths aged 9-17 years. Both versions include similar questions but use different pronouns depending on who is being interviewed. The DISC-IV includes an introduction module followed by 6 modules containing related diagnoses. The major depression and dysthymia module includes a total of 60 questions. The DISC-IV assesses the presence of a diagnosis occurring in the past 12 month and the past 4 weeks.(142)

I'm now going to ask you some questions about feeling sad and unhappy.

In the last year, that is since [NAME EVENT/NAME CURRENT MONTH of last year] was there a time when you often felt sad or depressed?

IF YES

A. Was there a time in the last year when you felt sad or depressed for a long time each day? IF NO, GO TO Q2

D W/11 = 1 = 1 = 1 = 0.14 (1 = 4 = 5.14 (1 = 4 = 5.14 (1 = 4 = 1.14 (1 = 4.14 (1

Appendix 2

In the last 12 months did your parents get divorced ?
Yes0
No2
Refused7
Don't know9
IF YES, ASK:
A. Was the divorce MOSTLY GOOD for you or
MOSTLY BAD for you?
Mostly good 1
Mostly bad 2
Refused7
Don't know9
IF A IS CODED "2", ASK B,
OTHERWISE, GO TO Q6
B. Please look at CARD 16 and tell me how your
parent's divorce has affected your life
Not at all or only a little 1
A lot
Refused7
Don't know9

A	pp	end	lix	3
	PP			•

LIFE EVENTS CHECKLIST FOR ALL AGES

1. During the last 12 months, did you **move to a new home** (permanently, nota temporary residence that is not your home)?

P	
	IF YES, ASK: A. Was moving to a new home MOSTLY GOOD for you
	or MOSTLY BAD for you?
	Mostly good
	Mostly bad
	Refused
	Don't know
	IF A IS CODED "2", ASK B.
	OTHERWISE, GO TO Q. 2.
	B. Please look at CARD 16 and tell me how moving to a
	new home has affected your life. (READ):
	Not at all or only a little
	A lot
	Refused
	Don't know
2. Dur	ing the last 12 months, did you start going to a new school?
	IF YES, ASK: A. Was starting a new school MOSTLY GOOD for you or
	MOSTLY BAD for you?
	Mostly good
	Mostly bad
	Refused
	Don't know
	IF A IS CODED "2", ASK B.
	OTHERWISE, GO TO Q. 3.
	B. Please look at CARD 16 and tell me how starting a
	new school has affected your life. (READ):
	Not at all or only a little
	A lot
	Refused
	Don't know
3. Dur	ing the last 12 months, did your parents argue more than previously?
	IF YES ASK: B. Please look at CARD 16 and tell me how yourparents' arguing
	has affected your life.
	Not at all or only a little
	A lot
	Refused
	Don't know

4. During the last 12 months, did your **parents separate?**

IF YES, ASK: A. Was your parents' separating MOSTLY GOOD for
you or MOSTLY BAD for you?
Mostly good
Mostly bad
Refused
Don't know
IF A IS CODED "2", ASK B.
OTHERWISE, GO TO Q. 5.
B. Please look at CARD 16 and tell me how your
parents' separating has affected your life.
Not at all or only a little
A lot
Refused
Don't know
5. In the last 12 months, did your parents get divorced?
IF YES, ASK: A. Was the divorce MOSTLY GOOD for you or MOSTLY
BAD for you?
Mostly good
Mostly bad
Refused
Don't know
IF A IS CODED "2", ASK B.
OTHERWISE, GO TO O. 6.
B. Please look at CARD 16 and tell me how your
parents' divorce has affected your life.
Not at all or only a little
A lot
Refused
Don't know
6. In the last 12 months, did a new brother or sister arrive in your home (because one
was born, or one moved into the household)?
IF YES, ASK : A. Was having a new brother or sister MOSTLY GOOD
for you or MOSTLY BAD for you?
Mostly good
Mostly bad
Refused
Don't know
IF A IS CODED "2", ASK B.

OTHERWISE, GO TO Q. 7.

B. Please look at CARD 16 and tell me how having a new brother or sister has affected your life.Not at all or only a littleA lotRefused

Don't know

7. During the last 12 months, did a new **stepmother or stepfather move intoyour house?**

IF YES, ASK: A. Was this person moving into your house MOSTLY GOOD for you or MOSTLY BAD for you? Mostly good Mostly bad Refused Don't know IF A IS CODED "2", ASK B. **OTHERWISE, GO TO Q. 8.** B. Please look at CARD 16 and tell me how this person moving into your house has affected your life. Not at all or only a little A lot Refused Don't know 8. During the last 12 months, did you get seriously sick or injured? IF YES, ASK: B. Please look at CARD 16 and tell me how being seriously sick or injured has affected your life. Not at all or only a little A lot Refused Don't know 9. During the last 12 months were you the **victim** of a crime, a violent act, orassault? IF YES, ASK: B. Please look at CARD 16 and tell me how being the victim of a crime or assault has affected your life. Not at all or only a little A lot Refused Don't know 10. During the last 12 months, did you see a crime or accident wheresomeone was mugged, hurt or killed? IF YES, ASK: B. Please look at CARD 16 and tell me how seeing a crime or accident where someone was mugged, hurt or killed has affected your life. Not at all or only a little A lot Refused Don't know

11. During the last 12 months, did you break up with a girlfriend or boyfriend?

	IF YES, ASK: A. Was breaking up MOSTLY GOOD for you or
	MOSTLY BAD for you?
	Mostly good
	Mostly bad
	Refused
	Don't know
	IF A IS CODED "2", ASK B.
	OTHERWISE, GO TO O. 12.
	B. Please look at CARD 16 and tell me how breaking up
	has affected your life.
	Not at all or only a little
	A lot
	Refused
	Don't know
12 Di	uring the last 12 months, did a close friend die?
12. 20	IF VES ASK . B Please look at CARD 16 and tell me how having a
	close friend die has affected your life
	Not at all or only a little
	A lot
	Refused
	Don't know
13. Du friend	uring the last 12 months, did you lose a close friend not by death, or dida close I move far away?
	IF YES, ASK: B. Please look at CARD 16 and tell me how losing a
	close friend has affected your life.
	Not at all or only a little
	A lot
	Refused
	Don't know
14. Dı	uring the last 12 months, was someone in your family arrested?
	IF YES, ASK: A. Was this person's being arrested MOSTLY GOOD for
	you or MOSTLY BAD for you?
	Mostly good
	Mostly bad
	Refused
	Don't know
	IF A IS CODED "2", ASK B.
	OTHERWISE, GO TO O. 15.
	B. Please look at CARD 16 and tell me how this person's
	being arrested has affected your life.
	Not at all or only a little
	A lot
	Refused

Don't know

16. During the last 12 months, did a **family member have a drug or alcohol problem**(not including you)?

IF YES, ASK: B. Please look at **CARD 16** and tell me how this person's drug or alcohol problem has affected your life.

Not at all or only a little
A lot
Refused
Don't know

17. During the last 12 months, did a **family member** (not including you) **have amental or emotional problem?**

IF YES, ASK: B. Please look at **CARD 16** and tell me how this person's mental or emotional problem has affected your life.

Not at all or only a little
A lot
Refused
Don't know

18. During the past 12 months, was **someone in your family seriously sick orinjured? IF YES, ASK:** B. Please look at **CARD 16** and tell me how this person's

sickness or injury has affected your life.

5 5 5
Not at all or only a little
A lot
Refused
Don't know

19. During the last 12 months, did someone in the family that you loved a lotdie? IF YES, ASK: B. Please look at CARD 16 and tell me how this person's
| death has affected your life. |
|-------------------------------|
| Not at all or only a little |
| A lot |
| Refused |
| Don't know |

20. During the last 12 months, did a **pet of yours die?** IF YES, ASK: B. Please look at CARD 16 and tell me how the death of your pet has affected your life. Not at all or only a little A 16

A lot
Refused
Don't know

Appendix 4

List of Life Events Categorized by Type of Life Events

Social Adversity

During the last 12 months were you the **victim** of a crime, a violent act, or assault? During the last 12 months, did you **see a crime or accident** where someone was mugged, hurt or killed?

During the last 12 months, was someone in your family arrested?

During the last 12 months, did a **family member have a drug or alcohol problem** (not including you)?

Separation

During the last 12 months, did you **move to a new home** (permanently, not a temporary residence that is not your home)?

During the last 12 months, did you start going to a new school?

During the last 12 months, did your parents separate?

In the last 12 months, did your parents get divorced?

During the last 12 months, did you **break up** with a girlfriend or boyfriend?

During the last 12 months, did you **lose a close friend** not by death, or did a **close friend move far away?**

During the last 12 months, did someone in your family go to jail?

Death

During the last 12 months, did a **close friend die?**

During the last 12 months, did **someone in the family that you loved a lot die?** During the last 12 months, did a **pet of yours die?**

Family Environment

During the last 12 months, did your **parents argue more than previously?** In the last 12 months, did **a new brother or sister arrive in your home** (because one was born, or one moved into the household)?

During the last 12 months, did a new **stepmother or stepfather move into your house?** During the last 12 months, did you get **seriously sick or injured**?

During the last 12 months, did a **family member** (not including you) **have a mental or emotional problem?**

During the past 12 months, was someone in your family seriously sick or injured?

Appendix 5

Child Maltreatment Questionnaire

Maltreatment Type	Question	Rating that met criteria
	1. Hit you with something like a belt, brush, a stick or some other hard	
Physical abuse	object?	2 or more times
	2. Hit you with a fist or kicked you hard?	1 or more times
	3. Beat you up very bad?	1 or more times
	4. Hurt you so badly that you were cut, you had bruises on yor body or you had a broken bone or something like that?	1 or more times
Sexual abuse	1. Has an adult or another kid older than you ever tried to force you to look at or to touch their private parts?	1 or more times
	2. Has an adult or another kid older than you ever tried to touch you, grab you, or kiss you in a sexual way, or had done something sexual that made you afraid, bad or used?	1 or more times
Neglect	1. Left you alone, even when an adult should have been with you?	2 or more times
C	2. Left you without food you needed?	1 or more times
	3. Did not take you to a doctor or hospital when you needed it?	1 or more times
	4. Been so drunk or high on drugs that they could not take care of you?	1 or more times

If an individual answered positively a question within a category, then they met the criteria for that category. Questions could be answered: never (0), once (1), 2-5 times (2), more than five times (3)

References

1. Hajat A, Lucas JB, Kington R. Health outcomes among Hispanic subgroups: data from the National Health Interview Survey, 1992-95. Advance data. 2000(310):1-14. Epub 2000/09/08. PubMed PMID: 10977762.

2. Potter LB, Rogler LH, Moscicki EK. Depression among Puerto Ricans in New York City: the Hispanic Health and Nutrition Examination Survey. Social psychiatry and psychiatric epidemiology. 1995;30(4):185-93. Epub 1995/07/01. PubMed PMID: 7491514.

3. Soni A. Top 10 Most Costly Conditions among Men and Women, 2008: Estimates for the U.S. Civilian Noninstitutionalized Adult Population, Age 18 and Older Statistical Brief #331. In: Quality AfHRa, editor. Rockville, MD2011.

4. Murphy JM. Symptom Scales and Diagnostic Schedules in Adult Psychiatry. . In: Tsuang M, Tohen M, editors. Textbook in Psychiatric Epidemiology. Second ed. New York, NY: Wiley-Liss, Inc.; 2002. p. 273-332.

5. U.S. Department of Health and Human Services. Mental Health: A report of the Surgeon General: Executive Summary. Rockville, MD: U.S. Department of Health and Human Services, Substance Abuse and Mental Health Services, National Institutes of Health, National Institute of Mental Health, 1999.

6. National Institute of Mental Health. Charting the Mental Health Status and Service Needs of Children: Recommendations from the UNOCCAP Oversight Board. NIHM Archive Material. Washington, DC: National Institute of Mental Health, 1998.

7. Merikangas K, Avenevoli S, Costello J, Koretz D, Kessler RC. National comorbidity survey replication adolescent supplement (NCS-A): I. Background and measures. Journal of the American Academy of Child and Adolescent Psychiatry. 2009;48(4):367-9. Epub 2009/02/27. doi: 10.1097/CHI.0b013e31819996f1. PubMed PMID: 19242382; PubMed Central PMCID: PMC2736858.

8. Weller EB, Weller RA, Danielyan AK. Mood Disorders in Adolescents. In: Wiener J, Dulcan M, editors. Textbook of Child and Adolescent Psychiatry Third ed. Washington, DC: American Psychiatric Publishing; 2004. p. 437-84. 9. Kessler RC, Avenevoli S, Merikangas K. Mood disorders in children and adolescents: an epidemiologic perspective. Biological psychiatry. 2001;49(12):1002-14. Epub 2001/06/30. PubMed PMID: 11430842.

10. Kessler RC, Berglund P, Demler O, Jin R, Merikangas KR, Walters EE. Lifetime prevalence and age-of-onset distributions of DSM-IV disorders in the National Comorbidity Survey Replication. Archives of general psychiatry. 2005;62(6):593-602. Epub 2005/06/09. doi: 10.1001/archpsyc.62.6.593. PubMed PMID: 15939837.

11. Merikangas KR, He JP, Burstein M, Swanson SA, Avenevoli S, Cui L, et al. Lifetime prevalence of mental disorders in U.S. adolescents: results from the National Comorbidity Survey Replication--Adolescent Supplement (NCS-A). Journal of the American Academy of Child and Adolescent Psychiatry. 2010;49(10):980-9. Epub 2010/09/22. doi: 10.1016/j.jaac.2010.05.017. PubMed PMID: 20855043; PubMed Central PMCID: PMC2946114.

12. Lewinsohn PM, Rohde P, Seeley JR. Major depressive disorder in older adolescents: prevalence, risk factors, and clinical implications. Clinical psychology review. 1998;18(7):765-94. Epub 1998/11/25. PubMed PMID: 9827321.

13. Harrington R, Fudge H, Rutter M, Pickles A, Hill J. Adult outcomes of childhood and adolescent depression. I. Psychiatric status. Archives of general psychiatry. 1990;47(5):465-73. Epub 1990/05/01. PubMed PMID: 2184797.

14. Bohman H, Jonsson U, von Knorring AL, von Knorring L, Paaren A, Olsson G. Somatic symptoms as a marker for severity in adolescent depression. Acta Paediatrica 2010;99(11):1724-30.

15. U.S. Department of Health and Human Services. Mental Health: Culture, Race, and Ethnicity - A Supplements to Mental Health: A Report to the Surgeon General. Rockville, MD: U.S. Department of Health and Human Services, Substance Abuse and Mental Health Services Administration, Center for Mental Health Services., 2001.

16. Pine DS, Goldstein RB, Wolk S, Weissman MM. The association between childhood depression and adulthood body mass index. Pediatrics. 2001;107(5):1049-56. Epub 2001/05/23. PubMed PMID: 11331685.

17. Jelleyman T, Spencer N. Residential mobility in childhood and health outcomes: a systematic review. Journal of epidemiology and community health. 2008;62(7):584-92. Epub 2008/06/19. doi: 10.1136/jech.2007.060103. PubMed PMID: 18559440.

18. Pine DS, Cohen E, Cohen P, Brook J. Adolescent depressive symptoms as predictors of adult depression: Moodiness or mood disorder? Am J Psychiat. 1999;156(1):133-5. PubMed PMID: ISI:000077841700021.

19. Angold A, Costello EJ, Farmer EM, Burns BJ, Erkanli A. Impaired but undiagnosed. Journal of the American Academy of Child and Adolescent Psychiatry. 1999;38(2):129-37. Epub 1999/02/10. doi: 10.1097/00004583-199902000-00011. PubMed PMID: 9951211.

20. Pine DS, Cohen P, Gurley D, Brook J, Ma Y. The risk for early-adulthood anxiety and depressive disorders in adolescents with anxiety and depressive disorders. Archives of general psychiatry. 1998;55(1):56-64. Epub 1998/01/22. PubMed PMID: 9435761.

21. Costello EJ, Angold A, Keeler GP. Adolescent outcomes of childhood disorders: the consequences of severity and impairment. Journal of the American Academy of Child and Adolescent Psychiatry. 1999;38(2):121-8. Epub 1999/02/10. doi: 10.1097/00004583-199902000-00010. PubMed PMID: 9951210.

22. Lewinsohn PM, Clarke GN, Seeley JR, Rohde P. Major depression in community adolescents: age at onset, episode duration, and time to recurrence. Journal of the American Academy of Child and Adolescent Psychiatry. 1994;33(6):809-18. Epub 1994/07/01. doi: 10.1097/00004583-199407000-00006. PubMed PMID: 7598758.

23. Lewinsohn PM, Rohde P, Klein DN, Seeley JR. Natural course of adolescent major depressive disorder: I. Continuity into young adulthood. Journal of the American Academy of Child and Adolescent Psychiatry. 1999;38(1):56-63. Epub 1999/01/20. doi: 10.1097/00004583-199901000-00020. PubMed PMID: 9893417.

24. Cicchetti D, Aber JL. Contextualism and developmental psychopathology. Development and psychopathology. 1998;10(2):137-41. Epub 1998/06/23. PubMed PMID: 9635218.

25. Cicchetti D, Toth SL. The development of depression in children and adolescents. The American psychologist. 1998;53(2):221-41. Epub 1998/03/10. PubMed PMID: 9491749.

26. Rao U, Hammen C, Daley SE. Continuity of depression during the transition to adulthood: a 5-year longitudinal study of young women. Journal of the American Academy of Child and Adolescent Psychiatry. 1999;38(7):908-15. Epub 1999/07/16. doi: 10.1097/00004583-199907000-00022. PubMed PMID: 10405510.

27. Kovacs M, Feinberg TL, Crouse-Novak M, Paulauskas SL, Pollock M, Finkelstein R. Depressive disorders in childhood. II. A longitudinal study of the risk for a subsequent major depression. Archives of general psychiatry. 1984;41(7):643-9. Epub 1984/07/01. PubMed PMID: 6732424.

28. Carver CS, Scheier MF, Weintraub JK. Assessing coping strategies: a theoretically based approach. Journal of personality and social psychology. 1989;56(2):267-83. Epub 1989/02/01. PubMed PMID: 2926629.

29. Zuckerbrot RA, Cheung AH, Jensen PS, Stein RE, Laraque D. Guidelines for Adolescent Depression in Primary Care (GLAD-PC): I. Identification, assessment, and initial management. Pediatrics. 2007;120(5):e1299-312. Epub 2007/11/03. doi: 10.1542/peds.2007-1144. PubMed PMID: 17974723.

30. Birmaher B, Ryan ND, Williamson DE, Brent DA, Kaufman J. Childhood and adolescent depression: a review of the past 10 years. Part II. Journal of the American Academy of Child and Adolescent Psychiatry. 1996;35(12):1575-83. Epub 1996/12/01. doi: 10.1097/00004583-199612000-00008. PubMed PMID: 8973063.

31. Fergusson DM, Boden JM, Horwood LJ. Recurrence of major depression in adolescence and early adulthood, and later mental health, educational and economic outcomes. The British journal of psychiatry : the journal of mental science. 2007;191:335-42. Epub 2007/10/02. doi: 10.1192/bjp.bp.107.036079. PubMed PMID: 17906244.

32. Aalto-Setala T, Marttunen M, Tuulio-Henriksson A, Poikolainen K, Lonnqvist J. Depressive symptoms in adolescence as predictors of early adulthood depressive disorders and maladjustment. The American journal of psychiatry. 2002;159(7):1235-7. Epub 2002/07/02. PubMed PMID: 12091207.

33. Twenge JM, Nolen-Hoeksema S. Age, gender, race, socioeconomic status, and birth cohort differences on the children's depression inventory: a meta-analysis. Journal of abnormal psychology. 2002;111(4):578-88. Epub 2002/11/14. PubMed PMID: 12428771.

34. Kim-Cohen J, Caspi A, Moffitt TE, Harrington H, Milne BJ, Poulton R. Prior juvenile diagnoses in adults with mental disorder: developmental follow-back of a prospective-longitudinal cohort. Archives of general psychiatry. 2003;60(7):709-17. Epub 2003/07/16. doi: 10.1001/archpsyc.60.7.709. PubMed PMID: 12860775.

35. Cohen P, Cohen J, Kasen S, Velez CN, Hartmark C, Johnson J, et al. An epidemiological study of disorders in late childhood and adolescence--I. Age- and gender-specific prevalence. Journal of child psychology and psychiatry, and allied disciplines. 1993;34(6):851-67. Epub 1993/09/01. PubMed PMID: 8408371.

36. McGee R, Feehan M, Williams S, Anderson J. DSM-III disorders from age 11 to age 15 years. Journal of the American Academy of Child and Adolescent Psychiatry. 1992;31(1):50-9. Epub 1992/01/01. doi: 10.1097/00004583-199201000-00009. PubMed PMID: 1537781.

37. Ge X, Lorenz FO, Conger RD, Elder GH, Simons RL. Trajectories of Stressful Life Events and Depressive Symptoms during Adolescence. Developmental psychology. 1994;30(4):467-83. doi: Doi 10.1037//0012-1649.30.4.467. PubMed PMID: ISI:A1994NW05300001.

38. Canino GJ, Bird HR, Shrout PE, Rubio-Stipec M, Bravo M, Martinez R, et al. The prevalence of specific psychiatric disorders in Puerto Rico. Archives of general psychiatry. 1987;44(8):727-35. Epub 1987/08/01. PubMed PMID: 3498456.

39. Bebbington P, Hurry J, Tennant C, Sturt E, Wing JK. Epidemiology of mental disorders in Camberwell. Psychological medicine. 1981;11(3):561-79. Epub 1981/08/01. PubMed PMID: 6973770.

40. Kessler RC, McGonagle KA, Swartz M, Blazer DG, Nelson CB. Sex and depression in the National Comorbidity Survey. I: Lifetime prevalence, chronicity and recurrence. Journal of affective disorders. 1993;29(2-3):85-96. Epub 1993/10/01. PubMed PMID: 8300981.

41. Caspi A, Sugden K, Moffitt TE, Taylor A, Craig IW, Harrington H, et al. Influence of life stress on depression: moderation by a polymorphism in the 5-HTT gene. Science. 2003;301(5631):386-9. Epub 2003/07/19. doi: 10.1126/science.1083968. PubMed PMID: 12869766.

42. Leaf PJ, Alegria M, Cohen P, Goodman SH, Horwitz SM, Hoven CW, et al. Mental health service use in the community and schools: results from the fourcommunity MECA Study. Methods for the Epidemiology of Child and Adolescent Mental Disorders Study. Journal of the American Academy of Child and Adolescent Psychiatry. 1996;35(7):889-97. Epub 1996/07/01. PubMed PMID: 8768348. 43. Burcusa SL, Iacono WG. Risk for recurrence in depression. Clinical psychology review. 2007;27(8):959-85. Epub 2007/04/24. doi: 10.1016/j.cpr.2007.02.005. PubMed PMID: 17448579; PubMed Central PMCID: PMC2169519.

44. Tompson MC, McNeil FM, Rea MM, Asarnow JR. Identifying and treating adolescent depression. The Western journal of medicine. 2000;172(3):172-6. Epub 2000/03/29. PubMed PMID: 10734807; PubMed Central PMCID: PMC1070798.

45. Zubenko G, Zubenko W, Spiker D, Giles D, Kaplan B. Malignancy of recurrent, early-onset major depression: A family study. American Journal of Medical Genetics, Part B-Neuropsychiatric Genetics. 2001;105:690-9.

46. Klein DN, Lewinsohn PM, Rohde P, Seeley JR, Durbin CE. Clinical features of major depressive disorder in adolescents and their relatives: impact on familial aggregation, implications for phenotype definition, and specificity of transmission. Journal of abnormal psychology. 2002;111(1):98-106. Epub 2002/02/28. PubMed PMID: 11866183.

47. Saluja G, Iachan R, Scheidt PC, Overpeck MD, Sun W, Giedd JN. Prevalence of and risk factors for depressive symptoms among young adolescents. Archives of pediatrics & adolescent medicine. 2004;158(8):760-5. Epub 2004/08/04. doi: 10.1001/archpedi.158.8.760. PubMed PMID: 15289248.

48. Stoolmiller M, Kim HK, Capaldi DM. The course of depressive symptoms in men from early adolescence to young adulthood: identifying latent trajectories and early predictors. Journal of abnormal psychology. 2005;114(3):331-45. Epub 2005/08/25. doi: 10.1037/0021-843X.114.3.331. PubMed PMID: 16117571; PubMed Central PMCID: PMC1698962.

49. Garber J, Keiley MK, Martin C. Developmental trajectories of adolescents' depressive symptoms: predictors of change. Journal of consulting and clinical psychology. 2002;70(1):79-95. Epub 2002/02/28. PubMed PMID: 11860059.

50. Ennis SR, Rios-Vargas M, Albert NG. The Hispanic Population: 2010. In: Commerce USDo, editor. 2011: U.S. Census Bureau; 2011.

51. Vega WA, Sribney WM. Understanding the Hispanic Health Paradox through a multi-generation lens: a focus on behavior disorders. Nebraska Symposium on Motivation Nebraska Symposium on Motivation. 2011;57:151-68. Epub 2010/12/21. PubMed PMID: 21166308.

52. U.S. Department of Health and Human Services. Mental Health: Culture, Race, and Ethnicity—A Supplement to Mental Health: A Report of the Surgeon General. In: U.S. Department of Health and Human Services SAaMHSA, Center for Mental Health Services, editor. Rockville, MD2001.

53. Eaton DK, Kann L, Kinchen S, Shanklin S, Ross S, Hawkins J, et al. Surveillance Summaries: Youth Risk Behavior Surveillance --- United States, 2009. In: Prevention CfDCa, editor.: Morbidity and Mortality Weekly Report; 2009. p. 1-142.

54. Nguyen L, Huang LN, Arganza GF, Liao Q. The influence of race and ethnicity on psychiatric diagnoses and clinical characteristics of children and adolescents in children's services. Cultural diversity & ethnic minority psychology. 2007;13(1):18-25. Epub 2007/01/18. doi: 10.1037/1099-9809.13.1.18. PubMed PMID: 17227173.

55. Bernal G, Rosello J. Depression in Latino children and adolescents: prevalence, prevention and treatment. In: Aguilar-Gaxiola A, Gullotta T, editors. Depression in Latinos: Assessment, Treatment, and Prevention. New York, NY: Springer Science + Business Media, LLC; 2008. p. 263-75.

56. Lopez MH, Velasco G. A demographic portrait of Puerto Ricans, 2009. Washington, DC: Pew Hispanic Center, 2011.

57. U.S. Department of Labor. State unemployment rates in 2010: Bureau of Labor Statistics; 2011 [August 2012]. Available from: http://www.bls.gov/opub/ted/2011/ted_20110301.htm.

58. Landale NS, Oropesa R, Bradatan C. Hispanic families in the United States: Family structure and process in an era of family change. In: Tienda M, Mitchell F, editors. Hispanics and the Future of America. Washington (DC): National Academies Press; 2006.

59. National Research Council. Risks and Opportunities: Synthesis of studies on adolescence. Washington, DC: The National Academies Press; 1999.

60. Community Service Society. Latino Youth in New York City: School, Work, and Income Trends for New York's Largest Group of Young People. 2010.

61. High School Youth Risk Behavior Survey [Internet]. 2011 [cited March 2014]. Available from: <u>http://apps.nccd.cdc.gov/youthonline/App/Results.aspx?LID=PR</u>.

62. Lipman TH. The epidemiology of type I diabetes in children 0-14 yr of age in Philadelphia. Diabetes care. 1993;16(6):922-5. Epub 1993/06/01. PubMed PMID: 8325207.

63. Carter-Pokras OD, Gergen PJ. Reported asthma among Puerto Rican, Mexican-American, and Cuban children, 1982 through 1984. American journal of public health. 1993;83(4):580-2. Epub 1993/04/01. PubMed PMID: 8460740; PubMed Central PMCID: PMC1694490.

64. Popkin BM, Udry JR. Adolescent obesity increases significantly in second and third generation U.S. immigrants: the National Longitudinal Study of Adolescent Health. The Journal of nutrition. 1998;128(4):701-6. Epub 1998/05/16. PubMed PMID: 9521631.

65. Martin JA, Hamilton BE, Ventura SJ, Menacker F, Park MM, Sutton PD. Births: final data for 2001. National vital statistics reports : from the Centers for Disease Control and Prevention, National Center for Health Statistics, National Vital Statistics System. 2002;51(2):1-102. Epub 2003/02/25. PubMed PMID: 12596439.

66. Goodman SH, Hoven CW, Narrow WE, Cohen P, Fielding B, Alegria M, et al. Measurement of risk for mental disorders and competence in a psychiatric epidemiologic community survey: the National Institute of Mental Health Methods for the Epidemiology of Child and Adolescent Mental Disorders (MECA) Study. Social psychiatry and psychiatric epidemiology. 1998;33(4):162-73. Epub 1998/05/06. PubMed PMID: 9567666.

67. Canino G, Shrout PE, Rubio-Stipec M, Bird HR, Bravo M, Ramirez R, et al. The DSM-IV rates of child and adolescent disorders in Puerto Rico: prevalence, correlates, service use, and the effects of impairment. Archives of general psychiatry. 2004;61(1):85-93. Epub 2004/01/07. doi: 10.1001/archpsyc.61.1.85. PubMed PMID: 14706947.

68. Ramos Olazagasti M, Shrout P, Yoshikawa H, Canino G, Bird H. Contextual Risk and Promotive Processes in Puerto Rican Youths' Internalizing Trajectories in Puerto Rico and New York. (in press). 2012.

69. Selye H. The stress of life. New York, NY: McGraw-Hill; 1956.

70. Lazarus RS. Psychological stress and the coping process. Mew York, NY, US: McGraw-Hill; 1966.

71. Dohrenwend BP, Raphael KG, Schwartz S, Stueve A, Skoldol A. The structured event probe and narrative rating method for measuring stressful life events. In: Mazure CM, editor. Does Stress Cause Psychiatric Illness? Washington, DC: American Psychiatric Press; 1993.

72. Mazure CM. Life stressors as risk factors in depression. Clin Psychol-Sci Pr. 1998;5(3):291-313. PubMed PMID: ISI:000075420900002.

73. Allen JL, Rapee RM, Sandberg S. Severe life events and chronic adversities as antecedents to anxiety in children: a matched control study. Journal of abnormal child psychology. 2008;36(7):1047-56. Epub 2008/06/04. doi: 10.1007/s10802-008-9240-x. PubMed PMID: 18521739.

74. Willeman AM, Koot HM, Ferdinand RF, Goossens FA, Schuengel C. Change in psychopathology in referred children: The role of life events and perceived stress. Journal of Child Psychology and Psychiatry, and Allied Disciplines. 2008;49(11):1175 -83.

75. Grant I, Gerst M, Yager J. Scaling of life events by psychiatric patients and normals. Journal of psychosomatic research. 1976;20(2):141-9. Epub 1976/01/01. PubMed PMID: 1271314.

76. McFarlane AC. Long-term psychiatric morbidity after a natural disaster. Implications for disaster planners and emergency services. The Medical journal of Australia. 1986;145(11-12):561-3. Epub 1986/12/01. PubMed PMID: 3796364.

77. Thienkrua W, Cardozo BL, Chakkraband ML, Guadamuz TE, Pengjuntr W, Tantipiwatanaskul P, et al. Symptoms of posttraumatic stress disorder and depression among children in tsunami-affected areas in southern Thailand. JAMA : the journal of the American Medical Association. 2006;296(5):549-59. Epub 2006/08/03. doi: 10.1001/jama.296.5.549. PubMed PMID: 16882961.

78. Kanner AD, Coyne JC, Schaefer C, Lazarus RS. Comparison of two modes of stress measurement: daily hassles and uplifts versus major life events. J Behav Med. 1981;4(1):1-39. Epub 1981/03/01. PubMed PMID: 7288876.

79. Bolger N, DeLongis A, Kessler RC, Schilling EA. Effects of daily stress on negative mood. Journal of personality and social psychology. 1989;57(5):808-18. Epub 1989/11/01. PubMed PMID: 2810026.

80. Leeb RT, Paulozzi L, Melanson C, Simon T, Arias I. Child Maltreatment Surveillance: Uniform Definitions for Public Health and Recommended Data Elements. In: Centers for Disease Control and Prevention NCfIPaC, editor. Atlanta, GA2007.

81. Fang X, Brown DS, Florence CS, Mercy JA. The economic burden of child maltreatment in the United States and implications for prevention. Child abuse & neglect. 2012;36(2):156-65. Epub 2012/02/04. doi: 10.1016/j.chiabu.2011.10.006. PubMed PMID: 22300910; PubMed Central PMCID: PMC3776454.

82. Brown J, Cohen P, Johnson JG, Smailes EM. Childhood abuse and neglect: specificity of effects on adolescent and young adult depression and suicidality. Journal of the American Academy of Child and Adolescent Psychiatry. 1999;38(12):1490-6. Epub 1999/12/22. doi: 10.1097/00004583-199912000-00009. PubMed PMID: 10596248.

83. Dube SR, Anda RF, Felitti VJ, Chapman DP, Williamson DF, Giles WH. Childhood abuse, household dysfunction, and the risk of attempted suicide throughout the life span: findings from the Adverse Childhood Experiences Study. JAMA : the journal of the American Medical Association. 2001;286(24):3089-96. Epub 2002/01/05. PubMed PMID: 11754674.

84. Lansford JE, Dodge KA, Pettit GS, Bates JE, Crozier J, Kaplow J. A 12-year prospective study of the long-term effects of early child physical maltreatment on psychological, behavioral, and academic problems in adolescence. Archives of pediatrics & adolescent medicine. 2002;156(8):824-30. Epub 2002/07/30. PubMed PMID: 12144375; PubMed Central PMCID: PMC2756659.

85. Boden JM, Horwood LJ, Fergusson DM. Exposure to childhood sexual and physical abuse and subsequent educational achievement outcomes. Child abuse & neglect. 2007;31(10):1101-14. Epub 2007/11/13. doi: 10.1016/j.chiabu.2007.03.022. PubMed PMID: 17996302.

86. Widom CS, White HR, Czaja SJ, Marmorstein NR. Long-term effects of child abuse and neglect on alcohol use and excessive drinking in middle adulthood. Journal of studies on alcohol and drugs. 2007;68(3):317-26. Epub 2007/04/21. PubMed PMID: 17446970.

87. Skeer M, McCormick MC, Normand SL, Buka SL, Gilman SE. A prospective study of familial conflict, psychological stress, and the development of substance use disorders in adolescence. Drug and alcohol dependence. 2009;104(1-2):65-72. Epub 2009/05/19. doi: 10.1016/j.drugalcdep.2009.03.017. PubMed PMID: 19446409; PubMed Central PMCID: PMC2728005.

88. Cicchetti D, Toth SL. Child maltreatment. Annual review of clinical psychology. 2005;1:409-38. doi: DOI 10.1146/annurev.clinpsy.1.102803.144029. PubMed PMID: ISI:000255487800015.

89. U.S. Department of Health and Human Services. Child maltreatment 2012. In: Administration for Children and Families, Administration on Children YaF, Children's Bureau, editors. Washington, DC2013.

90. National Center for Injury Prevention and Control. Child Maltreatment: Facts at a Glance 2013.

91. Lindholm KJ, Willey R. Ethnic Differences in Child Abuse and Sexual Abuse. Hipsnic Journal of Behavioral Sciences. 1986;8(2):111-25.

92. Elliott KS, Urquiza A. Ethnicity, CUlture, and Child Maltreatment. Journal of Social Issues. 2006;62(4):787-809.

Gilbert R, Widom CS, Browne K, Fergusson D, Webb E, Janson S. Burden and consequences of child maltreatment in high-income countries. Lancet.
2009;373(9657):68-81. Epub 2008/12/06. doi: 10.1016/S0140-6736(08)61706-7.
PubMed PMID: 19056114.

94. Herrenkohl TI, Herrenkohl RC. Examining the Overlap and Prediction of Multiple Forms of Child Maltreatment, Stressors, and Socioeconomic Status: A Longitudinal Analysis of Youth Outcomes. Journal of Family Violence. 2007(22):553-62.

95. Gonzalez-Tejera G, Canino G, Ramirez R, Chavez L, Shrout P, Bird H, et al. Examining minor and major depression in adolescents. Journal of child psychology and psychiatry, and allied disciplines. 2005;46(8):888-99. Epub 2005/07/22. doi: 10.1111/j.1469-7610.2005.00370.x. PubMed PMID: 16033637.

96. Taylor SE, Kemeny ME, Reed GM, Bower JE, Gruenewald TL. Psychological resources, positive illusions, and health. The American psychologist. 2000;55(1):99-109. Epub 2001/06/08. PubMed PMID: 11392870.

97. Lewis MJ. Overview of development from infancy through adolescence. In: Wiener JM, Dulcan MK, editors. Textbook of Child and Adolescent Psychiatry. Third ed. Arlington, VA: American Psychiatric Publishing; 2004. p. 13-44. 98. Ge X, Natsuaki MN, Conger RD. Trajectories of depressive symptoms and stressful life events among male and female adolescents in divorced and nondivorced families. Development and psychopathology. 2006;18(1):253-73. Epub 2006/02/16. doi: 10.1017/S0954579406060147. PubMed PMID: 16478562.

99. Rutter M. Meyerian psychobiology, personality development, and the role of life experiences. The American journal of psychiatry. 1986;143(9):1077-87. Epub 1986/09/01. PubMed PMID: 3529992.

100. Brown GW, Harris TO. Social Origins of Depression. London: Free Press; 1978.

101. Hammen C, Mayol A. Depression and cognitive characteristics of stressful lifeevent types. Journal of abnormal psychology. 1982;91(3):165-74. Epub 1982/06/01. PubMed PMID: 7096786.

102. Kessler RC, McLeod JD. Sex Differences in Vulnerability to Undesirable Life Events. American Sociological Review. 1984;49(5):620-31.

103. Compas BE, Connor-Smith JK, Saltzman H, Thomsen AH, Wadsworth ME. Coping with stress during childhood and adolescence: problems, progress, and potential in theory and research. Psychological bulletin. 2001;127(1):87-127. Epub 2001/03/29. PubMed PMID: 11271757.

104. Pearlin LI, Schooler C. The structure of coping. Journal of health and social behavior. 1978;19(1):2-21. Epub 1978/03/01. PubMed PMID: 649936.

105. Coyne JC, Holroyd K. Stress, coping and illenss: a transactional perspective. In: Millon T, Green C, Meagher R, editors. Handbook of clinical health psychology. New York, NY: Plenum; 1982. p. 103-27.

106. Lazarus RS, Folkman S. Stress, appraisal and coping. New York, NY: Springer; 1984.

107. Billings AG, Cronkite RC, Moos RH. Social-environmental factors in unipolar depression: comparisons of depressed patients and nondepressed controls. Journal of abnormal psychology. 1983;92(2):119-33. Epub 1983/05/01. PubMed PMID: 6863728.

108. Folkman S, Lazarus RS. An analysis of coping in a middle-aged community sample. Journal of health and social behavior. 1980;21(3):219-39. Epub 1980/09/01. PubMed PMID: 7410799.

109. Thoits PA. Stress, coping, and social support processes: where are we? What next? Journal of health and social behavior. 1995;Spec No:53-79. Epub 1995/01/01. PubMed PMID: 7560850.

110. Billings AG, Moos RH. The role of coping responses and social resources in attenuating the stress of life events. Journal of behavioral medicine. 1981;4(2):139-57. Epub 1981/06/01. PubMed PMID: 7321033.

111. Skinner EA, Edge K, Altman J, Sherwood H. Searching for the structure of coping: a review and critique of category systems for classifying ways of coping. Psychological bulletin. 2003;129(2):216-69. Epub 2003/04/17. PubMed PMID: 12696840.

112. Thuen E, Bru E. Coping Styles and Emotional and Behavioural Problems Among Norwegian Grade 9 Students. Scandinavian Journal of Educational Research. 2004;48(5):493-510.

113. Berkman LF, Glass T. Social integration, social networks, social support, and health. In: Berkman LF, Kawachi I, editors. Social Epidemiology. New York, NY: Oxford University Press; 2000. p. 137-73.

114. Henderson AS. Interpreting the evidence on social support. Social psychiatry Sozialpsychiatrie Psychiatrie sociale. 1984;19(2):49-52. Epub 1984/01/01. PubMed PMID: 6372113.

115. Berndt TJ. Children's Friendships: Shifts Over a Half-Century in perspectives on Their Development and Their Effects. Merril-Palmer Quarterly. 2004;50(3):206-23.

116. Jackson Y, Warren JS. Appraisal, social support, and life events: predicting outcome behavior in school-age children. Child development. 2000;71(5):1441-57. Epub 2000/12/07. PubMed PMID: 11108106.

117. Sheeber L, Hops H, Alpert A, Davis B, Andrews J. Family support and conflict: Prospective relations to adolescent depression. Journal of abnormal child psychology. 1997;25(4):333-44. PubMed PMID: ISI:A1997XV71300007.

118. Brooks-Gunn J, Petersen AC. Studying the emergence of depression and depressive symptoms during adolescence. J Youth Adolesc. 1991;20(2):115-9. Epub 1991/04/01. doi: 10.1007/BF01537603. PubMed PMID: 24265001.

119. Lakey B, Cohen S. Social Support Theory and Measurement. In: Cohen S, Underwood LG, Gottlieb BH, editors. Social Support Measurement and Intervention. New York, NY: Oxford University Press; 2000. p. 29-52.

120. Needham BL. Reciprocal relationships between symptoms of depression and parental support during the transition from adolescence to young adulthood. J Youth Adolescence. 2008;37:893–905.

121. Meadows S. Evidence of parallel pathways: Gender similarity in the impact of social support on adolescent depression and delinquency. Social Forces. 2007;85:623-52.

122. Barrera C. Distinctions Between Social Support Concepts, Measures, and Models. American journal of community psychology. 1986;14(4):413-45.

123. Langellier BA, Martin MA, Canino G, Garza JR, Ortega AN. The health status of youth in Puerto Rico. Clinical pediatrics. 2012;51(6):569-73. Epub 2012/04/20. doi: 10.1177/0009922812443123. PubMed PMID: 22514193; PubMed Central PMCID: PMC3881538.

124. Ramos-Olazagasti MA, Shrout PE, Yoshikawa H, Canino GJ, Bird HR. Contextual risk and promotive processes in Puerto Rican youths' internalizing trajectories in Puerto Rico and New York. Development and psychopathology. 2013;25(3):755-71. Epub 2013/07/25. doi: 10.1017/S0954579413000151. PubMed PMID: 23880390; PubMed Central PMCID: PMC3725656.

125. Evans GW. The built environment and mental health. Journal of urban health : bulletin of the New York Academy of Medicine. 2003;80(4):536-55. Epub 2004/01/08. doi: 10.1093/jurban/jtg063. PubMed PMID: 14709704.

126. Aguilar-Gaxiola SA, Kramer EJ, Resendez C, Magaña CG. The context of depression in Latinos in the United States. In: Aguilar-Gaxiola S, Gullotta T, editors. Depression in Latinos: Assessment, treatment, and prevention. New York, NY: SpringerScience + Business Media, LLC; 2008. p. 3-28.

127. Borges G, Medina-Mora ME, Benjet C, Zambrano Ruiz J, Breslau J, Posada J, et al. Descriptive epidemiology of depression in Latin America and Hispanics in the United

States. In: Aguilar-Gaxiola A, Gullotta TP, editors. Depression in Latinos: Assessment, Treatment, and Prevention. New York, NY: Springer Science+Business Media; 2008. p. 53-71.

128. Barrera M. Distinctions between Social Support Concepts, Measures, and Models. American journal of community psychology. 1986;14(4):413-45. doi: Doi 10.1007/Bf00922627. PubMed PMID: ISI:A1986D629700005.

129. Duarte CS, Bird HR, Shrout PE, Wu P, Lewis-Fernandez R, Shen S, et al. Culture and psychiatric symptoms in Puerto Rican children: longitudinal results from one ethnic group in two contexts. Journal of child psychology and psychiatry, and allied disciplines. 2008;49(5):563-72. Epub 2008/04/11. doi: 10.1111/j.1469-7610.2007.01863.x. PubMed PMID: 18400061; PubMed Central PMCID: PMC2597153.

130. Ben-Shlomo Y, Kuh D. A life course approach to chronic disease epidemiology: conceptual models, empirical challenges and interdisciplinary perspectives. International journal of epidemiology. 2002;31(2):285-93. Epub 2002/05/01. PubMed PMID: 11980781.

131. Kuh D, Ben-Shlomo Y, Lynch J, Hallqvist J, Power C. Life course epidemiology. Journal of epidemiology and community health. 2003;57(10):778-83. Epub 2003/10/24. PubMed PMID: 14573579; PubMed Central PMCID: PMC1732305.

132. McLeod J, Almazan E. Connections between childhood and adulthood. In: Mortimer J, Shanahan M, editors. Handbook of the life course. New York, NY: Kluwer Academic/Plenum Publishers; 2003. p. 391-412.

133. Elder GH. The life course as developmental theory. Child development. 1998;69(1):1-12. Epub 1998/03/21. PubMed PMID: 9499552.

134. Kessler RC, Amminger GP, Aguilar-Gaxiola S, Alonso J, Lee S, Ustun TB. Age of onset of mental disorders: a review of recent literature. Current opinion in psychiatry. 2007;20(4):359-64. Epub 2007/06/07. doi: 10.1097/YCO.0b013e32816ebc8c. PubMed PMID: 17551351; PubMed Central PMCID: PMC1925038.

135. McCubbin IK, Joy CB, Cauble E, Comeau JK, Patterson JM, Needle RH. Family Stress and Coping: A Decade Review. Journal of Marriage and Family. 1980;42(4):855-71.

136. Conger RD, Ge X, Elder GH, Jr., Lorenz FO, Simons RL. Economic stress, coercive family process, and developmental problems of adolescents. Child development. 1994;65(2 Spec No):541-61. Epub 1994/04/01. PubMed PMID: 8013239.

137. Cicchetti D, Lynch M. Toward an Ecological Transactional Model of Community Violence and Child Maltreatment - Consequences for Childrens Development. Psychiatry. 1993;56(1):96-118. PubMed PMID: ISI:A1993LC51600010.

138. Lynch M, Cicchetti D. An ecological-transactional analysis of children and contexts: The longitudinal interplay among child maltreatment, community violence, and children's symptomatology. Development and psychopathology. 1998;10(2):235-57. doi: Doi 10.1017/S095457949800159x. PubMed PMID: ISI:000074063500006.

139. Bird HR, Canino GJ, Davies M, Duarte CS, Febo V, Ramirez R, et al. A study of disruptive behavior disorders in Puerto Rican youth: I. Background, design, and survey methods. Journal of the American Academy of Child and Adolescent Psychiatry. 2006;45(9):1032-41. Epub 2006/08/24. doi: 10.1097/01.chi.0000227878.58027.3d. PubMed PMID: 16926610.

140. Bird HR, Davies M, Duarte CS, Shen S, Loeber R, Canino GJ. A study of disruptive behavior disorders in Puerto Rican youth: II. Baseline prevalence, comorbidity, and correlates in two sites. Journal of the American Academy of Child and Adolescent Psychiatry. 2006;45(9):1042-53. doi: DOI 10.1097/01.chi.0000227878.58027.3d. PubMed PMID: ISI:000240110700003.

141. Bird HR, Canino GJ, Davies M, Zhang H, Ramirez R, Lahey BB. Prevalence and correlates of antisocial behaviors among three ethnic groups. Journal of abnormal child psychology. 2001;29(6):465-78. Epub 2002/01/05. PubMed PMID: 11761281.

142. Shaffer D, Fisher P, Lucas CP, Dulcan MK, Schwab-Stone ME. NIMH Diagnostic Interview Schedule for Children Version IV (NIMH DISC-IV): description, differences from previous versions, and reliability of some common diagnoses. Journal of the American Academy of Child and Adolescent Psychiatry. 2000;39(1):28-38. Epub 2000/01/19. doi: 10.1097/00004583-200001000-00014. PubMed PMID: 10638065.

143. Bravo M, Ribera J, Rubio-Stipec M, Canino G, Shrout P, Ramirez R, et al. Testretest reliability of the Spanish version of the Diagnostic Interview Schedule for Children (DISC-IV). Journal of abnormal child psychology. 2001;29(5):433-44. Epub 2001/11/07. PubMed PMID: 11695544. 144. Breton JJ, Bergeron L, Valla JP, Lepine S, Houde L, Gaudet N. Do Children Aged 9 through 11 Years Understand the Disc Version-2.25 Questions. Journal of the American Academy of Child and Adolescent Psychiatry. 1995;34(7):946-54. doi: Doi 10.1097/00004583-199507000-00019. PubMed PMID: ISI:A1995RG09100019.

145. Jensen PS, Rubio-Stipec M, Canino G, Bird HR, Dulcan MK, Schwab-Stone ME, et al. Parent and child contributions to diagnosis of mental disorder: are both informants always necessary? Journal of the American Academy of Child and Adolescent Psychiatry. 1999;38(12):1569-79. Epub 1999/12/22. doi: 10.1097/00004583-199912000-00019. PubMed PMID: 10596258.

146. Alegria M, Canino G, Lai S, Ramirez RR, Chavez L, Rusch D, et al. Understanding caregivers' help-seeking for Latino children's mental health care use. Medical care. 2004;42(5):447-55. Epub 2004/04/15. PubMed PMID: 15083105.

147. Feldman JM, Ortega AN, McQuaid EL, Canino G. Comorbidity between asthma attacks and internalizing disorders among Puerto Rican children at one-year follow-up. Psychosomatics. 2006;47(4):333-9. Epub 2006/07/18. doi: 10.1176/appi.psy.47.4.333. PubMed PMID: 16844893; PubMed Central PMCID: PMC2966278.

148. Ortega AN, Goodwin RD, McQuaid EL, Canino G. Parental mental health, childhood psychiatric disorders, and asthma attacks in island Puerto Rican youth. Ambulatory pediatrics : the official journal of the Ambulatory Pediatric Association. 2004;4(4):308-15. Epub 2004/07/22. doi: 10.1367/A03-169R1.1. PubMed PMID: 15264963.

149. Lucas CP, Zhang HY, Fisher PW, Shaffer D, Regier DA, Narrow WE, et al. The DISC Predictive Scales (DPS): Efficiently screening for diagnoses. Journal of the American Academy of Child and Adolescent Psychiatry. 2001;40(4):443-9. doi: Doi 10.1097/00004583-200104000-00013. PubMed PMID: ISI:000167716000013.

150. Shaffer D, Fisher P, Lucas CP. The Diagnostic Interview for Children. In: Hilsenroth M, Segal D, editors. Comprehensive Handbook of Psychological Assessment: Wiley & Sons; 2003. p. 256-99.

151. Polo AJ, Lopez SR. Diagnosis of Depression Among Latino Children and Adults.
In: Aguilar-Gaxiola SA, Gullotta TP, editors. Depression in Latinos: Assessment,
Treatment, and Prevention. New York, NY: Springer Science+Business Media, LLC;
2008. p. 163-86.

152. Bravo M, Woodbury-Farina M, Canino GJ, Rubio-Stipec M. The Spanish translation and cultural adaptation of the Diagnostic Interview Schedule for Children (DISC) in Puerto Rico. Culture, medicine and psychiatry. 1993;17(3):329-44. Epub 1993/09/01. PubMed PMID: 8269713.

153. Bird HR, Canino G, Rubio-Stipec M, Gould MS, Ribera J, Sesman M, et al. Estimates of the prevalence of childhood maladjustment in a community survey in Puerto Rico. The use of combined measures. Archives of general psychiatry. 1988;45(12):1120-6. Epub 1988/12/01. PubMed PMID: 3264147.

154. Jensen P, Roper M, Fisher P, Piacentini J, Canino G, Richters J, et al. Test-retest reliability of the Diagnostic Interview Schedule for Children (DISC 2.1). Parent, child, and combined algorithms. Archives of general psychiatry. 1995;52(1):61-71. Epub 1995/01/01. PubMed PMID: 7811163.

155. Shaffer D, Fisher P, Dulcan MK, Davies M, Piacentini J, Schwab-Stone ME, et al. The NIMH Diagnostic Interview Schedule for Children Version 2.3 (DISC-2.3): description, acceptability, prevalence rates, and performance in the MECA Study. Methods for the Epidemiology of Child and Adolescent Mental Disorders Study. Journal of the American Academy of Child and Adolescent Psychiatry. 1996;35(7):865-77. Epub 1996/07/01. PubMed PMID: 8768346.

156. Carlisle L, McClellan JM. Diagnostic Interviews. In: Dulcan MK, editor. Dulcan's textbook on child and adolescent psychiatry. Arlington, VA: American Psychiatric Publishing; 2010. p. 79-88.

157. Johnson JH, McCutcheon S. Assessing life stress in older children and adolescents: preliminary findings with the Life Events Checklist. In: Sarason IG, Spielberger CD, editors. Stress and Anxiety. Washington, DC: Hemisphere; 1980. p. 111-25.

158. Brand AH, Johnson JH. Note on the reliability of the Life Events Checklist. Psychological reports. 1982;50:1274-.

159. Seidlitz L, Diener E. Memory for positive versus negative life events: theories for the differences between happy and unhappy persons. Journal of personality and social psychology. 1993;64(4):654-64. Epub 1993/04/01. PubMed PMID: 8473982.

160. Raphael KG, Cloitre M, Dohrenwend BP. Problems of recall and misclassification with checklist methods of measuring stressful life events. Health psychology : official journal of the Division of Health Psychology, American

Psychological Association. 1991;10(1):62-74. Epub 1991/01/01. PubMed PMID: 2026132.

161. Monroe SM. Modern approaches to conceptualizing and measuring human life stress. Annual review of clinical psychology. 2008;4:33-52. doi: DOI 10.1146/annurev.clinpsy.4.022007.141207. PubMed PMID: ISI:000255649100003.

162. Allen JL, Rapee RM, Sandberg S. Assessment of Maternally Reported Life Events in Children and Adolescents: A Comparison of Interview and Checklist Methods. J Psychopathol Behav. 2012;34(2):204-15. doi: DOI 10.1007/s10862-011-9270-5. PubMed PMID: ISI:000303456700006.

163. Grant KE, Compas BE, Thurm AE, McMahon SD, Gipson PY. Stressors and child and adolescent psychopathology: Measurement issues and prospective effects. J Clin Child Adolesc. 2004;33(2):412-25. doi: DOI 10.1207/s15374424jccp3302_23. PubMed PMID: ISI:000221108900023.

164. Kessler RC. The effects of stressful life events on depression. Annual review of psychology. 1997;48:191-214. Epub 1997/01/01. doi: 10.1146/annurev.psych.48.1.191. PubMed PMID: 9046559.

165. Hammen C. Stress and depression. Annual review of clinical psychology. 2005;1:293-319. Epub 2007/08/25. doi: 10.1146/annurev.clinpsy.1.102803.143938. PubMed PMID: 17716090.

166. Dohrenwend BP. Problems in defining and sampling the relevant population of stressful life events. In: Dohrenwend B, Dohrenwend BP, editors. Stressful life events: their nature and effects. New York, NY: John Wiley & Sons; 1974. p. 275-310.

167. Dohrenwend BS. Events as stressors: a methodological inquiry. Journal of health and social behavior. 1973;14(2):167-75. Epub 1973/06/01. PubMed PMID: 4717129.

168. Cohen LH, Towbes LC, Flocco R. Effects of induced mood on self-reported life events and perceived and received social support. Journal of personality and social psychology. 1988;55(4):669-74. Epub 1988/10/01. PubMed PMID: 3193354.

169. Finkelhor D, Dziuba-Leatherman J. Children as victims of violence: a national survey. Pediatrics. 1994;94(4 Pt 1):413-20. Epub 1994/10/01. PubMed PMID: 7936846.

170. Hill R. Families under stress: Adjustment to the crises of war separation and return. Oxford, England: Harper; 1949.

171. Wu P, Bird HR, Liu X, Duarte CS, Fuller C, Fan B, et al. Trauma, posttraumatic stress symptoms, and alcohol-use initiation in children. Journal of studies on alcohol and drugs. 2010;71(3):326-34. Epub 2010/04/23. PubMed PMID: 20409425; PubMed Central PMCID: PMC2859783.

172. Wu P, Bird HR, Liu X, Fan B, Fuller C, Shen S, et al. Childhood depressive symptoms and early onset of alcohol use. Pediatrics. 2006;118(5):1907-15. Epub 2006/11/03. doi: 10.1542/peds.2006-1221. PubMed PMID: 17079561; PubMed Central PMCID: PMC3072781.

173. Sledjeski EM, Dierker LC, Bird HR, Canino G. Predicting child maltreatment among Puerto Rican children from migrant and non-migrant families. Child abuse & neglect. 2009;33(6):382-92. doi: DOI 10.1016/j.chiabu.2008.11.004. PubMed PMID: ISI:000267118800011.

174. Kline P. Reliability of Tests: Practical Issues. The Handbook of Psychological Testing. Second ed. London, England: Routledge; 1993. p. 13.

175. Zimet GD, Dahlem NW, Zimet SG, Farley GK. The Multidimensional Scale of Perceived Social Support. J Pers Assess. 1988;52(1):30-41. doi: DOI 10.1207/s15327752jpa5201_2. PubMed PMID: ISI:A1988M281200002.

176. Kaplan BH, Cassel JC, Gore S. Social support and health. Medical care. 1977;15(5 SUPPL):47-58. Epub 1977/05/01. PubMed PMID: 853781.

177. Allison P. Statistical Horizons [Internet]. Allison P, editor. 2012. [cited 2014].

178. Ramos Olazagasti MA, Shrout PE, Yoshikawa H, Bird HR, Canino G. The longitudinal relationship between parental reports of asthma and anxiety and depression symptoms among two groups of Puerto Rican youth. Journal of psychosomatic research. 2012;73:283-8.

179. Folkman S, Lazarus RS. Manual for the Ways of Coping Scale. Palo Alto, CA: Consulting Psychology press; 1988.

180. Folkman S, Moskowitz JT. Coping: pitfalls and promise. Annual review of psychology. 2004;55:745-74. Epub 2004/01/28. doi: 10.1146/annurev.psych.55.090902.141456. PubMed PMID: 14744233.

181. Rexrode KR, Petersen S, O'Toole S. The ways of coping scale: A reliability generalization study. Educational and Psychological Measurement. 2008;68(2):262-80.

182. Vickers AJ. The use of percentage change from baseline as an outcome in a controlled trial is statistically inefficient: a simulation study. BMC medical research methodology. 2001;1:6. Epub 2001/07/19. PubMed PMID: 11459516; PubMed Central PMCID: PMC34605.

183. Research Triangle Institute International. User's Manual, Release 11. . Research Triangle Park, NC: 2001.

184. Greenland S, Mickey RM. Re: "The impact of confounder selection criteria on effect estimation. American journal of epidemiology. 1989;130(5):1066. Epub 1989/11/01. PubMed PMID: 2816895.

185. Maldonado G, Greenland S. Interpreting model coefficients when the true model form is unknown. Epidemiology. 1993;4(4):310-8. Epub 1993/07/01. PubMed PMID: 8347741.

186. Baron RM, Kenny DA. The Moderator Mediator Variable Distinction in Social Psychological-Research - Conceptual, Strategic, and Statistical Considerations. Journal of personality and social psychology. 1986;51(6):1173-82. doi: Doi 10.1037/0022-3514.51.6.1173. PubMed PMID: ISI:A1986F285400010.

187. Baron RM, Kenny DA. The moderator-mediator variable distinction in social psychological research: conceptual, strategic, and statistical considerations. Journal of personality and social psychology. 1986;51(6):1173-82. Epub 1986/12/01. PubMed PMID: 3806354.

188. Korn EL, Graubard BI. Epidemiologic studies utilizing surveys: accounting for the sampling design. American journal of public health. 1991;81(9):1166-73. Epub 1991/09/01. PubMed PMID: 1951829; PubMed Central PMCID: PMC1405642.

189. Vega W, Sribney W. Latino population demographics, rsik factors, and depression: A case study of the Mexican American Prevalence and Services Survey. In:

Aguilar-Gaxiola A, Gullotta T, editors. Depression in Latinos: Assessment, Treatment, and Prevention. New York, NY: Springer Science+Business Media; 2008. p. 29-52.

190. Moussavi S, Chatterji S, Verdes E, Tandon A, Patel V, Ustun B. Depression, chronic diseases, and decrements in health: results from the World Health Surveys. Lancet. 2007;370(9590):851-8. Epub 2007/09/11. doi: 10.1016/S0140-6736(07)61415-9. PubMed PMID: 17826170.

191. Mather M. Children in Puerto Rico: Results from the 2000 Census. In: Bureau TPR, editor. Baltimore, MD: Annie E. Casey Foundation; 2003.

192. Burns BJ, Costello EJ, Angold A, Tweed D, Stangl D, Farmer EM, et al. Children's mental health service use across service sectors. Health Aff (Millwood). 1995;14(3):147-59. Epub 1995/01/01. PubMed PMID: 7498888.

193. Tompson MC, McNeil FM, Rea MM, Asarnow JR. Identifying and treating adolescent depression. Western J Med. 2000;172(3):172-6. doi: DOI 10.1136/ewjm.172.3.172. PubMed PMID: ISI:000085685500013.

194. American Psychiatric Association. Diagnostic and statistical manual of mental disorders (4th Text Revision ed.). Washington, DC: American Psychiatric Association; 2000.

195. Allen JP, Hauser ST, Bell KL, O'Connor TG. Longitudinal assessment of autonomy and relatedness in adolescent-family interactions as predictors of adolescent ego development and self-esteem. Child development. 1994;65(1):179-94. Epub 1994/02/01. PubMed PMID: 8131646.

196. Cohen LH, Burt CE, Bjorck JP. Life Stress and Adjustment - Effects of Life Events Experienced by Young Adolescents and Their Parents. Developmental psychology. 1987;23(4):583-92. doi: Doi 10.1037/0012-1649.23.4.583. PubMed PMID: ISI:A1987H951100016.

197. Paykel ES. Life events and affective disorders. Acta psychiatrica Scandinavica. 2003;108:61-6. PubMed PMID: ISI:000185220900013.

198. Tennant C. Life events, stress and depression: a review of recent findings. The Australian and New Zealand journal of psychiatry. 2002;36(2):173-82. Epub 2002/05/02. PubMed PMID: 11982537.

199. Kendler KS, Karkowski LM, Prescott CA. Causal relationship between stressful life events and the onset of major depression. The American journal of psychiatry. 1999;156(6):837-41. Epub 1999/06/09. PubMed PMID: 10360120.

200. Hankin BL. Cognitive vulnerability-stress model of depression during adolescence: investigating depressive symptom specificity in a multi-wave prospective study. Journal of abnormal child psychology. 2008;36(7):999-1014. Epub 2008/04/26. doi: 10.1007/s10802-008-9228-6. PubMed PMID: 18437551; PubMed Central PMCID: PMC2763423.

201. Holmes T, Masuda M. Life change and illnes susceptibility. In: Dohrenwend BS, Dohrenwend BP, editors. Stressful life events: their nature and effects. ew York, NY: John Wiley & Sons; 1974. p. 45-86.

202. Paykel ES, Parker RR, Penrose RJ, Rassaby ER. Depressive classification and prediction of response to phenelzine. The British journal of psychiatry : the journal of mental science. 1979;134:572-81. Epub 1979/06/01. PubMed PMID: 476367.

203. Turner RJ, Lloyd DA. Lifetime traumas and mental health: the significance of cumulative adversity. Journal of health and social behavior. 1995;36(4):360-76. Epub 1995/12/01. PubMed PMID: 8719054.

204. Harter S. The perceived competence scale for children. Child development. 1982;53(1):87-97.

205. Miller L, Warner V, Wickramaratne P, Weissman M. Self-Esteem and Depression: ten year follow-up of mothers and offspring. Journal of affective disorders. 1999;52(1-3):41-9. doi: Doi 10.1016/S0165-0327(98)00042-1. PubMed PMID: ISI:000080305800005.

206. Franko DL, Striegel-Moore RH, Brown KM, Barton BA, McMahon RP, Schreiber GB, et al. Expanding our understanding of the relationship between negative life events and depressive symptoms in black and white adolescent girls. Psychological medicine. 2004;34(7):1319-30. Epub 2005/02/09. PubMed PMID: 15697058.

207. Ge X, Conger RD, Lorenz FO, Simons RL. Parents' stressful life events and adolescent depressed mood. Journal of health and social behavior. 1994;35(1):28-44. Epub 1994/03/01. PubMed PMID: 8014428.

208. Hammen C. Self-cognitions, stressful events, and the prediction of depression in children of depressed mothers. Journal of abnormal child psychology. 1988;16(3):347-60. Epub 1988/06/01. PubMed PMID: 3403815.

209. Kendler KS, Kessler RC, Walters EE, MacLean C, Neale MC, Heath AC, et al. Stressful life events, genetic liability, and onset of an episode of major depression in women. The American journal of psychiatry. 1995;152(6):833-42. Epub 1995/06/01. PubMed PMID: 7755111.

210. Gore S, Aseltine RH. Protective processes in adolescence: matching stressors with social resources. American journal of community psychology. 1995;23(3):301-27. Epub 1995/06/01. PubMed PMID: 7572834.

211. Swearingen EM, Cohen LH. Measurement of adolescents' life events: the junior high life experiences survey. American journal of community psychology. 1985;13(1):69-85. Epub 1985/02/01. PubMed PMID: 3969968.

212. Newcomb MD, Hubba GJ, Bentler PM. A Multidimensional Assessment of Stressful Life Events among Adolescents: Derivation and Correlates. Journal of health and social behavior. 1981;22(December):400-15.

213. Cohen S, Hoberman HM. Positive Events and Social Supports as Buffers of Life Change Stress. Journal of Applied Social Psychology. 1983;13(2):99-125.

214. Monroe SM, Rohde P, Seeley JR, Lewinsohn PM. Life events and depression in adolescence: Relationship loss as a prospective risk factor for first onset of major depressive disorder. Journal of abnormal psychology. 1999;108(4):606-14. doi: Doi 10.1037/0021-843x.108.4.606. PubMed PMID: ISI:000084123200007.

215. Fresco DM, Alloy LB, Reilly-Harrington N. Association of attributional style for negative and positive events and the occurrence of life events with depression and anxiety. Journal of Social and Clinical Psychology. 2006;25(10):1140-59.

216. Angst J, Merikangas K. The depressive spectrum: diagnostic classification and course. Journal of affective disorders. 1997;45(1):31-40.

217. Meadows SO. Evidence of Parallel Pathways: Gender Similarity in the Impact of Social Support on Adolescent Depression and Delinquency. Social Forces. 2007;85(3):1143-67.

218. Newman BM, Newman PR, Griffen S, O'Connor K, Spas J. The relationship of social support to depressive symptoms during the transition to high school. Adolescence. 2007;42(167):441-59. Epub 2007/12/01. PubMed PMID: 18047232.

219. Downey G, Coyne JC. Children of depressed parents: an integrative review. Psychological bulletin. 1990;108(1):50-76. Epub 1990/07/01. PubMed PMID: 2200073.

220. Vega WA, Sribney WM. Latino population demographics, risk factors, and depression: A case study of the Mexican American Prevalence and Services Survey. In: Aguilar-Gaxiola SA, Gullotta TP, editors. Depression in Latinos: Assessment, treatment, and prevention. New York, NY: Springer Science+Business Media, LLC; 2008. p. 29-52.

221. Fombonne E, Wostear G, Cooper V, Harrington R, Rutter M. The Maudsley longterm follow-up of child and adolescent depression. 2. Suicidality, criminality and social dysfunction in adulthood. The British journal of psychiatry : the journal of mental science. 2001;179:218-23. Epub 2001/09/05. PubMed PMID: 11532798.

222. Hammen C, Brennan PA, Keenan-Miller D, Herr NR. Early onset recurrent subtype of adolescent depression: clinical and psychosocial correlates. Journal of child psychology and psychiatry, and allied disciplines. 2008;49(4):433-40. Epub 2008/01/29. doi: 10.1111/j.1469-7610.2007.01850.x. PubMed PMID: 18221352.

223. Sarason IG, Johnson JH, Siegel JM. Assessing the impact of life changes: development of the Life Experiences Survey. Journal of consulting and clinical psychology. 1978;46(5):932-46. Epub 1978/10/01. PubMed PMID: 701572.

224. Williamson DE, Birmaher B, Ryan ND, Shiffrin TP, Lusky JA, Protopapa J, et al. The stressful life events schedule for children and adolescents: development and validation. Psychiatry research. 2003;119(3):225-41. Epub 2003/08/14. PubMed PMID: 12914894.

225. Dohrenwend BS, Krasnoff L, Askenasy AR, Dohrenwend BP. Exemplification of a method for scaling life events: the Peri Life Events Scale. Journal of health and social behavior. 1978;19(2):205-29. Epub 1978/06/01. PubMed PMID: 681735.

226. Compas BE, Davis GE, Forsythe CJ. Characteristics of life events during adolescence. American journal of community psychology. 1985;13(6):677-91. Epub 1985/12/01. PubMed PMID: 4083274.

227. Paykel ES. Recent life events in the development

of the depressive disorders. In: Depue R, editor. The psychobiology of the depressive disorders: Implications for the effects of stress. New York, NY: Academic Press; 1979.

228. Dohrenwend BS, Dohrenwend BP. A brief historical introduction to research on stressful life events. In: Dohrenwend BS, Dohrenwend BP, editors. Stressful life events: their nature and effects. New York, NY: John Wiley & Sons; 1974. p. 1-6.

229. Finlay-Jones R. Showing that life events are a cause of depression--a review. The Australian and New Zealand journal of psychiatry. 1981;15(3):229-38. Epub 1981/09/01. PubMed PMID: 6949581.

230. Vinokur A, Selzer ML. Desirable versus undesirable life events: their relationship to stress and mental distress. Journal of personality and social psychology.
1975;32(2):329-37. Epub 1975/08/01. PubMed PMID: 1239500.

231. Avison WR, Turner RJ. Stressful life events and depressive symptoms: disaggregating the effects of acute stressors and chronic strains. Journal of health and social behavior. 1988;29(3):253-64. Epub 1988/09/01. PubMed PMID: 3241066.

232. Timmermans M, van Lier PA, Koot HM. The role of stressful events in the development of behavioural and emotional problems from early childhood to late adolescence. Psychological medicine. 2010;40(10):1659-68. Epub 2010/01/09. doi: 10.1017/S0033291709992091. PubMed PMID: 20056023.

233. Mueller DP, Edwards DW, Yarvis RM. Stressful life events and psychiatric symptomatology: change or undesirability? Journal of health and social behavior. 1977;18(3):307-17. Epub 1977/09/01. PubMed PMID: 903599.

234. Felner RD, Stolberg A, Cowen EL. Crisis events and school mental health referral patterns of young children. Journal of consulting and clinical psychology. 1975;43(3):305-10. Epub 1975/06/01. PubMed PMID: 1159117.

235. Bonanno GA. Loss, trauma, and human resilience: have we underestimated the human capacity to thrive after extremely aversive events? The American psychologist. 2004;59(1):20-8. Epub 2004/01/23. doi: 10.1037/0003-066X.59.1.20. PubMed PMID: 14736317.

236. Mazure CM, Bruce ML, Maciejewski PK, Jacobs SC. Adverse life events and cognitive-personality characteristics in the prediction of major depression and

antidepressant response. The American journal of psychiatry. 2000;157(6):896-903. Epub 2000/06/01. PubMed PMID: 10831468.

237. Garmezy N. Resilience in children's adaptation to negative life events and stressed environments. Pediatric annals. 1991;20(9):459-60, 63-6. Epub 1991/09/01. PubMed PMID: 1945543.

238. Gliem JA, Gliem RR. Calculating, interpreting, and reporting Cronbach's alpha reliability coefficient for Likert-type scales. Midewest Research-to-Practice Conference in Adult, Continuing, and Community Education; The Ohio State University, Columbus, OH2003. p. 82-8.

239. Brown GW, Bifulco A, Harris TO. Life events, vulnerability and onset of depression: some refinements. The British journal of psychiatry : the journal of mental science. 1987;150:30-42. Epub 1987/01/01. PubMed PMID: 3651696.

240. Steinberg L. We know some things: parent-adolescent relationships in retrospect and prospect. Journal of Research on Adolescence. 2001;11(1):1-19.

241. Eckenrode J. Impact of chronic and acute stressors on daily reports of mood. Journal of personality and social psychology. 1984;46(4):907-18. Epub 1984/04/01. PubMed PMID: 6737199.

242. Wickrama KA, Conger RD, Lorenz FO, Jung T. Family antecedents and consequences of trajectories of depressive symptoms from adolescence to young adulthood: a life course investigation. Journal of health and social behavior. 2008;49(4):468-83. Epub 2009/02/03. PubMed PMID: 19181050; PubMed Central PMCID: PMC2741725.

243. Wittchen HU, Ustun TB, Kessler RC. Diagnosing mental disorders in the community. A difference that matters? Psychological medicine. 1999;29(5):1021-7. Epub 1999/11/27. PubMed PMID: 10576295.

244. Berndt ER, Koran LM, Finkelstein SN, Gelenberg AJ, Kornstein SG, Miller IM, et al. Lost human capital from early-onset chronic depression. Am J Psychiat. 2000;157(6):940-7. doi: DOI 10.1176/appi.ajp.157.6.940. PubMed PMID: ISI:000087421300014.

245. Langhinrichsen-Rohling J, Monson CM, Meyer KA, Caster J, Sanders A. The associations among family-of-origin violence and young adults' current depressed,

hopeless, suicidal, and life-threatening behavior. Journal of Family Violence. 1998;13(3):243-61. doi: Doi 10.1023/A:1022888905686. PubMed PMID: ISI:000075413700003.

246. Rotheram-Borus MJ, Mahler KA, Koopman C, Langabeer K. Sexual abuse history and associated multiple risk behavior in adolescent runaways. The American journal of orthopsychiatry. 1996;66(3):390-400. Epub 1996/07/01. PubMed PMID: 8827262.

247. Arata CM, Langhinrichsen-Rohling J, Bowers D, O'Farrill-Swails L. Single versus multi-type maltreatment: as examination of long-term effects of child abuse. Jouranl of Aggression, Maltreatment & Trauma. 2005;11(4):29-52.

248. Cohen S, Wills TA. Stress, Social Support, and the Buffering Hypothesis. Psychological bulletin. 1985;98(2):310-57. doi: Doi 10.1037//0033-2909.98.2.310. PubMed PMID: ISI:A1985AQC4600006.

249. Herman-Stahl M, Petersen AC. The protective role of coping and social resources for depressive symptoms among young adolescents. J Youth Adolescence. 1996;25(6):733-53.

250. Seiffge-Krenke I. Causal links between stressful events, coping style, and adolescent symptomatology. Journal of adolescence. 2000;23(6):675-91. doi: DOI 10.1006/jado.2000.0352. PubMed PMID: ISI:000166308100003.

251. Rosenbaum-Asarnow J, Carlson GA, Guthrie D. Coping strategies, selfperceptions, hopelessness, and perceived family environments in depressed and suicidal children. J Consult Clin Psychol 1987;55:361-6.

252. Dumont M, Provost MA. Resilience in Adolescents: Protective Role of Social Support, Coping Strategies, Self-Esteem, and Social Activities on Experience of Stress and Depression. J Youth Adolescence. 1999;28(3):343-63.

253. U.S. Department of Health and Human Services. Healthy People 2020 Washington, DC [March 2014]. Available from: <u>http://www.healthypeople.gov/2020/default.aspx</u>.

254. U.S. Preventive Services Task Force. Recommendations. Available from: http://www.uspreventiveservicestaskforce.org/recommendations.htm.